

Integrated Pollution Prevention and Control (IPPC)

Technical Guidance for Non-Ferrous Metals and the Production of Carbon and Graphite

Commissioning Organisation Environment Agency Rio House Waterside Drive Aztec West Almondsbury Bristol BS32 4UD

Tel 01454 624400 Fax 01454 624409

© Environment Agency

First Published 2002

ISBN 0 11 310173 2

This document is Environment Agency Copyright. We specifically allow the following:

- Internal business or personal use. You may use this document for your own private use or for use within your business without restriction.
- Giving copies to others. You may do this without restriction provided that you make no charge.

If you wish to use this document in any way other than as set out above including in particular for commercial gain, for example by way of rental, licence, sale or providing services you should contact:

Liz Greenland Environment Agency Scientific and Technical Information Service 2440 The Quadrant Aztec West Almondsbury Bristol BS32 4AQ

Record of changes

Version	Date	Change
Consultation	April 2001	Initial issue
Version 1	January 2002	Minor amendments in both content and style following full public consultation.

Note:

Queries about the content of the document should be made to Frank Farrell (0117 914 2871) or any member of the IPPC Project or Technical Guidance Teams.

Written comments or suggested improvements should be sent to Graham Winter, at the Environment Agency Technical Guidance Section by email at graham.winter@environment-agency.gov.uk or at:

Environmental Protection National Service Environment Agency Block 1. Government Buildings, Burghill Road, Westbury-on-Trym, Bristol. BS10 6BF Tel 0117 914 2868

Executive summary

This Guidance has been produced by the Environment Agency for England and Wales in collaboration with the Scottish Environment Protection Agency (SEPA) and the Northern Ireland Environment and Heritage Service (EHS). Together these are referred to as "the Regulator" or "the Regulators" in this document. Its publication follows consultation with industry, government departments and non-governmental organisations.

What is IPPC? Integrated Pollution Prevention and Control (IPPC) is a regulatory system that employs an integrated approach to control the environmental impacts of certain industrial activities. It determines the appropriate controls for industry to protect the environment through a single permitting process. To gain a Permit, Operators will have to show that they have systematically developed proposals to apply the "best available techniques" (BAT) and meet certain other requirements, taking account of local factors.

The Agencies intend to implement IPPC to:

- protect the environment as a whole
- · promote the use of "clean technology" to minimise waste at source
- encourage innovation, by making industrial operators responsible for developing solutions to
 environmental issues
- provide a "one-stop shop" for administering applications for Permits to operate.

Once a Permit has been issued, other parts of IPPC come into play. These include compliance monitoring, periodic Permit reviews, variation of Permit conditions and transfers of Permits between Operators. IPPC also provides for the restoration of industrial sites when the permitted activities cease to operate.

This Guidance and the BREF and the BREF This UK Guidance for delivering the PPC (IPPC) Regulations in the Non-Ferrous Metals sector is based on the BAT Reference document BREF (see Ref. 3) produced by the European Commission. The BREF is the result of an exchange of information between member states and industry. The quality, comprehensiveness and usefulness of the BREF is acknowledged. This Guidance is designed to complement the BREF and is cross-referenced to it throughout. It takes into account the information contained in the BREF and lays down the standards and expectations in the UK (England and Wales, Scotland and Northern Ireland) for the techniques and standards that need to be addressed to satisfy the Regulations. The reader is advised to have access to the BREF.

this Guidance The aims of this Guidance are to:

- provide a clear structure and methodology for operators to follow when making an application to
 ensure that all aspects of the PPC Regulations (see Appendix 2 for equivalent legislation in
 Scotland and Northern Ireland) and other relevant Regulations have been addressed (see Section
 1.3);
- minimise the effort by both Operator and Regulator in the permitting of an installation by use of clear indicative standards and the use of material from previous applications and from accredited Environmental Management Systems (EMS);
- improve the consistency of applications;
- increase the transparency of the permitting process by having a structure in which the Operator's
 response to each issue, and any departures from the standards, can be seen clearly;
- improve consistency of regulation across installations and sectors by facilitating the comparison of applications;
- provide a summary of some BAT techniques for pollution control which are common to all sectors, expressed, as clear indicative standards which need to be addressed by applicants;
- provide an arrangement of information which allows the reader to find, quickly, all of the guidance associated with:
 - a subject (e.g. accidents, energy or noise) (Sections 2.1 and 2.5–2.11);
 - the technical areas (e.g. smelting or effluent treatment) (Sections 2.3–2.4);
 - particular emissions (e.g. gases or particulate) (Section 3).

To assist Operators in making applications, separate technical guidance is also available on a range of topics such as waste minimisation, monitoring, calculating stack heights. The majority of this guidance is available free through the Environment Agency, SEPA or EHS (Northern Ireland) web sites (see References).

The aims of

CONTENTS

1	Intro	duction	1
	1.1	Understanding IPPC and BAT	. 1
	1.2	Making an application	
	1.3	Installations covered by this Note	
		1.3.1 The non-ferrous metals BREF document	
		1.3.2 BREF document contents	
	1.4	Review periods	
	1.5	Key issues for the sector	
	1.6	Summary of releases for each sector	
	1.7	Overview of the activities in this sector	
		1.7.1 Processes used in the manufacture of non-ferrous metals	
	1.8	Economic aspects for each sector	
2		niques for Pollution Control	
_	2.1	Management techniques	
	2.2	Materials inputs	
		2.2.1 Raw materials selection and pre-treatment	
		2.2.2 Waste minimisation (minimising the use of raw materials)	
		2.2.3 Water use	
	2.3	The main activities and abatement	
		2.3.1 Copper and copper alloys	
		2.3.2 Aluminium and aluminium alloys	
		2.3.3 Lead, zinc and cadmium	
		2.3.4 Precious metals	
		2.3.5 Refractory metals	
		2.3.6 Ferro alloys	
		2.3.7 Alkali and alkaline earth metals	55
		2.3.8 Nickel	
		2.3.9 Carbon and graphite electrodes	
		2.3.10 Abatement of point source emissions to air	
		2.3.11 Abatement of point source emissions to surface water and sewer	
		2.3.12 Control of <i>fugitive</i> emissions to air	
		2.3.13 Control of fugitive emissions to surface water, sewer and groundwater.	66
		2.3.14 Odour	67
	2.4	Emissions to groundwater	68
	2.5	Waste handling	.70
	2.6	Waste recovery or disposal	71
	2.7	Energy	.73
		2.7.1 Basic energy requirements (1)	73
		2.7.2 Basic energy requirements (2)	.74
		2.7.3 Sector-specific energy requirements	75
	2.8	Accidents and their consequences	.77
	2.9	Noise and vibration	80
	2.10	Monitoring	. 82
	2.11	Decommissioning	. 87
	2.12	Installation-wide issues	89
3	Emis	sion Benchmarks	90
	3.1	Emissions inventory and benchmark comparison	۵n
	3.2	The emission benchmarks	
	5.2	3.2.1 Emissions to air associated with the use of BAT	
		3.2.2 VOCs	
		3.2.3 Emissions to water associated with the use of BAT	
		3.2.4 Standards and obligations	
		3.2.5 Units for benchmarks and setting limits in Permits	
_	_	-	
4	Impa	.ct1	03
	4.1	Assessment of the impact of emissions on the environment1	03
	4.2	The Waste Management Licensing Regulations 1	

	4.3	The Habitats Regulations	105
Ref	erence	es	
Def	inition	s and abbreviations	
API	PENDI	X 1 Some Common Monitoring and Sampling Methods	
API	PENDI	X 2 Equivalent Legislation In Scotland and	111
Nor	thern	Ireland	111

TABLE OF FIGURES

Figure 1.1	Activities involved in the production of non-ferrous metals from ores or
	concentrates6

	DUCTION		CHNIQUES	S EMI	SSIONS		ACT						
IPPC and BAT	Making an application	Installations covered	Review periods	Key issues	Summary of releases	Sector overview	Economic aspects						
		FRODUC Inderstan		and BAT									
IPPC and the Regulations	approach t the approp gain a Per the "best a	Integrated Pollution Prevention and Control (IPPC) is a regulatory system that employs an integrated approach to control the environmental impacts of certain industrial activities. It involves determining the appropriate controls for industry to protect the environment through a single permitting process. To gain a Permit, Operators will have to show that they have systematically developed proposals to apply the "best available techniques" (BAT) and meet certain other requirements, taking account of relevant local factors.											
	The essence of BAT is that the selection of techniques to protect the environment should achieve an appropriate balance between realising environmental benefits and costs incurred by Operators. IPPC operates under the Pollution Prevention and Control (England and Wales) Regulations (see Ref. 2 and Appendix 2). These Regulations have been made under the Pollution Prevention and Control (PPC) Act 1999 (see Ref. 1) and implement the EC Directive 96/61 on IPPC. Further information on the overall system of IPPC, together with Government policy and more detailed advice on the interpretation of the Regulations, can be found in the Department for Environment, Food and Rural Affairs (DEFRA) document <i>IPPC: A Practical Guide</i> (see Ref. 5).												
Installation- based, NOT national emission limits	limits (exce	The "BAT" approach of IPPC is different from regulatory approaches based on fixed national emission limits (except where General Binding Rules have been issued by the Secretary of State). The legal instrument which ultimately defines BAT is the Permit and this can only be issued at the installation											
Indicative BAT standards	there is str technical of Section 3. which the i 3 of the Re account of environme	Indicative BAT standards are laid out in national Guidance (such as this) and should be applied unless there is strong justification for another course of action. It should be noted that BAT includes both the technical components of the installation given in Section 2 and the benchmark release levels quoted in Section 3. (The definitive interpretation of "best available techniques", which also includes the way in which the installation is designed, built, maintained, operated and decommissioned, appears in Section 3 of the Regulations (SI 2000 No 1973).) Departures from these standards may be required to take account of the technical characteristics of the installation, and its geographical location or local environmental conditions. All such departures, in whichever direction, must be justified at the local level. Notwithstanding this, if there are any applicable mandatory EU emission limits, they must be met											
BAT and EQSs	environme where this prevented environme as a recipion practicable reasonably and only th	The "BAT" approach is also different from, but complementary to, regulatory approaches based on environmental quality standards (EQS). Essentially BAT requires measures to be taken to prevent or, where this is not practicable, to reduce emissions. That is, if emissions can be reduced further, or prevented altogether, at reasonable cost, then this should be done irrespective of whether any environmental quality standards are already being met. It requires us not to consider the environment as a recipient of pollutants and waste, which can be filled up to a given level, but to do all that is practicable to minimise the impact of industrial activities. The process considers what can be reasonably achieved within the installation first (this is covered by Sections 2 and 3 of this Guidance) and only then checks to ensure that the local environmental conditions are secure (Section 4 of this Guidance and Ref. 7). The BAT approach is, in this respect, a more precautionary one, which may go											
	still threate However, f assessmer harm), sho	ned. The Regu his situation sho nt of harm. The	lations therefore ould arise very r BAT assessme as have come to	e allow for expen arely assuming t nt, which balanc	F may lead to a s diture beyond B, hat the EQS is s es cost against b usion about the o	AT where nec oundly based penefit (or prev	essary. on an vention of						
				al quality standar 5) and in Sectio	rds and other sta n 3.	ndards and ol	oligations is						
Assessing BAT at the sector level	BAT refere informatic flexibility to account th expectatio of all, repre- in that sec	ence document (on which member o member states e information co ns in the UK. At esent an approp tor. Secondly, th	BREF) for each er states should in its applicatio ntained in the B this national lev riate balance of ne techniques sl	sector. The BR take into accour n. The UK Sector REF and lays do vel, techniques v costs and benef	At the Europea EF is the result of the when determine or Guidance Not own the indicative which are conside its for a typical, vie e affordable with ide.	of an exchang ing BAT, but v e for that sect e standards a ered to be BA well-performin	ge of which leaves or takes into nd T should, first g installation						

INTRO	DUCTION	TEC	HNIQUES	5	EMIS	SSIONS	IMP.	ACT					
IPPC and BAT	Making an application	Installations covered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects					
BAI	application	oovered	periodo			10100000	overview	dopoolo					
Assessing BAT at the installation level	departures may deper costs and l company p	may be justified ad upon local fac benefits of the a profitability is no t	l in either directi tors and, where vailable options considered.	on as the a may t	described nswer is n be needed	BAT standards a above. The mo ot self-evident, a to establish the	st appropriate local assessr best option. Ir	technique nent of the ndividual					
	installation grounds of	In summary, departures may be justified on the grounds of the technical characteristics of the installation concerned, its geographical location and the local environmental conditions but not on grounds of individual company profitability. Further information on this can be found in the Guide for Applicants (see Refs. 5 and 6). Costs may only be taken into account at the local level:											
		•											
	item of differer	 where the BAT cost/benefit balance of an improvement only becomes favourable when the relevant item of plant is due for renewal/renovation anyway (e.g. BAT for the sector may be to change to a different design of furnace when a furnace comes up for rebuild). In effect, these are cases where BAT for the sector can be expressed in terms of local investment cycles. 											
	approp		it is not so long			d, a phasing pro rewarding a poo							
Innovation	meet the B performand appropriate the best av delay the in installation at the insta	AT criteria and a ce of the process indicative stand vailable techniqu ntroduction of im may allow for o	are looking for c s as a part of pro- dards at the time es relevant to th proved, availab poprtunities not cept in the case	ontinu ogress e of w le acti le tecl forese	ious impro sive sustai riting. Hov vity and th hniques. T een in the 0	n of new and inner vement in the ownable developme vever, Operators is Note may not The technical cha Guidance; as BA lid to consider th	verall environm ent. This Note s should keep be cited in an aracteristics of T is ultimately	ental describes the up to date with attempt to a particular determined					
New installations	departures requiremer such as wh	from them in th	e case of new a ally be in place b nent is for an au	ctivitie before idit of	es. For new	activities but it w w installations th encement of ope perations, this is	e indicative BA erations. In so	AT ome cases,					
Existing installations– standards	acceptable using differ case it may small decre	where the active rent plant or pro- y impose a dispr ease in emission	ity operates to a cesses from that oportionate cost is.	t stand t upor t to re	dard that is which the place the c	ed timescale) ma s very close to ar indicative requine and plant with the	n indicative rec rement is base new techniqu	quirement, but ed. In such a es for only a					
	whole prog	ramme of any o nit. Any longer t	ther items shou	d be (completed	ales agreed with <u>at the latest</u> with ed by the Operate	nin three years	of the issue					

The applicant should include a proposed timetable covering all improvements.



- **Note 1** The amount of detail needed to support the application should be sufficient to support the applicant's contention that either the conditions of the Guidance have been met or an alternative measure has been justified. The level of detail should be commensurate with the scale of the operation and its ability to cause pollution. An applicant is not required to supply detail that could not reasonably be expected to contribute to a decision to issue a Permit.
- **Note 2** For existing IPC or Waste Management permit holders, your response to each point in Sections 2, 3 or 4 may rely heavily on your previous application. The Regulator does not wish you to duplicate information as long as the previous information adequately addresses the issues. However, the more the information can be reorganised to demonstrate that all the issues have been adequately addressed, the better. You will need to send us copies of any information referred to.
- **Note 3** The contents of the outlined BAT boxes in Sections 2, 3 and 4, and additional blank tables, etc., are available electronically on the Environment Agency's web site, for the assistance of applicants.

INTRODUCTION		TECHNIQUES		EMISSIONS		IMPACT			
IPPC and BAT	Making an application		llations /ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

1.3 Installations covered by this Note

This Note covers installations described in paragraphs (a) to (f), (h) and (i) of Part A (Part A (1) in England and Wales) of Section 2.2 of Schedule 1 to the PPC Regulations (Ref. 2), and also installations described in Section 6.2 Part A (1) paragraph (a) and Section 6.3 Part A paragraph (a) subparagraph (ii). *Note: In Scotland Part A only is used and is a combination of the A1 and A2 processes.*

Non-ferrous metals

2.2 Part A (1)

- (a) Unless falling within Part A (2) of this Section, producing non-ferrous metals from ore, concentrates or secondary raw materials by metallurgical, chemical or electrolytic activities.
- (b) Melting, including making alloys, of non-ferrous metals, including recovered products (refining, foundry casting, etc.) where:
 - (i) the plant has a melting capacity of more than 4 tonnes per day for lead or cadmium or 20 tonnes per day for all other metals; and
 - (ii) any furnace, bath or other holding vessel used in the plant for the melting has a design holding capacity of 5 tonnes or more.
- (c) Refining any non-ferrous metal or alloy, other than the electrolytic refining of copper, except where the activity is related to an activity described in paragraph (a) of Part A (2), or paragraph (a), (d), or (e) of Part B, of this Section.
- (d) Producing, melting or recovering by chemicals means or by the use of heat, lead or any lead alloy, if:
 - (i) the activity may result in the release into the air of lead; and
 - (ii) in the case of lead alloy, the percentage by weight of lead in the alloy in molten form is more than 23 per cent if the alloy contains copper and 2 per cent in other cases.
- (e) Recovering any of the following elements if the activity may result in their release into the air: gallium; indium; palladium; tellurium; thallium.
- (f) Producing, melting or recovering (whether by chemical means or by electrolysis or by the use of heat) cadmium or mercury or any alloy containing more than 0.05 per cent by weight of either of those metals or, in aggregate, of both.
- (h) Manufacturing or repairing involving the use of beryllium or selenium or an alloy containing one or both of those metals if the activity may result in the release into the air of any of the substances listed in paragraph 12 of Part 2 to this Schedule; but an activity does not fall within this paragraph by reason of it involving an alloy that contains beryllium if that alloy in molten form contains less than 0.1 per cent by weight of beryllium and the activity falls within paragraph (a) or (d) of Part B of this Section.
- (i) Pelletising, calcining, roasting or sintering any non-ferrous metal ore or any mixture of such ore and other materials.

But this does not cover:

(g) Mining zinc or tin bearing ores where the activity may result in the release into water of cadmium or any compound of cadmium in a concentration which is greater than the background concentration.

Carbon activities

6.2 Part A (1)

(a) Production of carbon or hard burned coal by means of incineration or graphitisation.

Tar and bitumen activities

6.3 Part A (1)

(a) (ii) Heating tar or bitumen for the manufacture of electrodes or carbon based refractory materials, where the carrying out of the activity by the person concerned at the location in question is likely to involve the use in any period of 12 months of 5 tonnes or more of tar or of bitumen or, in aggregate, both.

INTRODUCTION		TECHNIQUES		EMISSIONS		IMPACT			
IPPC and BAT	Making an application		llations /ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Interpretation of Part A (1)

In paragraph (g), "background concentration" means any concentration of cadmium or any compound of cadmium which would be present in the release irrespective of any effect the activity may have had on the composition of the release and, without prejudice to the generality of the foregoing, includes such concentration of those substances as is present in:

- (i) water supplied to the site where the activity is carried out
- (ii) water abstracted for use in the activity
- (iii) precipitation onto the site on which the activity is carried out.

Installation boundaries

"Installation" is defined in Part 1 of the Regulations and means:

- (i) a stationary technical unit where one or more of the activities listed (above) are carried out; and
- (ii) any other location on the same site where any other directly associated activities are carried out which have a technical connection with the activities carried out in the stationary technical unit and which could have an effect on pollution.

They include as appropriate:

- the storage and handling of raw materials
- the storage and handling of fuels
- all smelting, melting and remelting and refining
- holding furnaces
- all transfer operations involving molten metal, whether by launder, ladle or any other means
- handling treatment and storage prior to removal of slags, drosses and other process wastes
- handling treatment and storage of wastes arising from maintenance activities e.g. insulation materials
- finishing activities such as rolling or extrusion where they may have an effect upon emissions and pollution
- waste-water treatment systems.

Holding capacity

When considering the holding capacity of a furnace for the purposes of Part A (1) (b), the following points shall be taken into consideration:

- Where the furnace is to be used to make a range of alloys, then the capacity shall be calculated on the basis of the densest alloy.
- Where it is necessary to retain a molten heel of metal within the furnace after pouring, this retained metal shall be included when calculating the holding capacity.
- The design capacity shall be taken to allow for a reasonable freeboard above the surface of molten material to allow for safe working or pouring.
- Design capacity shall be taken to mean the capacity of the furnace when lined with new refractory.

INTRODUCTION		TECHNIQUES		EMISSIONS		IMPACT			
IPPC and BAT	Making an application		llations /ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Production sequences

The main routes whereby non-ferrous metals may be produced from ores or concentrates Part A (1) (a), including Part A (1) (i) where appropriate, are shown in Figure 1.1.

Figure 1.1 Activities involved in the production of non-ferrous metals from ores or concentrates

Mining of ores Mining only requires an IPPC Permit if subsection (g), which relates to tin or zinc bearing ores, applies and then only where the activity may give rise to a release into water of cadmium or any compound of cadmium in a concentration which is greater than the background concentration. No such mines are currently in operation and mining is not considered further within this Note. **Preparation of concentrates** In most cases the desired mineral is present in the mined ore at a low concentration. Smelling such low-grade ore is uneconomic and would give rise to large quantities of slag. Concentration processes are used to reduce the proportion of gangue to the point at which smelting is economic. Processes involved include crushing, screening and flotation, which are normally carried on at the mine, and activities identified in subsection (i), which are normally carried out at the same location as the smelter. Concentration activities which are not defined in subsection (i) do not require a permit when carried on in isolation, but in view of their very high potential to cause pollution would do so if technically associated with an activity which requires a permit. Pre-treatment processes. Pre-treatment processes such as calcining, roasting or sintering as defined in subsection (i) may be practised in order to alter the chemical or physical form of the concentrate to one most suitable for the smelter, for example by converting sulphides to oxides. Where a pre-treatment process does not involve a process defined in subsection (i) it will not require a Permit unless it is regarded as a technically associated activity, or is an activity described elsewhere in Schedule 1, e.g. Section 4.2 Part A (j) Smelting The smelting stage involves conversion of a metal compound (frequently an oxide or sulphide, or from secondary sources a silicate) into metal. This may be done by displacing the metal from the compound using a more reactive entity, or by electrolysis. Refining Where thermal smelting techniques are used, the metal produced will contain as impurities a proportion of the other metals present in the concentrate. The refining techniques which can be used are dictated by the reactivity of the metal, and also its relative reactivity when compared to the impurity. More reactive metals can be removed by blowing air or oxygen through the melt, thereby oxidising the most reactive components. The unwanted oxide may be removed as a fume with the exhaust gases, as a solid dross or as a slag. Less reactive metals are normally removed by electrolytic processes. In some cases separation can be effected by using the physical properties of the metals, such as the use of distillation to separate cadmium and zinc. Alloying Where allovs are manufactured in furnaces which are technically associated with the smelting operation, they form part of the permitted installation. If the metal is transferred in molten form to the alloying or casting furnaces, then they may be presumed to be technically associated.

INTRODUCTION		TECHNIQUES		EMISSIONS		IMPACT			
IPPC and BAT	Making an application		llations /ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Owing to the need to minimise transport costs, concentration activities are normally carried out near the producing mines. Since very little non-ferrous metal ore is mined in the UK, these activities are not normally found here.

For similar reasons, primary smelting is frequently undertaken in mining areas, and few large primary smelting operations are located in the UK. The exceptions are the lead/zinc smelter at Avonmouth, and the aluminium smelters in Wales, the North of England and Scotland.

Transport costs also influence the location of some electrode manufacturing operations, and anodes for use in aluminium smelting are normally manufactured in dedicated plant adjacent to the smelter. In contrast, graphite electrodes are usually manufactured by specialist suppliers of electrodes.

The techniques appropriate to a smelting operation are restricted by the thermodynamics of the system involved, and influenced by the impurities and contaminants which are present in ores and concentrates to be used.

Most secondary non-ferrous metals installations in the UK depend to a greater or lesser extent upon scrap metal as a feedstock. The nature of this scrap, and in particular the nature and extent of contamination by oils, paint, plastics, other types of scrap and any other contaminant will influence the choice of process and the extent to which pollution abatement measures are required.

Figure 1.2 summarises routes whereby non-ferrous metals are recovered from scrap metal and other secondary raw materials.

Figure 1.2 Activities involved in the production of non-ferrous metals from secondary raw materials including melting and associated operations

Pre-treatment Including removal of organic contaminants (oil, grease, paint and other coatings) by heating in a furnace as defined in Section 2.2 Part B (b) where the activity is associated with an activity defined in Part A (1).



In assessing "effect on pollution" effects associated primarily with the semi-finishing operation should be included. Particular attention should be paid to noise.

INTRODUCTION		TECHNIQUES		EMISSIONS		IMPACT			
IPPC and BAT	Making an application		llations /ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Figure 1.3 summarises the activities involved in manufacturing carbon and graphite electrodes.

Figure 1.3 Manufacture of carbon electrodes Storage, handling, crushing, grinding and classification of coal, coke or petroleum coke. Hot storage of pitch or bitumen. Used electrodes may be cleaned, crushed and recycled. Mixing solids, blending in pitch or bitumen, forming green shaped block. Heat in bake furnace and soak at around 1100 °C to carbonise pitch or tar. For high conductivity electrodes, heat in electrical resistance furnace to 2800°C and soak in order to convert carbon into graphite. (Anodes for use in pre-bake aluminium furnaces are not graphitised.) Graphite electrodes may be machined to shape. All types of electrode may be fitted with metal connectors to aid installation in electrical systems.

Factors to be considered when classifying the production of non-ferrous metals from secondary raw materials

1. Secondary raw materials from which non-ferrous metals are produced are scrap metal, slags, drosses and other residues (such as scrap refractories).

Metal may be produced from these materials by the following activities:

- Pre-treatment (such as de-coating) followed by smelting or melting,
- crushing and screening to remove metallic inclusions from the more friable non-metallic portion, and/or
- smelting to convert compounds such as oxides and silicates into metal.
- 2. Smelting is a complex operation and is included within the flow chart of Figure 1.1 whether or not the secondary raw material has been treated to remove metallic inclusions.
- 3. Where crushing operations for the purpose of removing metallic inclusions from furnace wastes are carried out at the site where they are produced, then they have a clear technical connection with the melting or smelting activity, and in the light of their significance for causing pollution are to be included within the Permit for that installation.
- 4. Where a crushing operation is the first of a sequence of operations intended to produce metal from wastes, then subsequent melting operations are to be included within the Part A (1) Permit by virtue of the rules for interpretation of a "Part A Installation" in Part 3 of the Regulations.
- 5. If no metal is recovered from an activity which involves crushing metallurgical slag, and there is no associated Part A activity, then the installation falls under Section 3.5 Part B (a). (A metallurgical slag is a slag generated by a metallurgical activity, whether or not it contains a recoverable or even detectable amount of free metal.)

Drosses, skimmings, fluxes and slags

When metals melt, oxides tend to form and accumulate on the surface. Involatile impurities also float, and these, together with the oxides, can form a dry dusty layer referred to as **dross**.

To reduce the rate of oxidation, and to ease the removal of drosses, **fluxes** may be added which melt on the metal surface. (Fluxes may also be added with the intention of their reacting with impurities within the metal.)

In some operations the dross (with or without added flux) is removed from the molten metal surface by raking or scraping into dross trays beside the furnace door. The material removed may be referred to as dross, or as **skimmings**.

If sufficient flux has been added to combine with the dross to form a liquid layer on the surface of the metal, this layer is referred to as a **slag**. Slag is removed by tapping from the furnace as a fluid.

INTRODUCTION		TECHNIQUES		EMISSIONS		IMPACT			
IPPC and BAT	Making an application		llations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

General issues

The impact of the activities on the environment may be wider than just the on-site activities. This Note and the Regulations cover issues downstream of the installation such as the final disposal of wastes and waste waters.

Advice on the physical site which is contained within the installation, e.g. split sites, is given in IPPC Part A (1) Installations: Guide for Applicants (see Ref. 6). Operators are advised to discuss this issue with the Regulator prior to making their application.

Where associated activities are carried out in conjunction with the main activities and are not covered in this Guidance Note (e.g. a pre-heating furnace may be classified as a combustion plant), reference should be made to:

- other relevant IPPC Guidance Notes,
- relevant Guidance Notes issued under EPA 90, e.g. Refs. 22, 23
- where appropriate, the Secretary of State's Guidance for Local Authority Air Pollution Control in Northern Ireland this guidance is produced by the Department of the Environment.

INTRODUCTION		TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application		llations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

1.3.1 The non-ferrous metals BREF document

The European Commission Reference Document on Best Available Techniques in the Non-Ferrous Metals Industries (the BREF Note, see Ref. 3) covers the production of metals from both primary and secondary raw materials under the chapter headings below:

Chapter 3 – Copper (including Sn and Be) and its alloys

- 4 Aluminium
- 5 Zinc, lead and cadmium (+ Sb, Bi, In, Ge, Ga, As, Se, Te)
- 6 Precious metals
- 7 Mercury
- 8 Refractory metals
- 9 Ferro alloys
- 10 Alkali and alkaline earth metals
- 11 Nickel and cobalt
- 12 Carbon and graphite

Associated foundry and casting operations are covered in a separate BREF Note (Ref. 4). Carbon and graphite production is included because of its close technical association with metals manufacture.

1.3.2 BREF document contents

In the BREF document, information is presented in 13 chapters covering: general information about the industry in Chapter 1, common processes in Chapter 2 and then metallurgical production processes for ten groups of metals in Chapters 3 to 12 as shown above. Chapter 13 presents the conclusions and recommendations. Annexes covering costs and international regulations are also included. The common processes in Chapter 2 are divided as follows:

- Use of the chapter complex installations.
- Use and reporting of emission data.
- Management, design and training.
- Receipt, storage and handling of raw materials.
- Pre-processing and pre-treatment of raw materials and transfer to production processes.
- Metal production processes furnace types and process control techniques.
- · Gas collection and air abatement techniques.
- Effluent treatment and water re-use.
- Minimisation, recycling and treatment of process residues (including by-products and waste).
- Energy and waste heat recovery.
- Cross-media issues.
- Noise and vibration.
- Odour.
- Safety aspects.
- Decommissioning.
- BAT conclusions for some common processes such as material handling and storage, process control, gas collection and abatement, dioxin removal, sulphur dioxide recovery, mercury abatement and effluent treatment/water re-use.

Each of Chapters 3 to 12 includes sections:

- (1) applied processes and techniques,
- (2) present emission and consumption levels,
- (3) techniques to consider in the determination of BAT, and
- (4) BAT conclusions.

These sections give a more detailed description of the processes, including process diagrams in the sections on applied processes.

INTRODUCTION		TECHNIQUES			EMIS	SIONS	IMPACT		
IPPC and BAT	Making an application		llations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

1.4 Review periods

Permits can be reviewed or varied at any time. However, the PPC Regulations impose a requirement on Regulators to review Permits in certain specific circumstances such as where the pollution caused by the installation is of such significance that the existing emission limit values need to be revised or new limits set.

In addition, Regulators are required to review the conditions of Permits "periodically". The Government stated in its third consultation paper (England, Wales and Scotland) on the implementation of IPPC, that the new sector-specific IPPC Technical Guidance Notes would provide guidance on appropriate review periods for each sector. These would take into consideration guidance on the relevant criteria, to be provided by the Government. Examples of the likely relevant criteria for setting these review periods are "the risk and level of environmental impacts associated with the sector" and "the cost to the regulators and regulated industry of undertaking the reviews".

The Regulators consider that at the present time, having regard to those criteria, it is in fact appropriate to set indicative minimum review periods which differ only between those sectors which have been subject to integrated permitting (i.e. IPC or Waste Management Licensing) and those which have not. It is therefore proposed that Permit conditions should normally be reviewed on the following basis:

- for individual activities **not** previously subject to regulation under IPC or Waste Management Licensing, a review should normally be carried out within four years of the issue of the IPPC Permit;
- for individual activities previously subject to regulation under IPC or Waste Management Licensing, a review should normally be carried out within six years of the issue of the IPPC Permit.

This means that activities/installations not currently in IPC or Waste Management Licensing will be initially reviewed within four years and thereafter within six years.

This period will be kept under review and, if any of the above factors change significantly, may be shortened or extended.

1.5 Key issues for the sector

Potentially polluting emissions to air

Primary extraction processes which use sulphur containing ores generate substantial quantities of sulphur dioxide.

Most ores and concentrates contain a range of other elements, many of which can have a significant impact on the environment if allowed to escape, for example substances in list I and II of the Hazardous Substances Directive.

Scrap and other secondary raw materials may be contaminated with chlorinated organic compounds such as plastics, lubricants or solvents. These are a potential source of dioxins.

Fugitive emissions to air

Pyro-metallurgical processes subject plant to high temperatures and aggressive atmospheres. In addition the need to move heavy items such as ladles has a potential to cause physical damage to furnaces and ducting. The risk of fugitive emissions into the air is such that it warrants separate consideration as a key issue in any application.

Energy requirements

All pyro-metallurgical processes require significant energy input in order to melt the metal being processed. Primary extraction processes in particular require a high energy input to extract metals from their ores and to purify them.

Accident risk

Particular attention should be paid to the storage and use of liquefied gases such as oxygen, chlorine and LPG (see Section 2.8).

Loss of electrical supplies to control systems and to pollution abatement systems may lead to uncontrolled discharges to air and water.

Flooding, whether caused by rainfall or as a consequence of fire fighting activities, may give rise to uncontrolled emissions to water.

INTRODUCTION		TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application		llations /ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Noise

Many potential sources of noise may be found in non-ferrous metallurgical processes. Particular attention should be paid to:

- movement and storage of scrap;
- location and sound insulation of large fans and air filtration systems;
- rolling mills;
- · casting installations, especially billet casters;
- internal transport.

Long-distance and transboundary pollution

For the largest processors an assessment will be required of the potential for long-distance environmental impacts.

Monitoring

Many non-ferrous metals have the potential to release long-term contaminants, such as heavy metals. At each site there should be an assessment of the potential for releases and possible impacts. On the basis of such an assessment, a decision shall be made concerning the need for an environmental monitoring programme (see Section 2.10).

Solid waste recovery, recycling and disposal

The major contribution of the secondary non-ferrous metals industry to recycling metallic wastes is noted; however, the industry itself has significant waste arisings (see Section 2.6). Of particular interest are:

- slags and drosses from melting operations;
- sand from casting operations; and
- insulation and refractory materials from maintenance operations.

Site restoration

Areas for consideration concerning remediation will include:

- storage areas used for slags, drosses and other residues, and
- fuel storage areas.

The majority of non-ferrous metals processes are located on sites which have been used for industrial activities for many years. In many cases, mining, previous waste disposal activities or earlier metallurgical processes will have contaminated the ground. Since such contamination may be confused with potential future contamination arising from activities permitted under IPPC, where past contamination is likely it will be necessary to assess the degree of contamination as a baseline against which future operations can be judged.

Summary

The main issues for the production processes for each of the groups of metals comprise the potential for fugitive emissions and the following components:

- For the production of copper: SO₂, dust, metal compounds, organic compounds, waste water (metal compounds), residues such as furnace linings, sludge, filter dust and slag. Dioxin formation during treatment of secondary copper materials is also an issue.
- For the production of aluminium: fluorides (incl. HF), dust, metal compounds, SO₂, COS, PAH, VOCs, greenhouse gases (PFCs and CO₂), dioxins (secondary), chlorides and HCl. Residues such as bauxite residue, spent pot lining, filter dust and salt slag and waste water (oil and ammonia).
- For the production of lead, zinc and cadmium: dust, metal compounds, VOCs (including dioxins), odours, SO₂, other acid gases, waste water (metal compounds), residues such as sludge, the iron-rich residues, filter dust and slag.
- For the production of precious metals: VOCs, dust, metal compounds, dioxins, odours, NO_x, other acid gases such as chlorine and SO₂. Residues such as sludge, filter dust and slag and waste water (metal compounds and organic compounds).
- For the production of refractory metals, hard metal powder and metal carbides: dust, solid hard metal and metal compounds, waste water (metal compounds), residues such as filter dust, sludge and slag. Process chemicals such as hydrogen fluoride (HF) are used for processing tantalum and niobium and are highly toxic. This needs to be taken into account in the handling and storage of these materials.
- For the production of ferro alloys: dust, metal compounds, CO, CO₂, SO₂, energy recovery, waste water (metal compounds), residues such as filter dust, sludge and slag.

INTRODUCTION		TECHNIQUES			EMIS	SIONS	IMPACT		
IPPC and BAT	Making an application		Illations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

- For the production of alkali and alkaline earth metals: chlorine, HCl, dioxin, SF₆, dust, metal compounds, CO₂, SO₂, waste water (metal compounds), residues such as sludge, aluminate, filter dust and slag.
- For the production of nickel and cobalt: VOCs, CO, dust, metal compounds, odours, SO₂, chlorine and other acid gases, waste water (metal compounds and organic compounds), residues such as sludge, filter dust and slag.
- For the production of carbon and graphite: PAHs, hydrocarbons, dust, odour, SO₂, waste water prevention, residues such as filter dust.

The "Present Emission and Consumption Levels", "Techniques to Consider in the Determination of BAT" and the "BAT Conclusions" contained in all of the chapters of the BREF should be consulted in particular for a more complete understanding of the key issues.

INTRO	INTRODUCTION		TECHNIQUES			EMI	SSIONS	IMPACT	
IPPC and BAT	Making an application		llations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

1.6 Summary of releases for each sector

The potential releases to the environment from the various processes are summarised in the following tables.

More details are provided in the BREF in sub section 2 of Chapters 3 to 12.

		1			1	1	1							
	Raw material storage and handling	Pre-treatment, e.g. sintering or de-coating	Copper process	Aluminium processes	Lead, zinc and cadmium processes	Precious metal processes	Refractory metal processes	Ferro alloy processes	Alkali and alkaline metal processes	Nickel and cobalt processes	Carbon and graphite production	Residue handling, storage and re-use	Air abatement plant	Effluent plant
	Raw	Pre-t de-c	Copl	Alun	Leac	Prec	Refn	Ferr	Alka	Nick	Carb	Resi	Air a	Efflu
VOCs		A	A	A	A	A	A	A	A	A	A		A ^c	
Oxides of sulphur		A	A	A	A	A	A	A	A	A	A			
Oxides of nitrogen		А	А	А	А	А	А	А	А	А	А			
Particulates and PM 10	А	А	А	А	А	А	А	А	А	А	А	А	А	
Carbon monoxide			A ^a							А				
Mercury	AW	А	А		Α									W
Cadmium	W		Α		Α									W
Other heavy metals	Α	А	А	А	Α	Α	А	А	А	А	А	А	А	W
PFCs				AW										
Hydrogen chloride or fluoride				А		Α								
Dioxins and furans		А	А	А	Α	Α	А	А	А	А	А	А	А	
Ammonia				AW		AW	AW							
Cyanide				W		AW								
Suspended solids	W											W		W
Oil	W											W		W
Stibine/arsine					AW									
Tars and/or PAH				AW							AW			
Sulphides and mercaptans						Α								W
Chlorine				A ^b	Α	А	А		А					
Metallurgical slags, drosses and furnace residues			L	L	L	L	L	L	L	L				
Dust and dusty residues	L	L										L	L	
KEY	A – re	elease to	o air, M	/ – relea	ase to w	vater, L	– relea	se to la	nd					

Notes:

^a From shaft furnaces designed to operate with a reduced atmosphere in the melting zone.

^b From certain refining processes.

^c From afterburners.

NB – releases to air usually result in a subsequent, indirect emission to land and can therefore affect human health, soil and terrestrial ecosystems.

INTRODUCTION		TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application		Illations vered	Review periods Ke		y issues	Summary of releases	Sector overview	Economic aspects

1.7 Overview of the activities in this sector

Estimated number of non-ferrous metals installations in England and Wales which will require permits under Part A (1)

	Primary	Secondary
Copper	0	15
Aluminium	3	35
Zinc, lead, cadmium	1	35 ^a
Precious metals	1	10
Alkali and alkali earth metals	2	5
Others	1	10
Totals	6	110 ^a

^a This total includes an estimate of at least 10 activities with trivial potential for environmental impact as assessed under EPA 90, but which are identified under Part A (1) (d).

The structure of the non-ferrous metals industry within the United Kingdom is steadily changing with the final decades of the 20th century witnessing an increasing dependence upon imports for supplies of metal. Recent developments suggest that this trend will not be reversed.

Secondary smelting and recovery from scrap is also under pressure with the number of installations falling steadily.

The industry, nevertheless, remains an important sector of the economy with its products being essential to the generation and supply of electricity, plumbing and house construction, the aircraft and engineering industry and also the chemical industry.

Although the computing, electronic and telecommunications industries use smaller gross tonnages, their economic significance is high.

A primary factor influencing the above changes is that suitable ore deposits in the United Kingdom have now been largely exhausted, and that transport costs make movement of large tonnages of ores or concentrates less economic.

Recycling is of prime importance in this industry and scrap and other residues are of significant value. The energy requirement for recovering high-grade metals and alloys from scrap is much lower than for manufacturing similar alloys from ores or concentrates.

The industry is a major consumer of energy, with the potential for significant releases to land and to air.

1.7.1 Processes used in the manufacture of non-ferrous metals

Primary processes

- concentration and purification of ores
- conversion of ores into forms suitable for smelting, e.g. sulphide ores to oxides
- smelting
- refining
- alloying

Secondary processes

- scrap pre-treatment
- smelting, Melting and refining
- alloying

INTRODUCTION		TEC	TECHNIQUES			SSIONS	IMPACT		
IPPC and BAT	Making an application		Illations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

(a) Smelting processes

- Fused salt electrolysis Strongly electropositive metals such as the alkali and alkali earth metals can only be manufactured by electrolysis of fused salts, usually chlorides. The metals themselves are intensely reactive when molten, as is the by-product chlorine gas. The design of the cells must ensure that molten metal does not come into contact with air, and that all chlorine is removed and liquefied for sale.
- Electrolysis of oxides dissolved in fused salts Aluminium cannot be smelted by heating its oxide with carbon, but instead is manufactured by electrolysis of its oxide when dissolved in molten fused fluorides approximating to the composition of natural cryolite.

The oxygen produced as a by-product consumes the carbon anode to generate carbon monoxide, which in turn burns off at the surface of the electrolyte. This continuous flow of gas through the hot electrolyte generates an emission of volatile fluorides. This is accompanied by sulphur compounds and other decomposition products from the anodes. The regular disposal of old cell structures known as spent pot lining can have a major environmental impact.

 Displacement reactions – The most widely used process involves heating an oxide of the metal with carbon to yield carbon monoxide and metal. This process forms the basis for primary smelting of zinc, lead and tin.

Hydrogen or a reactive metal, e.g. aluminium used to smelt chromium, may also be used.

Air pollution problems are potentially highly significant for larger installations as also may be waste disposal issues arising from slag generation.

(b) Smelting and melting

For secondary raw materials smelting and melting are often carried together in the following stages.

Fossil fuel fired furnaces – These include rotary furnaces and static bath furnaces.

Rotary furnaces are difficult to seal and consequently fugitive emissions to air are hard to eliminate completely. They are, however, very efficient at mixing metal and flux, and are thus the favoured furnace for handling low grades of scrap.

Static bath furnaces can be designed to handle larger items of scrap, and are capable of being sealed more effectively than rotary furnaces.

For all fossil fuel fired furnaces, gas is the preferred fuel as it does not carry any risk of ground contamination due to spillage, and is also easier to burn efficiently.

Oxygen enrichment and oxyfuel burners should also be considered as the volume of the products of combustion is markedly reduced by the elimination of nitrogen. The benefit in terms of cost and fuel efficiency is off set to a degree by the costs arising from oxygen manufacture, but significant environmental benefit can arise as a consequence of the much smaller volumes of exhaust gas.

• Pre-treatment and cleaning operations for scrap, drosses, slags and other residues – Drosses, slags and other furnace residues are normally prepared for recycling by first crushing or milling to free the metallic inclusions, then screening. The metal content concentrates in the large fraction.

These activities may be carried on as part of a waste minimisation activity at the furnace, as a pretreatment, or as a stand-alone, independent operation.

In all cases, the potential for dust emissions is high. In some cases, e.g. aluminium drosses, there is a significant potential for odour, and for water pollution should drosses become wet – whilst in a few, e.g. zinc drosses, the gases given off by wet dross are hazardous to health.

Minimising the non-metallic content of the furnace charge reduces the amount of solid waste generated, and also the specific energy consumption.

 Scrap metal from cutting, milling and similar operations – These form a valuable feed for secondary smelters, particularly where proper segregation is practised at source, and crosscontamination with other swarf avoided (see Ref. 9 – Environmental Technology Best Practice Guide GG264).

They are, nevertheless, likely to be contaminated with cutting oils and require cleaning before being charged into a melting furnace.

INTRODUCTION		TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application		llations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

The extent to which cleaning is required will depend upon the design of the melting furnace, some of which can include combustion systems designed to cope with oily charges. In most cases, it will be necessary to remove most of the oil by:

- draining in designated de-oiling areas;
- centrifuging;
- heating to drive off volatile components, followed by destruction of VOCs in an after-burner.

These techniques may be used separately, or in combination, but whichever de-oiling system is used, it must be considered in combination with the furnace with which it is associated.

At present, chlorinated cutting oils remain in widespread use within the United Kingdom. Consequently, furnaces which handle oil swarf must be regarded as potential sources of dioxins.

 Scrap metal from extrusion and rolling operations – These will be contaminated to a greater or lesser extent by soaps and lubricants used in the forming operations. This contamination can be significant in certain cases such as tube ends from cold drawing benches.

The physical nature of the waste makes pre-cleaning more difficult, and greater emphasis must be placed upon the design and operation of melting furnaces.

Recovered scrap – This comes from a wide variety of sources. Some, such as motor vehicle
engine and transmission parts, may be so grossly contaminated that washing represents a
practicable means of pre-treatment.

Where recovered scrap is to be processed, careful attention should be paid to those aspects of the EMS which ensure that only material that is compatible with the furnace is sent forward for melting.

(c) Melting processes

Melting processes for clean metal such as ingots or very clean scrap can use the furnaces described above but the most frequently encountered is the following.

• Electric induction furnaces – These are widely used to melt metal, with feedstock ranging from pure ingot to scrap and this includes iron components recovered in a rodding plant of a primary aluminium smelter. They have the advantage of being relatively simple to enclose, thus minimising fugitive emissions to air. They can also be designed for complete enclosure, permitting the use of controlled atmospheres above the melt.

By virtue of the presence of electrical conductors in their construction, they are more vulnerable to corrosion and erosion by slags than directly fired furnaces, and are most suitable for making products from clean raw materials.

The following tables are derived from the Non-Ferrous Metals BREF Note.

Furnace	Metals used	Material used	Comment
Steam coil dryer Fluid bed dryer Flash dryer	Cu and some others	Concentrates	Drying operations
Rotary kiln	Most metals for drying; fuming ZnO; calcining alumina, Ni and ferro alloys; burning of photographic film for precious metal production; de-oiling Cu and Al scrap	Ores, concentrates and various scrap and residues	Drying, calcining and fuming applications
Fluidised bed	Copper and zinc; Al ₂ O ₃	Concentrates, AI(OH) ₃	Calcining and roasting
Up draught sintering machine	Zinc and lead	Concentrates and secondary	Sintering
Down draft sintering machine	Zinc and lead	Concentrates and secondary	Sintering
Steel belt sintering machine	Ferro alloys, Mn, Nb	Ore	Other applications possible
Herreshoff	Mercury, molybdenum (rhenium recovery)	Ores and concentrates	Roasting, calcining

Drying, roasting, sintering and calcining furnaces

INTRODUCTION		TECHNIQUES			EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application		llations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

Smelting and refining furnaces

Furnace	Metals used	Material used	Comment
Enclosed refractory lined crucibles	Refractory metals, special ferro alloys	Metal oxides	
Open pit	Refractory metals, special ferro alloys	Metal oxides	
Baiyin	Copper	Concentrates	
Electric arc furnace	Ferro alloys	Concentrates, ore	
Contop/cyclone	Copper	Concentrates	
Submerged electric arc furnace	Precious metals, copper, ferro alloys	Slag, secondary materials, concentrates	For the production of ferro alloys, the open, semi-closed and closed types are used
Rotary	Aluminium, lead, copper, precious metals	Scrap and other secondary, blister copper	Oxidation and reaction with substrate
Tilting rotary furnace	Aluminium	Scrap and other secondary	Minimises salt flux use; may have restrictions on size of material
Reverberatory	Aluminium, copper, others	Scrap and other secondary, black copper	Smelting of Cu concentrates elsewhere in the World
Vanyucov	Copper	Concentrates	
ISA smelt/ausmelt	Copper, lead	Intermediates, concentrates and secondary materials	
QSL	Lead	Concentrates and secondary	
Kivcet	Lead, copper	Concentrates and secondary	
Noranda	Copper	Concentrates	
El Teniente	Copper	Concentrates	
TBRC TROF	Copper (TBRC), precious metals	Most secondary incl. slimes	
Mini smelter	Copper/lead/tin	Scrap	
Blast furnace and ISF	Lead, lead/zinc, copper, precious metals, high carbon ferro-manganese	Concentrates, most secondary	For ferro-manganese production, it is only used together with energy recovery
Inco flash furnace	Copper, nickel	Concentrates	
Outokumpu flash smelter	Copper, nickel	Concentrates	
Mitsubishi process	Copper	Concentrates and anode scrap	
Peirce Smith	Copper (converter), ferro alloys, metal oxide production	Matte and anode scrap	
Hoboken	Copper (converter)	Matte and anode scrap	
Outokumpu flash converter	Copper (converter)	Matte	
Noranda converter	Copper (converter)	Matte	
Mitsubishi converter	Copper (converter)	Matte	

INTRO	DUCTIO	N	TEC	CHNIQUES	S	EMIS	SSIONS	IMPACT		
IPPC and BAT	Making an application		Illations vered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects	

Melting furnaces

Furnace	Metals used	Material used	Comment
Induction	Most	Clean metal and scrap	Induced stirring assists alloying; vacuum can be applied for some metals
Electron beam	Refractory metals	Clean metal and scrap	
Rotary	Aluminium, lead	Various scrap grades	Fluxes and salts used for complex matrices
Reverberatory	Aluminium (primary and secondary)	Various scrap grades	Bath or hearth configuration can vary; melting or holding
Contimelt	Copper	Copper anode, clean scrap and blister copper	Integrated furnace system
Shaft	Copper	Copper cathode and clean scrap	Reducing conditions
Drum (Thomas)	Copper	Copper scrap	Melting, fire refining
Heated crucibles (indirect kettles)	Lead, zinc	Clean scrap	Melting, refining, alloying
Direct heated crucibles	Precious metals	Clean metal	Melting, alloying

Hydrometallurgical processes are also used. Acids and alkalis (NaOH, sometimes also Na₂CO₃) are used to dissolve the metal content of a variety of calcines, ores and concentrates before refining and electro-winning. The material to be leached is usually in the form of the oxide, either as an oxidic ore or as an oxide produced by roasting. Direct leaching of some concentrates or mattes is also performed at both elevated and atmospheric pressure. Some copper sulphide ores can be leached with sulphuric acid or other media, sometimes using natural bacteria to promote oxidation and dissolution, but very long residence times are used.

Air, oxygen, chlorine or solutions containing ferric chloride can be added to leaching systems to provide the appropriate conditions for dissolution. The solutions that are produced are treated in a number of ways to refine and win the metals. Common practice is to return the depleted solutions to the leaching stage, where appropriate, to conserve acids and alkaline solutions. Nitric acid and other reactive reagents are used to recover precious metals.

Overview of chemical treatment methods for some gaseous components

Process/reagent used	Component in off-gas	Treatment method
Use of arsenic or antimony oxide (refining of Zn/Pb)	Arsine/stibine	Permanganate scrubbing
Pitch etc.	Tars and PAH	After burner, condenser and EP or dry absorber
Solvents, VOCs	VOC, odour	Containment, condensation; activated carbon, bio-filter
Sulphuric acid (+ sulphur in fuel or raw material)	Sulphur dioxide	Wet or semi-dry scrubber system; sulphuric acid plant
Aqua regia	NOCI, NO _x	Caustic scrubber system
Chlorine, HCl	Cl ₂	Caustic scrubber system
Nitric acid	NO _x	Oxidise and absorb, recycle, scrubber system
Na or KCN	HCN	Oxidise with hydrogen peroxide or hypochlorite
Ammonia	NH ₃	Recovery, scrubber system
Ammonium chloride	Aerosol	Recovery by sublimation, scrubber system
Hydrazine	N ₂ H ₄ (possible carcinogen)	Scrubber or activated carbon
Sodium borohydride	Hydrogen (explosion hazard)	Avoid if possible in PGM processing (especially Os, Ru)
Formic acid	Formaldehyde	Caustic scrubber system
Sodium chlorate/HCI	Cloxides (explosion hazard)	Control of process end-point

Process descriptions and diagrams are given in the BREF in subsections 1 and 2 of each the Chapters 3 to 12.

INTRO	DDUCTION TEC			CHNIQUES	S	EMIS	SSIONS	IMPACT	
IPPC and BAT	Making an application		llations /ered	Review periods	Ke	y issues	Summary of releases	Sector overview	Economic aspects

1.8 Economic aspects for each sector

Most non-ferrous metals are international commodities, with prices more heavily influenced by supply and demand than by the cost of production. The major metals (aluminium, copper, lead, nickel, tin and zinc) are traded on one of two futures markets, the London Metal Exchange and Comex in New York. The collectively named "minor" metals have no central market-place; price levels are imposed by either producers or merchants trading in the free markets. In most applications non-ferrous metals are in competition with other materials, notably ceramics, plastics and other ferrous and non-ferrous metals. The influence upon world prices is illustrated in Figure 1.4, which shows the variation in price of aluminium alloy between January 1998 and December 1999, the information being published the by London Metal Exchange. During this period the variation in production costs was not significant.

The profitability of each metal or metals group, and thus the economic viability of the industry, varies, both absolutely and on a short-term basis, depending upon the current metal price and a wide range of other economic factors. The general economic rule applies however, i.e. that the nearer a material or product approaches global market conditions and international commodity status, the lower is the return on the capital invested.



Figure 1.4 Aluminium alloy price variation

Annex 1 of the BREF provides some specific cost data for plant and abatement equipment used in this sector.

It is effective to consider normal investment and expenditure cycles when improvement programmes are being planned. Many win–win solutions are reported in the Non-Ferrous Metals BREF in the sections dedicated to "Techniques to Consider in the Determination of BAT" and these techniques can reduce environmental impact and increase metal yield and efficiency.

INTROD	UCTIO	TEC	HNIQ	JES	E	VISSIO	NS	I	MPAC	Т
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
	inputs	abatement	water							135065
	2 '	TECHNI	QUE	S FO	R PO	LLUTI	ON C	ONTR	OL	
	This s	ection summa	rises in t	he outline	od B∆T b	OVES				
BAT boxes to help in		at is required				0,03,				
preparing applications	• the	e indicative BA plication will b	T require		e. what is	BAT in most	t circums	tances) agai	nst which	the
		top of each B								
	Regul Regul	igh referred to ations and req ations (see Ap dwater Regula	uirements	s of other for equiva	Regulational Regulational Regulation Regulat	ons (such as slation in Sco	the Was	te Managem d Northern Ir	ent Licen	sing
Indicative BAT requirements		e it has been p tive requireme						n what will no	rmally be	BAT, the
		ou propose to , if this is not o							be how yo	ou will do
	de	ou propose to partures may	be stricter	or less s	trict than	the indicativ			our propo	sal. Such
	• Str	ricter proposal	-				ublication	of the guida	nco.	
	-		-			-		kes higher sta		racticable;
		the local er		-		-		J		,
	loc rec is l	ss strict propo cal environmer quirement, but based. In sucl chniques for or	nt. For ex using diff h a case i	ample, yo erent pla t may imp	ou may o nt or proc bose a dis	perate to a si cesses from t sproportionat	tandard t hat upon	hat is very closed which the in	ose to an dicative r	indicative equirement
	installa	er cases, the n ation-specific b nce at all.								
Justifying	Wheth	ner you are								
proposals	-	tifying departu				-				
		sessing option					•		•	
		veloping propo		· ·		,				0
	depen where implica analys	sts and benefi ds on the envi the options av ations are a m sis of the costs t in the IPPC H	ronmenta vailable w ajor factor and bene	I signification ould lead r) it will be efits of op	ance of th to signifi e necessa tions. Th	e matter in q cantly differe ary to develo ne Regulator	uestion. ent enviro p propos s methoo	In the more onmental effe als through a dology for su	complex cts, or wh a more de ch assess	cases (e.g. here the cost stailed sments is
	examp	ny situations, h ble, where an i lly minor additi	ndicative	standard	is inappr	opriate for ol	ovious te	chnical reaso	ons, or wh	nere there
Prevention is	-	oonding to the	-							
the priority		a first principl the possibility	of preve r	ting the	release c	of harmful sul	ostances		ation has	been given
	-	substituting n preventing re		-	1 - C			-		
	-	preventing wa					<u>د.د.ی</u> , ۱۵			
		ily where that ich may cause	is not pra	-		-	ciple be	adopted of re	educing e	missions
	Furthe	er explanation iques in greer	of the req			-				

INTRODUCTION TECHNIQUES				JES	EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.1 **Management techniques**

Within IPPC, an effective system of management is a key technique for ensuring that all appropriate pollution prevention and control techniques are delivered reliably and on an integrated basis. The Regulators strongly support the operation of environmental management systems (EMSs). An Operator with such a system will find it easier to complete not only this section but also the technical/regulatory requirements in the following sections.

The Regulators recommend that the ISO 14001 standard is used as the basis for an environmental management system. Certification to this standard and/or registration under EMAS (EC Eco Management and Audit Scheme) (OJ L168, 10.7.93) are also strongly supported. Both certification and registration provide independent verification that the EMS conforms to an assessable standard. EMAS now incorporates ISO 14001 as the specification for the EMS element. For further details about ISO 14001 and EMAS contact the British Standards Institute (BSI) and the Institute of Environmental Management and Assessment (IEMA) respectively.

The steps required in this and subsequent sections may help the Operator to make good any shortfalls in their management system. An effective EMS will help the Operator to maintain compliance with regulatory requirements and to manage other significant environmental impacts. While the requirements below are considered to be BAT for IPPC, they are the same techniques as required in a formal EMS and are also capable of delivering wider environmental benefits. However, it is information on their applicability to IPPC that is primarily required in this application.



With the application the Operator should:

1. Describe their management system to demonstrate how it meets the "Requirements for an effective management system" below. The description should make clear who holds responsibility for each of the requirements. The second column explains where in the application the response to each requirement is best dealt with to avoid duplication. Copies of all procedures are not needed, but examples may be included in your application.

If you are certified to ISO 14001 or registered under EMAS (or both) you may provide a statement derived from certification records/assessments to support your application.

Further specific management procedures are dealt with under the appropriate section on the remainder of the document. It is recommended that you understand all the requirements of the application before completing this section, as many management issues are dealt with in other sections.

2. The type of management system employed will depend upon the scale and complexity of the operations undertaken. The Operator should demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements below or by justifying departures (as described in Section 1.2 and in the Guide to Applicants) or alternative measures.

Indicative BAT requirements

The Operator should have a management system in place for the activities which delivers the requirements given in the first column below. The development of any aspects of the management system not already in place should be completed within the timescale given in Section 1.1.

Re	quirement for an effective management system	How delivered for IPPC			
1.	Clear management structure and allocated responsibilities for environmental performance, in particular meeting the aspects of the IPPC Permit	Describe in this section who has allocated responsibilities			
2.	Identification, assessment and management of significant environmental impacts	By responding to the requirements in Section 4.1 in the application			
3.	Compliance with legal and other requirements applicable to activities impacting on the environment	Compliance with the Permit satisfies this requirement			

Non-Ferrous Metals

BAT for

BREF

2.3

Section

techniques

INTRODU	JCTIO		CHNIQ	UES	FI	MISSION	NS		MPAC	т
	Materials									Installation
Management	inputs	abatemen		Waste	Energy	Accidents	Noise	Monitoring	Closure	issues
BAT for management techniques (cont.)		requirement performance	and targets s and contine	s to preve nually imp	nt pollutio prove env	on, meet lega ironmental	al s F ii	The applicant signed copy of policy stateme nstallation. T nake proposa	f their envent application for the application of t	vironmental able to the ant should
		Environmei implement					e T ii	rake proposa each of Sectio These proposa ncorporated v mprovement p	ons 2.2 to als may b vithin the	2.12. e Permit
		Establish o minimise si	gnificant e	environm	i	By responding n Sections 2.2 2.12 in the app	2 to 2.7, 2			
		Preventive plant and e				or relevant ing and revie		Describe syste procedures in		List
	8.	Emergency	planning a	and accio	lent prev	vention		By responding		
	9.	Monitoring	and measu	uring per	formanc	e	C	Describe in thi	is Sectior	1
		establish an	d maintain cators to er	a progran	nme to m	erformance a easure and nprovement o				
	10.	Monitoring	and contro	ol system	IS			By responding		
BREF Chapter		• to ensu	re that the i	nstallatior	n function	is as intende	d; ⁱⁱ	n Section 2.10	in the a	pplication
2.17.2		 to detect 	t faults and	l unintend	led opera	tions;				
		• to detec	t slow char	nges in pla	gger					
			ive mainter	nance						
	11.	Training						o be describe		
		Provision of relevant stat equipment a	f (including	contracto	ors and th	nose purchas	ing t	confirming tha he areas cove o 2.3 and 2.5	ered by S	ections 2.2
		require	d for each j	ob;		npetencies				
			ess of the r activity and			ons of the Pe s;	ermit			
						ntal effects fr al circumstan				
		•	ion of accid hen accide			nd action to b cur;	e			
		Expertise re out. Howev whom the in sufficient qu	quired dep er, both teo stallation's alifications may be ass	ends on th chnical an complian , training a sessed ag	he activiti d manag ce deper and expe	raining recom- es being carri- erial staff upo- ids need rience for the industry sec	ried on eir			
						ents of actua	al or [Describe in thi	is Sectior	1
		potential no	-		-					
		Actions take to operation		ise, and a	about pro	posed chang	es,			
	13.	Auditing					[Describe in thi	is Sectior	1
			e being carr s. All of the	ied out in ese requir	conformi	to check that ty with these should be				
										Cont.

INTRODU		N TEO		UES	E	MISSION	NS	1	MPAC	т	
	Materials	Activities &	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation	
management	inputs	abatement	water	Waste	Energy	recidents	Noise	Monitoring	Closure	issues	
BAT for											
management techniques (cont.)	14.	Corrective a recurrence Define respo	nsibility an	d authori	ty for han	dling and	i: 2	Describe in this dealt with fo 2.2 to 2.3 and	r each of	Sections	
		investigating any impacts corrective an	caused an	d for initia	ating and		te d	ppropriate			
	15.	Reviewing a	nd reporti	ng envir	onmenta	l performan	се				
		Senior mana and ensure a ensure that p remains rele programmes	oppropriate olicy comr vant. Revi	action ta nitments ew progre	ken wher are met a	e necessary ind that polic	to y	Describe in thi	s Sectior	1	
		Incorporate e aspects of th IPPC, in part	e business					Describe in thi	s Sectior	1	
		• the cont	ol of proce	ess chang	e on the	installation;					
			nd review pital projec		cilities, er	ngineering ar	nd				
		 capital a 									
			ation of res								
			and scheo	-	tal aanaa	to into norma					
		operatin	g procedui		tai aspec	ts into norma	1				
		•	ng policy; ng for envi	ronmonto	l conto o	nainat tha					
			involved ra			•					
		Report on er results of ma audit cycle),	nagement					his will becor equirement	ne a Peri	nit	
		3 <i>y</i> ·	on require	d by the F	Regulator	; and	0	Describe in thi	s Sectior	ı	
			es and targ			tem against nned		Describe in thi	s Sectior	1	
		Report extern statement	nally, prefe	rably via	public en	vironmental					
	16.	Managing d	ocumenta	tion and	records						
		List the core responsibilitie documentatio update docum	es, procedu on in order	ures, etc.)) and links	s to related		Describe in thi	s Sectior	1	
		Describe how and reviews					dits				

			HNIQ	JES	EMISSIONS			IMPACT		
Management	Materials inputs	Activities/ abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Selection of raw materials	2.2	Materials inputs
	use a	section covers the use of raw materials and water and the techniques for both minimising their nd minimising their impact by selection. Depending upon furnace design, pre-treatment of raw rials may be required before melting.
	As a g to:	general principle, the Operator should demonstrate the steps which have been, or may be, taken
Reduce	• re	duce the amount of flux required by improving the quality of raw materials; ubstitute less harmful materials or those which can be more readily abated and when abated lead
Substitute Understand	to	substances which in themselves are more readily dealt with;
	• ur	nderstand the fate of by-products and contaminants and their environmental impact (Section 4).
		1 Raw materials selection and pre-treatment
		section looks at the selection of raw materials used while Section 2.2.2 describes the techniques nimise their use.
		lication Form stion 2.2 (part 1) Identify the raw and auxiliary materials, other substances and water that you propose to use.
BAT for selection of raw materials	With	the application the Operator should:
BREF Sections:		Supply a list of the materials used, which have the potential for significant environmental impact, including:
2.4 and 2.5		 the chemical composition of the materials where relevant; the guartition used;
		the quantities used;the fate of the material (i.e. approximate percentages to each media and to the product);
	•	 environmental impact where known (e.g. degradability, bioaccumulation potential, toxicity to relevant species);
	· ·	 any reasonably practicable alternative raw materials which may have a lower environmental impact (the substitution principle).
		A suitable template is included in the electronic version of this document.
	1	Generic information about materials, and grouping information of those of a similar type, is normally adequate rather than listing every commercial alternative used. A common-sense approach to the level of detail should be used; ensuring that any material could have a significant effect of the environment is included. Product data sheets should be available on-site.
	t	Justify the continued use of any raw material for which there is a less harmful alternative, and that the raw material selection is therefore BAT. Justification may be based on effect upon product quality; security of supply; use of a material which would otherwise be disposed of as a waste; lower energy consumption. Economic factors may also be considered.
		For existing activities, identify shortfalls in the above information, e.g. the environmental impact of certain substances, which the Operator believes require longer-term studies to establish.
	Indic	cative BAT requirements
BAT for selection	1.	Improvement conditions should be agreed within a timescale approved by the Regulator;
Selection		 complete any longer-term studies (item 3 above), carry out any substitutions identified,
	2. ⁻	The Operator should maintain a detailed inventory of raw materials used on-site.
		The Operator should have procedures for the regular review of new developments in raw materials and the implementation of any suitable ones which are less hazardous.
		The Operator should have quality assurance procedures for the control of the content of raw materials.

INTRODUC	CTION TECHNIQUES EMISSIONS IMPACT									
Management	aterials Activities/ Ground Waste Energy Accidents Noise Monitoring Closure Installation issues									
Use of raw	2.2.2 Waste minimisation (minimising the use of raw materials)									
materials	The prevention and minimisation of waste and emissions to the environment is a general principle of									
Principles	IPPC. Operators will be expected to consider the application of waste minimisation techniques so that, wherever practicable, all types of wastes and emissions are prevented or reduced to a minimum. The steps below will also help to ensure the prudent use of natural resources.									
	Waste minimisation can be defined simply as:									
BREF Chapter	"a systematic approach to the reduction of waste at source, by understanding and changing processes and activities to prevent and reduce waste".									
2.6.8	A variety of techniques can be classified under the general term of waste minimisation and they range from basic housekeeping techniques through statistical measurement techniques to the application of clean technologies.									
	In applying this general principle it is necessary to bear in mind that many non-ferrous metals operations are waste recovery systems in their own right. Therefore in the context of waste minimisation and this Guidance, waste relates to the inefficient use of raw materials and other substances at an installation; replacing dross millings with virgin metal will reduce the amount of waste produced by the installation, but at an overall environmental cost. A consequence of waste minimisation will be the reduction of gaseous, liquid and solid emissions.									
	Key operational features of waste minimisation will be:									
	the ongoing identification and implementation of waste prevention opportunities;									
	 the active participation and commitment of staff at all levels including, for example, staff suggestion schemes; 									
	 monitoring of materials usage and reporting against key performance measures. 									
	For the primary inputs to waste activities, e.g. the waste to landfill, the requirements of this section <u>may</u> have been met "upstream" of the installation. However, there may still be <u>arisings</u> which are relevant.									
	References (see Ref. 9) provide detailed information, guides and case studies on waste minimisation techniques. Existing sector guidance should cover cleaner technologies and waste minimisation opportunities specific to the relevant sector.									
	Application Form Question 2.2 (part 2) Identify the <u>raw and auxiliary materials, other substances</u> and water that you propose to use.									
BAT for	With the application the Operator should:									
minimisation	1. Identify, from a knowledge of the plant, the main opportunities for waste minimisation and supply									
BREF Section 2.10	information on waste minimisation audits and exercises and the improvements made or planned. Indicative BAT requirements									
Emission and consumption data are given in subsection 2 of Chapters 3 to 12	 A regular waste minimisation audit should be carried out. Where one has not been carried out recently, an initial comprehensive audit should be carried out at the earliest opportunity within the improvement programme. New plants will need to have been operating for some time before an audit will be meaningful. Further audits should be at least as frequent as the IPPC Permit reviews. The audit should be carried out as follows: 									
	 The Operator should analyse the use of raw materials, assess the opportunities for reductions and provide an action plan for improvements using the following three essential steps: (i) process mapping; (ii) raw materials mass balance; (iii) action plan. 									
	The use and fate of all raw materials should be mapped onto a process flow diagram (see Ref. 9) using data from the raw materials inventory (see Section 2.2.1) and other relevant data. Particular attention should be paid to impurities in raw materials which will be concentrated within particular waste streams. Pollutants (e.g. dioxins) which will be created by the activity should also be identified. Data should be incorporated for each principal stage of the operation in order to construct a mass balance for the installation.									
	Using this information, opportunities for improved efficiency, changes in process and waste reduction should be generated and assessed and an action plan prepared for the implementation of improvements to a timescale approved by the Regulator.									

INTRODU	CTION	CTION TECHNIQUES EMISSIONS IN						MPAC	т			
Management	aterials Ac	ctivities/ atement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Water use				er is as a	coolant,	though some	e process	ses use wate	r in gas s	crubbing		
		The use of water should be minimised within the BAT criteria for the prevention or reduction of emissions and be commensurate with the prudent use of water as a natural resource.										
Reasons for reducing water use	Reducing water use may be a valid environmental (or economic) aim in itself, perhaps because of local supply constraints. In addition, from the point of view of reducing polluting emissions, any water passing through an industrial process is degraded by the addition of pollutants, and there are distinct benefits to be gained from reducing the water used, in particular:											
		 reducing the size of (a new) treatment plant thereby supporting the cost-benefit BAT justification of better treatment; 										
		•		•		sposed off to						
	pumpin	 associated benefits within the process such as reduction of energy requirements for heating and pumping, and reduced dissolution of pollutants into the water, leading to reduced sludge generation in the effluent treatment plant. 										
	The use of	f a simple	mass bal	ance for v	vater use	e will reveal v	vhere red	luctions can	be made.			
		In addition to the BREF, advice on cost-effective measures for minimising water can be found in Envirowise publications (see Ref. 10).										
		Application Form Question 2.2 (part 3) Identify the raw and auxiliary materials, other substances and <u>water</u> that you propose to use.										
	With the	applica	tion the	Operato	r should	d:						
BAT for water efficiency	1. Supp	ly informa	ation on w	ater cons	umption	and compari	son with	any available	e benchm	arks.		
-	2. Supp	ly a diagr	am of the	water cire	cuits with	indicative flo	ows.					
BREF Section 2.9 Emission and	 Describe the current or proposed position with regard to the indicative BAT requirements below, or any other techniques which are pertinent to the installation. 									nts below,		
consumption data are given in subsection 2 of Chapters 3 to 12	 Demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide to Applicants) or alternative measures. 											
	 Describe, in particular, any water audits already conducted and the improvements made or planned. 											
	Indicative	e BAT re	equirem	ents								
BAT for water efficiency	been oppoi some	carried o rtunity wit	ut recently hin the im ore an au	y, an initia proveme dit will be	al compre nt progra meaning	hensive aud mme. New j	lit should plants wil audits sh		ut at the e ve been o	earliest		
	• TI	he Opera	tor should	produce	flow diag	rams and wa	ater mass	s balances fo	or the acti	vities.		
	wi th be or	here not a lese (see elow and n reducing	available, Section 1 those in th	national t .2), or wh ne existin se beyond	oenchma ere benc g sector g	rks <mark>(see Ref</mark> hmarks are i guidance sho	. 11). In j not availa ould be ta	rison with se ustifying any able, the tech aken into acc tified by eac	departur niques de ount. The	es from escribed e constraints		
	so	crubbing s	systems a	re in use,	to identit		unities fo	r maximising		y where wet nd		
	as	ssessed a		ion plan p				se should be n of improve				
	с _г		,	,						Cont		

Cont.

INTRODU	CTIC	N TEC	HNIQU	IES	EN	IISSION	IS	I	ЛРАС	Т	
Manadamant	lateria input		Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	
Water use	2.	The following g	•			• •	•		issions to	water:	
BAT for water efficiency (cont.)		 Closed circuit cooling systems should be used where possible. Water should be recycled within the process from which it issues, by treating it first if necessary. Where this is not practicable, it should be recycled to another part of the process which has a lower water quality requirement. Drainage systems should be designed to avoid contamination of roof and surface water. Where possible this should be retained for use. That which cannot be used should be discharged separately. 									
	3.	Measures shou Section 2.3.13)		emented	to minim	se contamin	ation ris	k of process	or surfac	e water (<mark>see</mark>	
	4.	To identify the sassociated with waters, should treatment.	n each use	should b	e identifie	ed. Less cor	ntaminat	ed water stre	ams, e.g	cooling	
	5.	Ultimately wast applications, th be usable in the can vary, it can the quality falls treated water fr	e best con e process o be recycle below tha	ventiona directly o ed select t which tl	l effluent r when m ively, whe ne system	treatment pr ixed with fre in the quality i can tolerate	oduces a sh water / is adeq e. The C	a good water . While treat uate, revertin operator shou	quality w ted efflue ng to disc uld identif	hich may nt quality harge when y where	
		In particular, th individual proce they could com remains, howey particularly whe system could b benefits of prov	ess stream pletely rep ver, a conc ere waste h e produce	s or to th lace the centrated neat is av d. Where	e final eff ETP plan effluent s ailable fo appropr	luent from th t, leading to stream but, v r further trea	ne effluer much re vhere thi atment by	nt treatment duced efflue s is sufficient y evaporatior	plant. Ult nt volume tly small, n, a zero	imately, e. There and effluent	
	6.	Where water isvacuuming,evaluating ttrigger cont	scraping of the scope f	or moppi or re-usi	ng in pref ng wash v	erence to ho vater;	osing dov	vn;	d by:		

INTRODUCTION					EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.3 The main activities and abatement

(includes **directly associated activities** in accordance with the PPC Regulations)

Application Form Question 2.3

•

Describe the proposed installation activities and the proposed techniques and measures to prevent and reduce waste arisings and emissions of substances and heat (including during periods of start-up or shut-down, momentary stoppage, leak or malfunction).

With the application the Operator should:

- 1. Provide adequate **process descriptions** of the activities and the abatement and control equipment for all of the activities such that the Regulator can understand the process in sufficient detail to assess the Operator's proposals and, in particular, to assess opportunities for further improvements. This should include:
 - process flow sheet diagrams (schematics);
 - piping and instrumentation diagrams for systems containing potentially polluting substances;
 - diagrams of the main plant items where they have environmental relevance, e.g. furnace layout, extraction systems, fan capacities, abatement plant, chimney heights;
 - details of any chemical reactions and their reaction kinetics/energy balance;
 - control system philosophy and how the control system incorporates environmental monitoring information;
 - failure warning devices for abatement plant;
 - annual production, mass and energy balance information;
 - venting and emergency relief provisions;
 - summary of extant operating and maintenance procedures;
 - a description of how protection is provided during abnormal operating conditions such as startup, shut-down and momentary stoppages.

If there is uncertainty, the degree of detail required should be established in pre-application discussions.

- 2. Describe the current or proposed position for all of the requirements listed below, for any additional requirements listed in subsections 2.3.1 to 2.3.14, and for any others pertinent to the installation.
- 3. Identify shortfalls in the above information which the Operator believes require longer-term studies to establish.
- 4. Demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide to Applicants) or alternative measures.

In assessing the integrated impacts of proposals and balancing the impacts of different techniques it should be noted that energy should be taken into account whether or not there is a Climate Change Levy Agreement in place (see Section 2.7.3).

Indicative BAT requirements

The following techniques should be applied where appropriate to all installations. Further requirements are identified or specific requirements emphasised in the process-specific subsections below.

- 1. Matching of the design against proposed raw materials and products. Features to which particular attention should be paid are:
 - Fuel selection and burner design.
 - Minimising the number and frequency of transfers of hot metal.
 - Matching air extraction and filtration systems to requirements.
 - Constraints on purchasing policy and raw material selection.

Cont.

BREF Sections 3.4, 4.4, 5.4, 6.4, 7.4, 8.4, 9.4, 10.4, 11.4 and 12.4

BAT for the main

BREF Sections

techniques and

give process

descriptions

3.1, 4.1, 5.1, 6.1,

7.1, 8.1, 9.1, 10.1,

11.1, 12.1 describe

activities

the applied

INTRODUCTION			TECHNIQUES			EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	

 Where exhaust volumes are large, avoid the use of after-burners as a means of co emissions of carbon monoxide, VOCs and dioxins. Preferred alternatives include: only processing material free from oil, grease plastic or paint; 											
	o remove										
installing and treatment systems such as weakers, contributes or sworf drivers t	o remove										
 installing pre-treatment systems such as washers, centrifuges or swarf dryers to organic contamination. 											
 In assessing proposals which may involve the use of after-burning, the quantity of dioxide generated should be balanced against the quantity of VOC destroyed. Eac metre of exhaust gas passed through an after-burner could generate some 30 gm dioxide. 	ch cubic										
 Provision of dry storage facilities for those raw materials and solid wastes which may with or contaminate water. 	ay react										
 Provisions for handling and storage of dusty materials including dust removed from 	n air filters.										
 Bunding for tanks and pipes containing fuels and other fluids. Use above-ground f and pipelines so that leaks can be detected quickly and repaired. 	uel lines										
 Design of hydraulic systems so as prevent contamination of furnace charges with h oils in the event of leakage. 	nydraulic										
Emergency power supplies for safety and environmentally critical plant.											
2. Minimising the impact upon water consumption and pollution by:											
 Design, control and operation of storage and process areas so as to avoid contami storm water. 	ination of										
Design of the storm water drainage system so as to minimise the risk of contamina	tion.										
Use of a closed cycle system for cooling.											
3. Minimise waste generation by:											
 Optimising the ratios of slag to metal used in melting operations. 											
Where possible, re-using slags.											
Recovering re-usable materials from slags.											
4. Avoiding the risk of creating secondary pollution by:											
 Using biodegradable hydraulic oils in any location where failure could lead to water contamination. 											
 Not using chlorinated cutting oils on any activity that gives rise to scrap which could recharged into a melting furnace. 	d be										
Avoiding charging scrap contaminated with chlorinated cutting oils into melting furn	aces.										
Avoid using chlorinated solvents for degreasing metal.											
5. Minimising the potential for noise nuisance, in particular from:											
Elevated machinery such as conveyors or skip hoists.											
 Mechanical handling, such as discharge from billet casters 											
Vehicle movement, especially internal load carriers such as fork lift trucks.											
6. Ensuring proper operation by:											
 Operation of an effective environmental management system. 											
 Minimising the impacts of accidents and failures through an emergency plan. 											
 Further requirements of relevance to specific activities are identified within each subset below. 	ection										
INTRODUC	TION	TEC	HNIQL	JES	E٨	IISSION	IS	1	MPAC	Т	
--	---	---	---------------------------	--------------------------	-----------------------	--	------------	---------------	-------------	------------------------	--
Management		ities & ement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	
Copper & copper alloys		· · ·	ind cop	per all	oys						
		melting									
BREF Sections 3, 3.1.1, 3.1.2	Process:	likeliho		being con	structed	eration in the within the for					
	2.3.1.2 N	/ire, rod	and tub	e produ	ction						
BREF Section 3.1.4	Process:	reducir	g conditio	ns. For tu	ube prod	er is melted i uction the me ce, then cast	etal is fu	urther deoxid	lised by a	idding	
SI 1973 2.2 A (1) (b)(ii) or A (2) (a)						cut to length before being					
			s annealing on/drawing		face trea	tment operat	ions ma	ay be incorpo	orated inte	o the	
			rom extrus from the			ations may be	e recycl	ed to the sha	aft furnac	e, as are	
	Water:					ner than as a billage of lubr				otential for	
	Land:							potential for	ground co	ontamination	
	Air:	from spillage of lubricants and de-greasing fluids. Air: Melting must be done under reducing conditions in order to minimise the amount of oxygen absorbed in the molten metal and in the case of rod to protect the refractory. With natural gas as a fuel this can be achieved by operating the burners fuel-rich to achieve a carbon monoxide (CO) concentration of under 1% in the products of combustion at the burners. The counter-current flow whereby hot gases rising up the shaft give up their heat to descending solid copper means that any oils or grease on th copper are distilled out, giving a VOC release. Carbon monoxide and VOCs could be greatly reduced by after-burning, but at the cost of a significant reduction in energy efficiency.									
		release from th	s of produ	icts of cor This clea	nbustion aning ope	neating furna . After proce eration has a	essing, l	ubricants wil	I need to	be removed	
	Energy:	•				f gases and i e minimised.		the shaft fu	rnace, en	ergy losses	
	Noise:	opportu	unity for ra	ndom noi	se if han	ing large am dled careless burners and	sly. Sav	ws and produ	uct handli	ing are also	
	Accidents: The environmental potential of accidents is not great, although the possibility of ground contamination being caused by spilling cleansing fluids should not be overlooked.										
	Application Form Question 2.3 (cont.)										
	 With the application the Operator should: Supply the general application requirements for Section 2.3 on page 28 for this aspect of the 								t of the		
	activitie	-	a applica	adon requ			2.0 011	aye 20 101 li	ns aspec		
BAT for copper wire, rod and tube production	 Car 	 dicative BAT requirements Carbon monoxide generation to be minimised by sequential monitoring of combustion conditions and independent automatic control of fuel/air ratios at the burners. 									

- Charge handling systems designed to avoid the possibility of contaminating feedstock with hydraulic oils or lubricants.
- Purchasing and inspection procedures to prevent charging of material contaminated with drawing or extrusion lubricants.

3.1.4.1

BREF Section

INTRODUC	CTION	TECHNIC	UES	E١	/IISSION	IS	11	MPAC	Т
		ities & Groun ement water	d Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Copper & copper alloys	2.3.1.3 N Process:	lelting copper, Copper can be a wide range o	alloyed wit	h zinc, tir	n, nickel, alui	- minium,	lead and oth		
Summary of the activities		bronze (copper or zinc).	and tin), c	upro-nick	el, aluminiur	n bronz	e and gun m	etal (copp	per with tin
BREF Section		In some alloys frequently alloy (electrical cont	s which inc	corporate	cadmium (o	verhead	l conductor v		
3.1.5.1 Section 2.2 part A (1) (b) or (c), or Par A (2). Also, if directly associated with such a melting		ed produced to booded to bosure sys	n which it is ct. Where permit the tem must be elting and						
activity, Part B (b)		In the case of r materials, they destroy smoke	shall only	be proces	sed in furna	ces fitte	d with suitab		
		A variety of fur product.	naces can	be used,	depending u	pon the	source of ra	w materia	al and
		Induction furna without recours				ed alloy	composition	can be a	chieved
		Where it is nec be generated,							
		In all furnaces, only for copper considered BA	or cupro-n						
BREF Section 2.5.1.9 BREF Table 3.3.7, Section 3.4.3		Machining was cutting oils. If not fitted with a they are charge centrifuges or a	he melting in after-bur ed to the fu	furnace is ner, then rnace. Tl	s not design the oils mus nis may be a	ed to proto	event the en noved from s I by washing	ission of uch wast	smoke, or is es before
000101 0.4.0		A typical swarf approximately passed through air.	400°C to d	rive off vo	latile fluids.	The ex	haust gas fro	om this dr	um is then
		If fine millings a will be necessa							dled, then it
		If material cont swarf dryer, the reducing the co	en that furn	ace or dr	yer shall be i	fitted wit	th an abatem		
	Water:	Water is used a contaminated v designed to pre	vith oil. Wh	nere oily s	warf is to be	used, s	storage and	handling	must be
	Waste:	Furnace opera will also arise v quantities of du	vhen furna						
	Air:	Fugitive releas metal, will requ filtration to rem present, dual fi releases of VO	ire careful ove metal Itration is n	attention. oxides. V	Contained /here highly	emissio toxic m	ns to air will aterials such	normally as cadm	require iium are
	Energy:	Energy require In particular, af energy input.							

INTRODUC	CTION	TEC	HNIQU	JES	EN	IISSION	IS	I	MPAC	Т
Management		ities & ement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Copper & copper alloys	Noise:	machin	es are par nisms and	ticularly p	prone to o	ause annoy	ance. S	round. Chan Sound insula be well desig	tion arou	nd cleaning
	Accidents:		air filtratio			can cause si	gnifican	t air pollutior	n whilst th	e fire is
	Application Question 2		>	Melting	copper, r	nelting and i	making	copper alloy	S	
	<i>With the a</i>					for Section :	2.3 on p	age 29 for th	nis aspec	t of the
BAT for melting	activitie Indicative		quiremen	its						
copper, melting and making copper alloys								sure that un in furnace c		naterials
						and operate hise the risk		to minimise	the quant	ity of dust
	release This as	es which i sessmen	may arise	as a cons ude cons	sequence	of a fire in e	extractio	on the poter n ducts, filte he furnace d	rs or the	oxide area.

INTRODU	INTRODUCTION		HNIQL	JES	EN	/IISSION	IS	I	MPAC	Т
		tivities & atement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Aluminium & aluminium		Aluminiu								
alloys BREF Section	2.3.2.1	Primary s Section 2	smelting 2.3.9)	(exclud	ing car	bon anode	manu	facture wh	ich is co	overed in
4.1.1	Process		st stage in a) by the l			um from bau	ixite is t	o manufactu	re alumin	ium oxide
SI 1973 Section 2.2 Part A (1) (a)		connec	tion with n	nanufactu	ure of alu	minium meta	al. UK s	United Kingo melters are l ly that addition	likely to re	ely upon
BREF Section 4.1.1.1		underta		lude this	process	in a Chemica		F and the No		
SI 1973 Section 4.2 Part A (1)	(a) Pro	oduction o	f alumini	ium by e	electrol	ysis				
(a) (v)	Process.	which r		sists of m				on of alumina iium fluoride)		
BREF Section 4.1.1.2		ratio fo These	r cryolite. I	Many also are made	o add fluo with the	orides of mag intention of i	gnesium	in excess of n, calcium or ng conductivi	lithium to	the melt.
		suppor	ted within	a steel sh	nell, form	s the cathod	e. The	ody surround anode consis r rods attach	sts of a ro	ow of pre-
								ccumulates of carbon ano		
		Molten the cas		n is remov	ved perio	dically by va	cuum s	yphon into cr	ucibles fo	or transfer to
						naintained at s are replace		en 2% and 6% ey burn off.	% by com	puter-
			electrolysi um fluorid			of fume cont	aining t	ooth hydroge	n fluoride	and sodium
		fluoride alumini contam	in the bat um fluorid	h, additic e is also he alumir	ons are m required na feed.	ade of alumi to neutralise	inium flu sodium	sodium fluor uoride or cryo compounds s operating a	plite. Son	ne as
		during	operation,	with failu	ire usuall	y showing its	self by ir	However, it on contamin ctor bars of th	ation of th	he metal.
		Cathod replace		ave a life	of betwe	en five and e	eight ye	ars before th	ey need t	o be
	(b) Re	fining prio	r to cast	ing						
BREF Section 4.1.1.3	Process.	some c		cles, toge				contain sodi n or lithium if		
								holding furna h nitrogen or		
		After ga	as treatme	ent the me	etal is filte	ered to remov	ve solid	inclusions b	efore cas	ting.
		Alloys	can be ma	de prior t	o casting	by adding a	ppropria	ate metals.		
				-		-	-	secondary m		
	Water:							he casting sl ater by dust f		

INTRODUC	CTION	TE	CHNIQU	JES	EN	/ISSION	IS	IMPACT			
		ctivities 8 batement		Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	
Aluminium & aluminium alloys	Air:	There hydro fluori poter chlori oxide comb	e are potent ocarbons (P des from the tial for signi de, oxides o s are gener ustion arisin sions from tl	FCs), par e reductio ficant rele of nitrogen ated durin ng from th	ticulates, n cells. P eases of s n and me ng electro ne casting	eases of oxic hydrogen flu FCs are rele sulphur dioxi tallurgical fu olysis and in g shop and o ling operatio	uoride a eased du de from me from smaller ther ope	nd metallurg uring anode the cells, ar the refining quantities as erations. Th	ical fume effects. T nd chlorin operation s product ere are po	There is the e, hydrogen ns. Carbon s of otential dust	
	Waste:	The o	arbon is sa	turated w	ith fluorid	g periodic re e salts, and l into soluble	also the	e nitrogen pro			
						II generate d ertheless ulf				nt in	
	Energy:	and 1 conce	6.5 kWh/tor entration fall	nne of me s below a	etal. Ener	y-intensive. rgy efficiency evel, causing on is dramati	y in the g an "an	cell is reduce ode effect".	ed if the a For the d	lumina uration of	
	Acciden		of-control an ase in fluorio			use the surf	ace crus	st to melt, giv	ving a ma	ssive	
			tected weal			can cause r	un-outs	of electrolyt	e and me	tal, with	
	Noise:	Air po to de	ollution cont sign of fans	trol equipment is on a massive scale, and proper attention must be give s and bag filter cleaning systems.							
	Application Form Question 2.3 (cont.) Primary smelting of aluminium and aluminium alloys										
BAT for primary smelting BREF Section 4.2.2.1	 With the application the Operator should: Supply the general application requirements for Section 2.3 on page 29 for this aspect of activities. Indicative BAT requirements Electrolysis should be carried out using centre worked pre-baked cells with automatic mulalumina feed points. Cell process control is based upon active cell databases and monitoring cell operating patto minimise energy consumption and to reduce the frequency and duration of anode effect Complete hood coverage of the cells connected to a gas exhaust and filter is required. Thoods must be robust and extraction rates adequate. A sealed anode butt cooling syster be provided. More than 99% fume collection from the cells is to be achieved on a long-term basis. Thin to open hoods and change anodes is to be kept to a minimum. A programmed system is used for cell operations and maintenance. Efficient cleaning methods should be used to remove fluorides from returned butts. The reflective extraction and filtration is necessary in this area. Where local, regional or longer-range environmental impact studies require reductions in dioxide emissions, low sulphur carbon should be used for electrode manufacture.									multiple g parameters ffects. . The cell stem must Time taken n is to be he use of he use of s in sulphur remove at	
		uction cells		esent. I	ne alumin	na used in th	e scrud		recycled t	o the Cont.	

		TEO								T		
INTRODUC			HNIQU	JES		AISSION	12		MPAC			
		ties & ment	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Aluminium & aluminium alloys	scrubbe	er and fal						nove fluoride forming proce				
	2.3.2.2 Se	econda	ry meltin	ng and r	emelt of	^r aluminiur	n and a	aluminium	alloys			
	Process:		vers a wic um scrap.		of activitie	es, from rem	elting al	uminium ing	ots or ver	y clean		
		Furnac	es used in	clude:								
BREF Section 4.3.4			uction furn erberatory			ner grades of	f scrap v	with limited c	ontamina	tion;		
SI 1973 Part A (b) and (c)	Water:		There is limited risk of water contamination arising from storage of raw materials, e.g. swarf, and also furnace wastes such as slag or dross.									
	Air:	Where	salt fluxes	are used	d, there is	significant p	ootentia	l for air pollut	ion.			
	<i>Waste:</i> Where furnaces are used to recover metal from lower grades of scrap, there is a potential for generating drosses and slags.									is a		
	Energy:					s for recove d to make vi		l remelting al al.	uminium	are but a		
	Accidents:	Where liquid chlorine is used to refine molten metal, all the potential for accidents associated with liquid chlorine exists. Otherwise the most frequently encountered accident will be associated with charging oily, greasy or plastic coated scrap – this will generate much smoke and heat in the exhaust system and may destroy the filtration plant. Other accidents such as may arise if water or munitions are introduced into the furnace can have catastrophic local effects, but are of limited environmental significance.										
	Application Question 2.		\geq	Second	lary melti	ng and reme	lt of alu	minium and a	aluminiur	n alloys		
BAT for secondary melting and remelt of aluminium and aluminium	With the ap 1. Supply activitie	the gene s.	ral Applic	ation requ			2.3 on p	bage 29 for ti	his aspec	t of the		
alloys		· · · · · · · · · · · · · · · · · · ·										

1. No additional requirement.

INTRODUC		TECHN		JES	EN	/ISSION	IS		MPAC	Т			
Wanadement			round /ater	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues			
Aluminium & aluminium alloys	2.3.2.3 E. Process:	Secondary drosses ar processed	raw m id engi using	naterials f ineering v swarf dry	rom whic vastes su ers as de	om second h aluminium ich as swarf escribed in S	is extra millings ection 2	acted are scr and turning 2.3.1.3.	ap metal, s. The la	tter may be			
BREF Sections 4.1.2.3 and 4.1.2.4		down the r	nore fr	iable non	-metallic	gs and dross portion, then n then be rec	screen	ing. The lar	ger size f	raction will			
SI 1973 Part A (1) (a) Part B (b) when directly associated		 rotary f 	urnace		can be us					ap; slags, and a			
with a Part A (1) activity		normally us particles fr	sed in om oxi	rotary fur dation. V	naces wł Vetting ca	r mixtures of here they aid an be aided t n and magne	heat tra by mino	ansfer and p r additions o	rotect sm	all metal			
	Smaller amounts of salt flux may be used in other furnaces to minimise metal loss through oxidation, and also to assist dross removal. It is normal to avoid fluoride fluxes when using electric furnaces.												
		It is norma	l to ave	oid fluorid	e fluxes	when using e	electric	furnaces.					
	Some reverberatory furnaces fitted with after-burning systems are capable of has scrap contaminated with oil, grease or paint; otherwise all such materials should treated to remove contamination before being added to the melting furnace.												
	Water: No water is used in this process, but slags and dross contaminate water if they come into contact. Any lead dissolved salts and sulphide as well as suspended so								chate is likely to contain ammonia,				
	Air:	Dry milling wet – see '			dust and	effective filt	ration is	required. N	laterials r	nust not get			
	Waste:	portion of s	slags a	ind dross	es, these	are not yet r	eliably	make beneficial use of the non-metallic iably established on a commercial nain a significant portion of unusable					
	Accidents:	Flooding o		s of dros	s, slag or	oily swarf ca	an give	rise to seriou	us water				
	Noise:	Milling ope	rations	s are pote	entially no	bisy.							
	Application Form Extraction of aluminium metal from secondary raw materials With the application the Operator should:								als				
	 Supply the general application requirements for Section 2.3 on page 29 for this aspect of the activities. 								t of the				
BAT for recovery from	Indicative												
aluminium drosses and slags	• Use	t emissions of of enclosur rging, drossi	es or h	loods pro	vided wit	h extraction f	to the fu	ume arrestm	ent syster	n over			
BREF Section			-			ms for feedin	g and f	or dross coo	ling.				
4.2.2.2		t unintended oof buildings		ig of slag	s, drosse	s and recove	ered me	tallics handli	ing and st	oring within			
	BAT for swar	f drvers is o	overed	l in Sectio	on 2.3.1.3	above.							

BAT for swarf dryers is covered in Section 2.3.1.3 above.

INTRODUC	CTION	TECH	NIQU	ES	EN	ISSION	IS	IN	ЛРАС	Т
Manadement			round /ater	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
			ater							133003
Lead, zinc & cadmium	2.3.3 Le	ad, zinc a	and c	admiu	ım					
	2.3.3.1 F	Primary sm	_							
						ated in ores arate and pu		ncentrates, a metals.	nd a num	ber of
	(a) Smel	ting lead/zi	nc sul	phide c	ores usi	ng the ISF	blast f	urnace		
BREF Sections 5.1.1.1, 5.1.5.1 and 5.5.1.7	Process:	the purpos	e of ma	nufactur	ring lead	and zinc. Su	ulphuric	o process mi acid is manu so recovered	factured	
SI 1973 Section 2.2 Part A (1) (a)		agglomera sulphide m	ted into inerals or char	pellets i burn to ging to th	in prepara oxides ar ne blast fi	ation for sinte	ering. E id oxide	other recove During the sire mix which c ng reaction is	iter proce an be cru	ess, the ished and
		with some	of the a is clean	arsenic p ied to rei	resent in move and	the feed. The feed the	ney also	m all of the r contain cad nents, then pa	mium an	d lead. The
		heated col	ke. Air,	sometim	nes enrich		gen, is l	ace, and is fe plown in at th		
								admium are ed as liquids		l, and
		stage fract all of the c fed to a se	ionating admium cond, s	g column n and mo lightly co	i. In the f ost of the poler colu	irst stage the zinc are dist	e molter illed. Th e cadm	lenser, then n zinc is fed in ne vapours a ium distils ar n zinc.	nto a coli re conde	umn where nsed and
		the lead re and arseni antimony a	cycled f c. Fina as sodiu	to the sp Ily, it ma ım arsen	lash cono y be trea ides or a	denser. Zinc ted with sodi ntimonides.	rfrom th um to c This zir	timony and c his stage is co onvert residu nc is less pur le for galvani	ooled to i al arseni e than se	remove iron ic and
						ecious metal ion 2.3.3.2 b		nt in the sme	lter feed	and will be
	Water:	give rise to before fina compound	liquid e I discha s in dus ations is	effluents arge. Th sty form;	contamir e process some of	nated with he s involves ha these have s	eavy me andling a significa	ions prior to itals. These a range of he nt solubility in potential to po	must be avy-meta n water.	treated al The scale of
	Air:	There is si si si sintering, s					ing fron	n raw materia	als handli	ing,
	Land:	Pollution a	rising in	ndirectly	from emi	ssions to air	can be	significant.		
	Waste:	Smelting w	/ill give	rise to a	range of	slags.				
		Recycling	opportu	inities ma	ay be res	tricted by the	e accum	ulation of un	desirable	e elements.
	Energy:							heat from expositible exha		
	Accidents:					ontainment se otentially se		or either the l	olast furn	ace or the

INTRODUC	TION	TEC	INIQU	JES	EN	IISSION	IS	11	MPAC	Т		
		ities & ement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure		llation ues	
Lead, zinc & cadmium	Application Question 2.	Form	\geq	Smeltin	g lead/zir	nc sulphide c	ores usi	ng the ISF b	last furna	ce		
 With the application the Operator should: Supply the general application requirements for Section 2.3 on page 29 for this aspect activities. BAT for an integrated ISF smelting operation Indicative BAT requirements Emergency power should be provided for safety and environmentally critical systems of that they are not deprived of power or water during reasonably predictable failure mode Vibration or other form of blockage detection is needed for the distillation stages. 										to ens		
	(a) Recov	very of l	ead fron	n scrap i	batterie				⁹ (lood or	, the c	vido	
BREF Section 5.1.2.1	Process:	Process: A typical lead battery can contain up to 30% lead alloy and up to 45% lead as the oxid or sulphate in a paste form. Scrap batteries are a major source of secondary lead, ar not reprocessed, constitute an important hazardous waste stream.										
SI 1973 Section 2.2 Part A (1) (a) or Section 2.2 Part A (1) (d)		Whole battery treatment is not practised in the UK; instead, batteries are drained of a and broken down by means of impact crushers. Crushed material is passed through series of screening, wet classifying and filtration processes to provide separate fracti These contain metallic components, lead oxide, sulphate paste, polypropylene, non-recyclable plastics, rubber and dilute sulphuric acid.									gh a ctions.	
		sodium	carbonate	e or sodiu	ım hydrox	paste is norr kide. This ca atmosphere	an reduc					
		rota	g in the U ry kilns; smelt furr		ominantly	carried out	in:					
		slag and	furnaces a d metal. N compoun	Most of th	or oil-fired le sulphur	batch opera	ations w ge is fixe	hich involve ed in the slag	separate g as a soc	tappir lium–i	ng of iron–	
		the furn a steel l maximu	ace, and ance sub	lead bullio merged ir e of slag,	on is perion the moltadditions	rised paste a odically tapp en bath. Wh of reductant	ed. Fue	el oil and air process ves	are inject sel conta	ed thr	ough e	
	Water:	Risk of	spillages	of acid fro	om raw m	aterials hand	dling.					
	Waste:		ude unus int dusts.	able mate	erials arisi	ing from the	breakin	g activity, sl	ags from I	meltin	g and	
	Air:		e operatio ity of acid		/e rise to	metallic fum	e, and s	some sulphu	rous exha	iusts a	and a	
	Energy:	Acid mis	sts in exh	austs limi	t the pose	sibility of rec	overing	energy from	n furnace g	gases		

INTRODUC		TECH	INIQU	JES	EN	AISSION	IS		MPAC	Т		
Management	terials Activi puts abate	ties & ment	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Lead, zinc & cadmium	Application Question 2.		>	Recove	ery of lead	l from scrap	batteries	S				
	With the ap	plicatio	n the O	perator	should:							
	1. Supply activitie		al applica	ation requ	uirements	for Section 2	2.3 on <mark>p</mark>	age 29 for t	nis aspec	t of the		
BAT for	Indicative BAT requirements											
recovery of lead from	1. Effective control of raw materials to prevent excessive accumulation.											
scrap batteries	2. Good d	esign and	mainten	ance of r	eception	area to preve	ent acid	spillage.				
	3. Reliable	e operatio	n of brea	ker to en	sure cons	sistent feed to	o smeltir	ng furnace.				
	 Good containment and reliable temperature control to prevent fugitive emissions of lead fu the atmosphere. 											
	 Effective management of waste produced by the process, in particular to avoid accumulations cross-contamination. 											
	 (b) Recovery of lead from other scrap and residues Metallic lead may occur in a variety of forms, i.e. contaminated with plastic or bitumen, and alloyed with 											
BREF Section 5.1.2.2	<i>Process:</i> In some existing plants, clean scrap is melted in kettles, which are indirectly fired by gas											
SI 1973 Section 2.2 Part A (1) (a)	Process:	or oil. D	ross and	tramp m	aterials a	o is melted in re skimmed ally be recycl	from the					
and Section 2.2 Part A (1) (d)		e.g. in ca be melte extractin combust	ases whe ed in kettl ig all of th tion of an nt, scrap	ere it cannes. The ne gases y materia	not easily gas extra from the al present	o containing o be segregat ction system melting pot o , in particula contaminan	ed at so and aft during pe r during	ource or at a er-burner sh eak producti charging an	treatmen ould be o on of gas d drossin	it site, it may capable of ses from the ig off. For a		
			s, includi es in a ro			ags, togethe	r with me	etallic lead s	crap, are	smelted		
			ntaining r atory furr		from the p	production of	f tetraalk	xyllead are n	nelted in	gas-fired		
	Water:	Storm w	ater cont	aminatio	n is the m	ost significa	nt sourc	e.				
	Waste:	Includes those drosses which cannot be recycled, together with scrap plant items.										
	Air:				-	, also smoke			-	taminants.		
	Energy:			-		inities for en		-				
	Accidents:				-	molten meta			the heati	ng gases.		
			-			using furnac						
		Fires in fabric filters caused by the carry-over of incandescent particles.										



(C) Refining	load
L) Remmu	<i>ieau</i>

	Process:	Lead produced from both primary and secondary processes may contain varying amounts of copper, silver, bismuth, antimony, arsenic and lead.
BREF Section 5.1.3 SI 1973 Section		UK lead refineries are pyro-metallurgical, consisting of a series of kettles indirectly heated by oil or gas. Copper is removed first as a sulphide dross, following sulphur additions if necessary.
2.2 Part A (1) (b) or Section 2.2 Part		Arsenic, antimony and tin are removed as an oxide dross following reaction with a mix of sodium nitrate and caustic soda. Air can also be used as the oxidising agent.
A (1) (d)		Silver is collected by addition of zinc and removed in an intermetallic crust of zinc and silver. Zinc is removed from the crust by vacuum distillation and fine silver is produced following further refining with oxygen.
		Calcium and magnesium are used to remove bismuth by forming a calcium– magnesium–bismuth alloy dross, which is removed by skimming. The dross is then oxidised using lead chloride, chlorine gas as a caustic soda/sodium nitrate mixture, and the calcium magnesium oxide is removed by skimming. A bismuth–lead alloy is recovered and undergoes further refining to produce bismuth.
		The pure lead is cast into blocks or ingots. Fume, drosses and other residues can be recycled by smelting in a rotary or small blast furnace.
	Water:	The process does not require water. Potential contamination of storm water must be avoided.
	Waste:	Drosses that cannot be recycled may contain arsenic and are potentially hazardous if exposed to moisture.
	Air:	The process involves removal of dry drosses and has potential for generating dusts rich in heavy metals.
		Metallurgical fume can be generated at the metal surface.
	Accidents:	Overheating molten metal caused by control failure, or by failure of container allowing metal to spill into firing zone.

INTRODUCTION TECHNIQUES EMISSIONS IMPACT													
Wanadement		ities & Ground ement water	¹ Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues				
Lead, zinc & cadmium					'		1	'					
Caumum	Application Question 2		Lead r	efining									
	With the a	pplication the	Operator	should:									
	1. Supply activitie	the general Apples.	ication req	uirements	for Section	2.3 on	page 29 for t	his aspec	t of the				
	Indicative	BAT requirem	ents										
BAT for lead refining	1. Good te	emperature contr	ols with fai	ilsafe syst	em to preve	nt overh	eating.						
	2. Collecti	ion and filtration of	of dust and	l fume fro	m drossing c	peratio	ns.						
	(d) Meltin Process:	ng and alloying Melting and allo				directly	heated cruci	ihle furna	ces or				
BREF Section 5.1.4	1100033.	kettles, using e	lectricity, c	oil or gas.	Refined lead	d is mel	ted in a kettle						
SI 1973 Section 2.2 Part A (1) (b) or Section 2.2 Part		elements are added, taking care to control temperature. Lead and lead alloys can be cast into permanent cast iron moulds or into ingots using conveyor-type casting machines, or into rod or sheet, using continuous casting machines.											
A (1) (d)		For potential is	sues and E	BAT see <mark>S</mark>	ection 2.3.3	. <mark>2.(c)</mark> at	oove.						
	(e) Reco	very of zinc fro	m scrap	, dross a	nd other r	esidue	s						
		nd scrap which are n copper alloying;		and signif	icant to the l	JK secc	ondary zinc ir	ndustry in	clude:				
BREF Section		from the die cast		ry;									
5.1.6	•	er's ashes; from the shreddi	na of steel	scrap, inc	cluding scrar) cars:							
SI 1973 Section 2.2 Part A (1) (a)	zinc dros	sses;	.g e. e.ee.	со. ср,	naamig eerep	,							
		c or spelter; e from the burnin	q of tyres.										
	Process:	The process ro the degree of c	ute for zind		depends or	n the for	m and conce	entration	of zinc, and				
		Galvaniser's as with ammoniun separate the no fraction is then	n and zinc on-metallic	chloride. phase fro	They are bo	th mille	d and classifi	ed or scr	eened to				
		Hard zinc or sp holding furnace generated from cast industry co of treatment sc chloride remove	s and galv hot dip ga ontain a mi hemes are	vanising ta alvanising xture of zi used for	inks. Top dr Drosses, s inc metal and these materi	oss is a kimminę d oxide	i zinc–iron–al gs and other with little or r	luminium residues 10 chlorid	alloy from the die e. A variety				
		Processing scr subjected to a i non-metallic fra reverberatory fr about 440°C. A inner liner, thro	ange of te ctions. Th urnace, wh An alternat	chniques ne non-fer nich allows ive proces	designed to rous fraction lead to be t ss uses an ir	separat is melte apped f idirectly	te the ferrous ed in two stag first at 340°C fired rotary l	, non-fer ges in a g , followed kiln with a	rous and gas-fired I by zinc at				

				HNIQUES		EMISSIONS			IMPACT			
Management	Materials inputs	Activities abatemen	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		

Lead, zinc & cadmium	Water:	The processes do not normally use water.
	Waste:	Potentially hazardous wastes are produced.
	Air:	Raw materials handling and waste handling are potentially dusty.
		Operations involving molten zinc are potential sources of metallurgical fume.
	Accidents:	If drosses which contain arsenic are allowed to come into contact with water, then arsine can be formed.
	Noise:	No special significance.
	Potential poll	utants and BAT comment as per Section 2.3.3.2.(c) above.
	(f) Meltin	g and alloying processes for zinc
BREF Section	Process:	Melting and alloying are usually carried out in indirectly fired crucible furnaces or induction furnaces with strict temperature controls, to prevent metal fuming.
5.1.7		Alloy additions can be in either solid or molten form.
		When alloys are made from impure, raw materials, fluxes containing zinc chloride, magnesium chloride and sodium silicofluoride may be used. This will result in the release of gaseous silicon tetrafluoride, which can only be removed by hydrolysis to hydrogen fluoride in a wet scrubber.
	Potential poll	utants and BAT comment as per Section 2.3.3.2.(c) above.

IPPC

INTRODU	DDUCTION TECHNIQUES EMISSIONS IMPACT						Т					
Wanadement		tivities & batement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Precious metals		Precious										
BREF Section 6	involving	gold, and th	ose relatin	g to the p	olatinum g	group (PGMs	s). The	to those invo re are a wide s of processe	e variety o	of processes		
0000000	2.3.4.1	Silver										
	(a) Se	condary si	melting o	peratio	ns							
BREF Section 6.1.1.1	Process	rich ash. and parti blended	The exha	ust gases fore discl materials	s must be harge into	e treated to d the atmosp	lestroy o here. 1	a rotary kiln dioxins and t he ash from nder and flux	o remove the kiln o	acid gases an be		
		The silver containing residue is smelted with either copper or lead oxides. Normally a rotary furnace is used for the copper route and a blast furnace for the lead route. The metals act as a carrier for the silver and are subsequently oxidised to separate them in a cupellation furnace or within the rotary furnace using an excess of oxygen.										
								hole. Accum ation and disp		ags are re-		
		Gases a	re collecte	d and filte	ered befo	re discharge						
	Water:							ubbing mediu s for surface				
	Waste:	Include b	last furnad	e slags a	and waste	e refractory g	generate	ed during ma	intenance	Э.		
	Air:	be gelati Incinerat give rise	ne based, or ash is d	in which ry, dusty ontaining	case inac and easi lead and	dequate incir ly blown. Th	neration le smelt	on. Photogra can give rise ing and cupe / which enter	e to foul c ellation st	dours. ages can		
	Energy:							e is such that d re-use will				
	Noise:	Fans and	d filters to r	ninimise	emission	s to air are t	he majo	or potential so	ource.			
	Acciden	ts: Failures	in the extra	action sys	stem cau	sing emissio	ns of ur	treated gas	to air.			

Fire in the paper or film storage hopper causing air and water pollution.



INTRODU		TEC				AISSION	19	II	MPAC	т 1				
М		ities &	HNIQU Ground							I Installation				
Wanadement		ement	water	Waste	Energy	Accidents	Noise	Monitoring	Closure	issues				
Precious	2.3.4.2 G	old												
metals	(a) The M	liller pro	ocess fo	r <mark>purific</mark>	ation of	reclaimed	gold							
BREF 6.1.2.1								vill contain a						
SI 1973 Section 4.2 Part A (1) (b)		indirectly the meta stable liq Other me	heated cr l, and at th uid or vola etals are d	ucible or his tempe atile chlor riven off	electric f erature go ide. Molf as volatile	urnace to 10 ld is the only en silver chle e chlorides a	00°C. (/ metal v oride ris nd are (ss the impure Chlorine gas which does r ses to the sur collected as a cally if requir	is then in not react t face of th a slag by	jected into o form a ne melt.				
		the fume selection this hygr	produced of abaten oscopic m	will vary nent plan aterial ar	from pro t. Two-st nd to remo	cess to proce age alkaline ove any exce	ess and wet scr ess chlo	er degree. The will conseque rubbers are r rine. If signifi- re the dew po	ently affe equired to ficant qua	ect the cope with antities of				
	Air:	Excess c	hlorine, co	ondensed	d chloride	s and other r	netallur	gical fume a	re genera	ited.				
	Water:	Contamii	nated wate	er must b	e dispose	ed of.								
		are recycled to smelters. The scale is in any case very small.												
	Noise:	Noise: Is insignificant.												
	Accident:	Accident: There is a potential for loss of containment on the chlorine system.												
	<i>Energy:</i> The small scale of the operation renders heat recovery impracticable.													
	Application Question 2.		>	The M	iller proce	ess for purific	ation of	f reclaimed g	old					
	With the a	oplicatio	on the O	perator	should:									
	1. Supply activitie	•	eral applica	ation requ	uirements	for Section	2.3 on p	bage 29 for tl	nis aspec	t of the				
	Indicative	BAT red	quiremer	nts										
BAT for the Miller	1. Control chlorine		to control t	the temp	erature of	the melt and	d rate o	f lancing to n	ninimise I	oss of				
process	2. A scrub	bing sys	tem to ren	nove vap	orised me	etal chlorides	and m	etal oxides.						
	3. Effectiv	e treatm	ent for scr	ubber eff	luent.									
	2.3.4.3 P	latinum	group n	netals										
BREF 6.1.3	The six platin													
	osmium. Th conductivity,													
SI 1973 Section 2.2 Part A (1) (a)	Hence they a fabrication a	are widel	y used as											
	PGMs occur Russia and (nickel and co	Canada.	Gold and	silver are	e also to l	be found in the	ne same							

Because of the high value of precious metals, recycling is also important and recovers a large proportion from fabricated products.

nickel and copper refining are a primary source of precious metals.

The range of potential raw materials is very wide and techniques must be adapted to the materials to be processed. Not all of the activities described below will be encountered on all sites, and similar ends may be achieved using alternative techniques.

INTRODUC			HNIQU	JES	EN	AISSION	IS	11	MPAC				
Management Mate		vities & tement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues			
metals		ed at insta	allations e	engaged	in recov	order to illuering and pu BAT.							
	Process:	Carbon b ceramic l then subj ceramic o	based cata based cata jected to fu componen	lysts are alysts are urther trea t. Some	incinerate incinerate atment w raw mate	consist of a ed to remove red to remove ith sodium hy erials require t higher pres	e carbor e residu ydroxide more a	n prior to diss al carbon ar or sulphurio ggressive co	solution. d hydroc c acid to o onditions;	Similarly, arbons, and dissolve the in some			
		feedstock Some ray	k, then trea w material	ating it wi s require	th hot hy more age	Dissolution is drochloric ac gressive con cases by the	id throu ditions;	gh which ch in some cas	lorine is p				
		material i their vola scrubbing separate chloride.	including s itile tetroxi g with hydi d by furthe Both are	silver as s des by ox rochloric er <i>selectiv</i> recovere	silver chlo kidising w acid whe /e oxidati d by prec	achieved, the ride. Initially ith sodium b re they are c on and distill ipitation with ous metals is	r, osmiu romate. onverte ation to ammor	m and ruthe These are d to chloride yield a pure nium chloride	nium are recovere s. Osmit solution e. The so	removed as d by um is of osmium olution			
		Gold is removed by <i>solvent extraction,</i> using an oxygenated organic solvent into which the gold is selectively extracted. The gold is subsequently removed from this solvent by reduction, e.g. by oxalic acid.											
		solvent, s a tetram reduced used to r phospha	such as a f mine comp with formic ecover pla	thio-ether blex using c acid to y tinum fro ganic pha	 After e aqueous yield palla m the pa ase is stri 	vent extracti xtraction, pal s ammonia. adium black. Iladium raffin pped with wa	lladium The sol Solven ate, one	is recovered ution is neut t extraction e such solve	from the ralised ar technique nt being f	solvent as nd the salt es are next tributyl			
		subjected	d to an ion	exchang	e proces	ns only rhodi s to extract ir um chloride.							
		treatmen This is th and finall conclude decompo	ts. One si en conver y reduced d by an <i>ig</i> osition, sor	uch route ted to am to rhodiu nition ste ne of whi	involves imonium im black ep, where ch require	the rhodium precipitation hexachlororh using formic e ammonium e an inert or vders or melt	of an a nodate t acid. T salts ar reducin	mmonium n by the action he hydrome nd blacks un g atmospher	itro-rhodi of hydro tallurgica dergo the re. The re	um salt. chloric acid I stages are ermal			
		efficient I provides lead carb the smell precious separate the silver	nydrometa an efficier oonate, can ting operat metals wh and the sl recovered	Ilurgy. Le to route for ton and tion the m tile base is lag is rem d as silve	ead smel or the rec fluxing a nix is hea metals co noved. T r chloride	naterials whe ting is used t overy for silv gents such a ted to 1100° oncentrate in he lead is pa which is ulti onate for re-u	o upgra er. The s soda a C during the slag rted to r mately o	de the mate raw materia ash, lime bo g which lead g. On tappir recover the p	rial which Is are mi rax or sili dissolves og, the lea precious r	a also xed with ca sand. In s silver and ad and slag metals and			
	Water:	compour emission	nd. Waste	liquors a ises, halc	re also g ogens of a	ch of the filtra enerated by s ammonia, int spillage.	scrubbe	ers that are u	sed to pr	event the			
	Air:	involves nitrogen the solve	handling c and nitros nt extracti	hlorine a yl chloride on phase	nd hydrog e are gen s. <i>Decol</i>	ntial to relea gen chloride. lerated. Amr <i>mposition</i> of ogen chloride	When noniaca complex	aqua regia i al solutions a x compound:	s used, o re used i s contain	xides of n some of ing			

IPPC

INTROD	UCTION	TEC	HNIQU	JES	EN	/ISSION	IS	I	MPAC	Т		
Management		vities & tement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Precious metals		atmosph potential	ere used i sources c	n the dec f dust, an	omposition d some of	liquid ammor on furnaces. of the compo ne that conta	<i>Materia</i> unds in	als handling volved are h	operatior	ns are		
	Wastes:	processe		rned for f		p metals mea ocessing in p						
	Energy:	and treat		ems for a	II emissio	nmental impa ons. Energy						
	Accidents	with loss	of contair	ment. M	ajor spilla	re used on-s ages of react n system ma	ion liqui	ds could cau	ise water			
	Noise:	Accident equipme		om vehicl	e moven	ients. Noise	from fa	ns and pollu	tion abat	ement		
	Applicatio Question	on Form 2.3 (cont.)	\geq	Platinun	n group r	netals refine	ry					
	With the	applicati	on the O	perator	should:							
	1. Supp activi		eral applica	ation requ	irements	for Section	2.3 on p	age 29 for th	nis aspec	t of the		
	Indicativ	e BAT red	quiremer	nts								
BAT for platinum	BAT for pla	atinum grou	up metals	refinery s	hould inc	lude						
group metals		tion to proc nable solve		n and sto	ck contro	ol so as to mi	inimise i	inventories o	f liquefie	d gases and		
						se the reage such as amn						
		uate bundi ruptures do				eactors. Effe reas.	ective pi	pework desi	gn to ens	sure that		
		ent liquid si ms to cate			o ensure	tanks are no	ot overfil	led. Compre	ehensive	alarm		
	5. Effec	tive proced	lures for ic	lentifying	solid ma	terials in pro	cess or	awaiting disp	oatch.			
		nature of th				s where pote cognisance c						
	7. Comprehensive treatment systems to minimise emissions into the atmosphere of acid gases, halogens, oxides of nitrogen, ammonia, metallic fume and particulate matter of any description.											
	 Comprehensive treatment systems to ensure that liquid discharges meet acceptable limits before discharge. 											

INTRODUC	TION TECHNIQUES EMISSIONS IMPACT													
		ties & ement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues				
Refractory metals	2.3.5 Re The term "ref titanium, tant	ractory r	netals" inc	ludes chr		nanganese,	tungste	n, vanadium	, molybde	enum,				
BREF Section 8.1.1	Historically, t in the closing no UK produ	decade	s of the 20	th centur	y this pos	sition was lo	st. For	most of these						
	Since many of ferro alloys, a					nufacture of	alloy st	eels, produc	tion is in t	the form of				
	2.3.5.1 C	hromiu	m											
BREF Section 8 SI 1973 Section 2.2 Part A (1) (a) (i)	Process:	ore with fused n chromic more of genera	n sodium o nass with s c acid and ften alumir	arbonate sulphuric finally to nium pow aluminiur	to form of acid to po chromic der. It is m thermit	chromates. roduce sodiu oxide. The o then fired to re reaction is	These a um chro oxide is o initiate	oncentrates lare extracted mate which i then mixed v a thermite re ent to melt the	by leach s first cor with carbo eaction.	ing the overted to on, silicon or The heat				
	Water:		nching ope n potentia		volve aqı	ueous solutio	ons of h	exavalent ch	Iromium V	vith a high				
	Air:		ermite proc llurgical fu			•	hot, with	n a high pote	ntial for d	lischarge of				
	Land:													
	Waste:													
	Accidents:	for envi	ironmental , cannot be	l damage e stopped	in the ev I before r	ent of spillag	ge is hig mplete.	valent chrom h. The therr Any failure gation.	nite stage	e, once				
	Energy:	leachin but nev release	g and eva vertheless e, and also	poration s an energ	stages. 1 y consum	he thermite her by virtue	process of the a	energy input, s itself is mar bsence of co stems to prev	kedly execution l	othermic, neat				
	Noise:	Not sig	nificant.											
	Application Question 2.		>	Chromi	um metal	manufactur	е							
	With the ap	oplicatio	on the Oj	perator :	should:									
	1. Supply activitie	-	eral applica	ation requ	irements	for Section	2.3 on <mark>p</mark>	bage 29 for th	nis aspec	t of the				
	Indicative	BAT req	quiremen	its										
BAT for	1. Effectiv	e control	of the lea	ching and	l concent	rating stage	s in ord	er to prevent	water po	llution.				
manufacture of chromium	2. Prevent	tion of fu	gitive discl	harges in	to the air	during roast	ing and	the thermite	process.					
metal	3. Minimis	ation of s	solid waste	e producti	ion by ore	e pre-treatm	ent.							

INTRODUC												
Management		ities & ement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues		
Refractory metals	This section				-	roduction of	pure m	anganese:				
BREF Section 8.1.2	silico-thealumino-f	rmic red thermic r n of ferro sis of fus	reduction o-mangane sed salts.	ese		d none are c	arried o	ut in the Unit	ted Kingd	om.		
	2.3.5.3 T	ungste	n									
	(a) Produ	iction f	rom ores	or con	centrate	S						
BREF Section 8.1.3 SI 1973 Section 2.2 Part A (1) (a)	Process:	a sodiu paratu by eva blue (V require	um tungste ngstate (A poration a V ₄ O ₁₁) tun es an inert	en solution PT) by ei nd crysta gsten oxio atmospho	n. After r ther solve llisation. des. The ere and te	ntrates by pr emoval of im ent extractior It is then cal yellow oxide emperature to is required	purities or ion cined to forms petween	, this is conv exchange. T make either at 250°C, bu 400°C and	verted to a The APT i r yellow (\ it the blue	s removed NO ₃) or e oxide		
		Reduc		gsten me	tal powde	r is achieved		-	le in hydr	ogen in a		
	Water:	The in	itial leachir	ng phase	will gene	rate a liquid	waste.					
	Air:		eleases ar en evolveo			er dry mater oceeds.	ials are	handled. Ar	nmonia is	consumed		
	Waste:		en is not n			generated by t this stage,						
	Accidents:	Not sig	gnificant.									
	Noise:	Not sig	gnificant.									
	Application Question 2			Tungst	en manuf	acture						
	With the a	nnlicati	ion the O	nerator	should							
		the gen				for Section	2.3 on p	age 29 for t	his aspec	t of the		
	Indicative	BAT re	quiremei	nts								
BAT for	1. Preven	tion of lie	quid losses	s during le	eaching s	tage.						
tungsten manufacture			safe dispo									
manuracture			oss of amm									
			ust losses									

INTRODU	JCTION	TECH	NIQU	JES	EN	IISSION	IS	I	ИРАС	Т			
Management	Vaterials Activi inputs abate		Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues			
Refractory metals	Two major pr	ocesses a	ire availa	able for p	rocessing	-		crap: I ends at bel	ow ombic	t			
BREF Section 8.1.3.2	temperatu are then a • The zinc p the cobalt used. The Tungsten car	ires again air classifie process in matrix. T e zinc is re bide scrap	st a stati ed for re- volves tr he tungs ecovered o, which	onary tar use. eating th sten carb by distill cannot be	get. Adia e carbide ide is left ation. e recover	ibatic cooling rod with mo as a spongy ed by the ab	g shatte Iten zind matrix ove me	rs the embrit c which form which can be thods, can b	tled carb s an ama e crushec	ides which Igam with I and re-			
BREF Section 8.1.4	The only vana	nadium	duced in						on 2.3.6.				
	(a) Roasti	ing sulpl	hide ore	es									
BREF Sections 8.1.5 and 9.1.6.1	Process:							ombination w otation.	vith coppe	er. The ore			
SI 1973 Section 2.2 Part A (1) (i)		is crushed and the molybdenite present separated by flotation. Conversion to molybdenum trioxide is achieved by roasting, which produces sulphur dioxide, which can then be converted into sulphuric acid. Selenium and rhenium may also be present in the concentrate, and if so, will need to be removed from the gases before the acid plant.											
	Water:	Potential	losses a	associate	d with op	eration of ac	id plant						
	Air:	Dust from from filtra			aterial an	d product. L	oss of s	sulphur dioxid	le to air,	and dust			
	Waste:	Gangue	from flota	ation plar	nt, solids f	from filtration	i systen	n and sludge	s from ac	id plant.			
	Accidents:	Loss of e	extraction	n leading	to signific	ant acid gas	release	es.					
	(b) Produ	ction of	molyba	lenum n	netal								
BREF Section 8.1.5.1	Process:		ybdate o					trioxide, am he compoun		drogen			
SI 1973 Section		The proc	ess take	s place i	n two stag	ges:							
2.2 Part A (1) (a)			irst is the at abou		ion of mo	lybdenum di	oxide.	This is exoth	ermic and	d takes			
		count	er-curre	nt rotary	kiln. The		nen be o	hydrogen to compressed a nace.					
	Water:	Water is	not usec	I. Small-	scale pro	cess.							
	Air:	Products	of comb	oustion.									
	Waste:	Negligibl	e.										
	Land:	Negligibl	e.										
	Energy:	The sma	ll scale a	and high t	emperatu	ires make th	e proce	ss relatively	inefficien	t.			
	Accidents:	With hot	hydroge	en there i	s always	a possible in	nplosior	ı risk.					

INTRODU	ICTION	TEC	HNIQL	JES	E١	/IISSIO	IS	IN	ИРАС	Т				
Management		tivities & atement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues				
Refractory metals	Molybden	um and var	nadium cat			ndary mat used in the p		mical industr	y. The m	netals are				
BREF Section 8.1.5.2	 heatin their c 	xides;	00°C to re			bon and hyd	rocarbo	ns – this con	verts the	metals to				
SI 1973 Section Section 4.2 Part A (1) (a) (v)	 separa 	ng out the r ation of mol nent of alum	ybdate and	d vanadat	e;	t and nickel.								
	2.3.5.6	Manufac	ture of ti	tanium										
BREF Section 8.1.6.1 SI 1973 Section	Process:	metal i control	n an inert a	atmosphe rate of a	ere. The ddition of	reaction is e	xotherm	tetrachloride nic and the ra The operating	ite of rea	ction				
2.2 Part A (1) (a)						n magnesium prides of titar		le, unreacted	l magnes	ium,				
		 Purification is achieved by: acid leaching, using dilute nitric or hydrochloric acid; or heating under an inert atmosphere or a vacuum to 1000°C. The titanium sponge can be melted using a consumable electrode vacuum arc melting 												
		The tita		nge can b	e melted	using a con	sumabl	e electrode v	acuum a	rc melting				
	Water:													
	Air:	Planne	d releases	s are insig	nificant.									
	Land/was	ste: As higl	n-purity co	mpounds	are used	l, wastes are	insigni [.]	ficant.						
	Energy:					nagnesium a e extra energ		n-energy" raw eded.	v materia	ls, but apart				
	Accident	fumes		loric acid	and titar	nium dioxide		r moist air to ate. Any acc						
	Applicat Question	ion Form n 2.3 (cont.)	>	Titaniur	n manufa	acture								
	With the	applicati	on the O	perator	should:									
		ply the gene vities.	eral applica	ation requ	irements	for Section	2.3 on p	bage 29 for th	nis aspec	t of the				
	Indicativ	ve BAT re	quiremer	nts										
BAT for titanium		-						anium chlorid	le.					
manufacture	2. Wel	-rehearsed	emergenc	y procedu	ures to ha	andle any ac	cidental	l spillage.						
BREF Section 8.1.7		Tantalun sses define		'3 are in c	peration	in the UK.								
DDEE Soction	2.3.5.8	Niobium		.	-									
BREF Section 8.1.8 SI 1973 Section 2.2 Part A (1) (a)	Process:	thermit		. This is s	similar to			niobium allog manufacture						

INTRODUC														
Wanadement		ties & ment	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues				
Ferro alloys BREF Section 9	2.3.6 Fe A major use alloy steels.	of eleme Since it i	nts such a s frequent	ly more o	convenien	t to add the								
SI 1973 Section 2.1 Part A (1) (b)	reductant	pon the r ing: - metal c oxide ar	aw materi xide/ore v nd slag.	als used, vith iron c	the proce pre/scrap	ess can be c plus reducta	ant com	d as primary bine to yield ⁻						
	Seconda The reductar	-				s ferro alloy		Im						
		erro-chi		useu are	carbon, s			um.						
BREF Section 9.1.1	Process:	The ma oxides.	iin raw ma For high	carbon fe	erro-chror		t freque	a mixture of i ntly used), co ion.						
			charging t ed then sir		ctric arc r	eduction furi	nace the	e raw materia	als are no	rmally				
			smelting, t at regular				and ferro	o-chrome an	d slag mu	ist be				
								e, which can processes.	be drawn	off, cleaned				
		then burned as a fuel for pre-heating, drying or similar processes. The slag can be crushed to recover entrained metal, and is sufficiently inert to permit its use in civil engineering.												
	Water:	Signific	ant quanti	ties are u	ised as co	polant and fo	or slag g	granulation.						
	Waste:					ainly influen and filtratior		the quantity o lusts.	of ores us	ed. Other				
	Air:		s significaı I handling				portant	potential sou	rces bein	g raw				
	Energy:		a high-ene use, name		ation and	there are si	gnifican	t opportunitie	es for min	imising				
						ise slag ger gases prior f								
			-			by using ext		-						
	Noise:	Arc furr	naces can	be very r	noisy.									
	Application Question 2.		>	Ferro-c	hrome pr	oduction usi	ng arc f	urnace techr	nology					
	With the ap 1. Supply activitie	the gene				for Section	2.3 on p	bage 29 for th	nis aspec	t of the				
	Indicative I	BAT req	uiremen	its										
BAT for ferro-	1. Effective	e control	over raw	materials	acquisitio	on to ensure	optimu	m feed to the	e process	i.				
chrome production	2. Dust co	ntrol.												
using arc furnace	3. Fume c	ontrol.												
technology	4. Energy	recovery	'.											
	5. Minimis	ation of v	waste-wate	er genera	ation.									

INTROD	UCTION	TEC	HNIQL	JES	EN	/ISSIO	1S	IMPACT		
Management		vities & ement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Ferro alloys	2.3.6.2	Other fer	ro alloys	;						
	Ferro silicon, ferro-manganese, ferro-nickel and ferro-boron can be produced using arc furnace technology, which is substantially similar to that described in Section 2.3.6.1.							ace		
BREF Section 9.1.7.1.1	Ferro-tungsten can also be made in an arc furnace, though due to its high melting point, it cannot be removed by tapping. Instead, when the amount of metal has reached the desired weight, the furnace is switched off and allowed to cool. The refractory is removed and the ingot taken for crushing and screening.							ne furnace		
	2.3.6.3 Production of ferro-manganese in blast furnaces									
BREF Section 9.1.3.3.1	Process:	manufa	acture pig i	iron. The	e reaction	sequence a	nd heat	es which rese balance are formed by n	more co	mplex as a
			e top. The					ced by carbo relatively hig		
		Conse	quently the	e furnace	requires	much more	coke or	irkedly endot supplementa used on pre-	ary fuel th	
	Energy:		nese in a l					ments, produ ered BAT wh		
	No other iss	ues cons	idered.							

No other issues considered.

INTRODU	CTION	TEC	HNIQU	JES	EN	AISSION	IS	1I	MPAC	Т
Mananement		ities & ement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
										100000
Alkali metals	2.3.7 All	cali an	d alkali	ne ear	th met	als				
		odium I								
BREF Section 10.1.1	Process:	Process: Sodium metal is produced by the electrolysis of fused sodium chloride. The first stage in the process is the preparation of pure salt from crude salt. Calcium and barium chlorides are added to the electrolyte to reduce the melting point and to increase its density.								
SI 1973 Section 2.2 Part 1(1) (a)		cathod	e by a stee	el mesh d	liaphragm	n. The purpo	ose of th	s separated f ne diaphragn nlorine and se	n is to mir	
		Chlorine is drawn off at the surface, liquefied and stored in storage tanks prior to sale. The sodium also rises to the surface of the melt, and is gathered by a collector assembly from which it runs into receivers which are tapped at regular intervals.								
		Small amounts of calcium are also formed, and this solidifies from the molten sodium as the latter cools.								
		After the calcium has been removed by filtration, the molten sodium is transferred to hot storage tanks pending use on the chemical process, or sale.								
	Water:							It by dissolut		coming salt.
	Air:	Air: Large quantities of chlorine are generated and stored during this process.								
		Molten sodium in contact with air fumes violently and is prone to fire.								
		Waste: Not significant.								
	Energy:	Energy: The process is energy-intensive, involving a dissolution and recrystallisation stage as well as the electrolysis step.								
	Noise:									
	Accidents: Chlorine leaks and spillages. Spillages of molten sodium. Fires.									
	Application Question 2		>	Sodium	n metal					
	With the a	oplicati	on the O	perator	should:					
	1. Supply activitie	•	eral applica	ation requ	uirements	for Section	2.3 on p	bage 29 for th	his aspec	t of the
	Indicative	RAT roa	nuiromor	ite						
BAT for sodium netal	1. Operati	ng and n	·	ce proced	lures sha	II be devised	l so as f	to minimise t	he amour	nt of sodium
		e collecti roductior	•	e and har	ndling sys	tems shall c	omply v	vith the requi	rements f	or chlor –
	3. Emerge	ency pow	ver supplie	s for chlo	rine extra	action fans.				
	4. Emerge shut-do	-	ubbing sys	tems cap	able of al	bsorbing chlo	orine pr	oduction to a	llow for e	mergency
			fire and e	mergenc	y procedu	ires with pro	vision f	or regular an	d realistic	drill.
	2.3.7.2 L	ithium								
BREF Section	Lithium meta	l is also	produced	by fused	salt elect	rolysis using	Downs	cells.		
10.1.2	Apart from u	-							-	
	Lithium is mutering to prevent ox		reactive ar	nd can be	e skimme	d off and cas	st into m	oulds. Thes	e are cov	ered with oi
	Much smalle sufficient to v									
	There are no requirements						duced.	Otherwise th	e factors	and

		TECHNIQ			IISSION	19	IN	ЛРАС	т	
INTROD		TECHNIQ ities & Ground			1133101	13	IN	NPAC	I Installation	
Management		ement water	Waste	Energy	Accidents	Noise	Monitoring	Closure	issues	
	2272									
Alkali metals		lagnesium					24.			
	-	is manufactured b			-			T 1 1	1	
BREF Section 10.1.5.2		Magnesium hydroxide is first prepared by slaking calcined dolomite with sea water. The hydroxide is then recovered by concentrating the solids to a pulp in a thickener, then drying and calcining in a rotary kiln.								
10.1.3.2		The magnesium oxide is then mixed with carbon and pelletised prior to charging into chlorinator towers. These are brick-lined shaft furnaces into the base of which chlorine from the electrolysis is								
		are run at tempera nd transferred via						oride, wh	ich is	
		From the cells, molten magnesium is withdrawn and transferred in closed cells to the cast house where the metal is cast as ingot or alloy. Sulphur hexafluoride can be used to prevent oxidation during casting.								
		Off-gases from the chlorination furnaces are scrubbed to remove chlorine and passed through precipitators to remove dusts before finally being subject to incineration.								
	Water:	r: Contaminated water from the chlorinator off-gas treatment plant will require treatment before discharge.								
	Air:	Dusts arising fro	om the calo	cination p	rocess.					
		Chlorine from th	e electroly	sis proce	SS.					
		Magnesium oxic	le from mo	olten meta	al handling.					
	Waste:	Not significant.								
	Noise:	Not significant.								
	Energy:	Preparation of n	nagnesium	n chloride	has a high e	energy o	demand, as d	loes its e	lectrolysis.	
	Accidents:	Chlorine gas is l	peing gene	erated in	quantity and	transfe	rred between	stages.		
		Molten magnesi	um is bein	g handle	d.					
	Application Question 2		Magne	sium met	al					
	With the a	pplication the C	Operator	should:						
	1. Supply activitie	the general applices.	cation requ	uirements	for Section 2	2.3 on p	bage 29 for th	nis aspec	t of the	
	Indicative	BAT requireme	nts							
BAT for		n must be conside		essina RA	T are:					
magnesium		effluents from the		•						
metal		ts of combustion a				m the c	alcination pro	cess.		
		o omissions of chl	orino from	the elect	rolutio roduci	tion nra	0000			

- 3. Fugitive emissions of chlorine from the electrolytic reduction process.
- 4. Fugitive emissions of magnesium oxide from metal transfer and casting operations.
- 5. Emergency systems to prevent losses of chlorine during power failure.

2.3.7.4 Calcium and strontium metal

Small quantities of calcium or strontium metal can be manufactured by a metallo-thermic reaction using the relevant oxide and aluminium powder.

The oxide is ground and mixed with aluminium powder, then pressed into briquettes.

These are heated to 1300°C in a vacuum furnace, at which temperature the calcium (or strontium) distils off and can be condensed in the cooler part of the furnace.

A very small-scale process with limited potential for pollution.

BREF Section

10.1.4

		N TEC	HNIQUES		EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Nickel 2.3.8 Nickel

Nickel is produced from oxide or sulphide ores, with about 60% of production coming from the latter.

Neither ore is processed in the United Kingdom, but the carbonyl process is used to refine an imported oxide prepared by smelting sulphide ore.

2.3.8.1 The carbonyl process

BREF Section 11.1.3.5

Process: Impure nickel oxide is first reduced to nickel powder using hydrogen. It is activated and then exposed to carbon monoxide at near ambient temperature and pressure, when it reacts to form nickel carbonyl. The carbonyl is volatile and easily separated from the residue. It is also thermally unstable, and when warmed it readily decomposes to yield nickel and carbon monoxide. Depending upon the structure of the reactor and the reaction temperature, the product can be obtained as pellets, as powder, or as a nickel coating on other materials.

The recovered carbon monoxide is recycled to manufacture more nickel carbonyl, with a small bleed to waste to prevent a build-up of inert gases. This bleed gas must be incinerated to destroy the nickel carbonyl and carbon monoxide present. It must then be filtered to remove nickel dust.

- *Water:* Provided that water scrubbers are not used to scrub exhaust gases, the process does not give rise to a water effluent.
- *Waste:* The residue from the carbonyl extraction stage consists mainly of nickel oxide, but contains sufficient precious metals to warrant further processing. This material is not discarded but transferred for further processing.
- Air: Nickel carbonyl is highly toxic and also a carcinogen.
- Significant quantities of dusty raw materials are handled.
- *Energy:* The carbonyl process is energy-efficient by virtue of the relatively low operating temperatures required.
- Accidents: A major release of carbonyl could have serious health and environmental consequences.

Application Form Question 2.3 (cont.) Nickel by the carbonyl process

With the application the Operator should:

1. Supply the general application requirements for Section 2.3 on page 29 for this aspect of the activities.

Indicative BAT requirements

- 1. Minimisation of carbonyl process inventory to minimise potential accident impact.
- 2. Back-up and redundancy in gas handling systems to minimise risk of leaks arising from breakdown.
- 3. Back-up power supplies for safety and environmentally critical items.
- 4. Effective incineration and filtration systems for handling waste gases.
- 5. Comprehensive gas leak detection systems.
- 6. An effective emergency control system with regular and realistic exercises.

BAT for nickel by the carbonyl process

INTRODUC		TEC						IN		т	
Ma		ities &	HNIQU Ground			/ISSION			MPAC	I Installation	
Manadement		ement	water	Waste	Energy	Accidents	Noise	Monitoring	Closure	issues	
Carbon & graphite		e of carbo	n electrod	es is inclu	uded with	in this note t		e a major pro th installatior			
BREF Section	Section 2.2	installations where this activity is carried on are directly associated with installations defined within Section 2.2 Part A (1). This was recognised by the authors of the BREF Note, who included the manufacture of carbon and graphite electrodes within the Non-Ferrous Metals BREF.									
SI 1973 Sections 6.2 and 6.3		Carbon electrodes and furnace linings are used for a variety of metal production processes, in particular primary aluminium smelting, making ferro alloys and steel-making.									
0.2 010 0.0	The most fre	equently u	ised raw r	naterials	are petro	leum coke, a	nthracit	te, pitch and	bitumen.		
	2.3.9.1 N	lanufact	ure of p	re-bake	d carbo	n anodes f	or alur	ninium ma	nufactu	re	
BREF Section	Process:							nanufactured n to act as a l		ixture of	
12.1.1.2		sizes w	hich on bl	ending w	ill genera	ite a mixture	of maxi		. Materia	al recovered	
SI 1973 Section 6.3 Part A (1) (a) (ii)		sizes which on blending will generate a mixture of maximum density. Material recovered from used anodes is similarly treated and added to the blend prior to mixing with the molten pitch. This mixture is referred to as green electrode paste. After mixing "green" electrodes are formed by pressing then transferred to a ring furnace for baking.								ng "green"	
	The ring furnace consists of a series of refractory lined pits into which the anod stacked then covered with loose coke to prevent oxidation. Coke is also placed the layers of anodes to prevent them from sticking together.										
		rearran sequen hotter p remove	ged on a ce, and a its. The s d and pits	regular ba lso allows system ai s recharge	asis. This newly cl ds fuel eo ed whilst	s permits dire harged pits to conomy and other pits are	ection o o be wa also allo e heatin	gned so that f hot gases to rmed by exh ows for bake g, soaking of ume into the	o groups aust gase d anodes r cooling.	of pits in es from a to be	
	Water:					closed circuit Id a fire occu		ns, is not sigr	nificant. N	Main risk	
	Waste:							e during main ble to recycle			
	Air:	All oper	ations inv	olving ho	t pitch ar	e potential s	ources	of fume.			
		All trans emissio		itions of p	etroleum	coke are po	tential s	sources of gr	it and du	st	
			me evolve the chim		baking n	nay not be co	omplete	ly burned be	fore discl	narge	
		emissio	ons throug	h the bak	e furnace		s fluoride	to cause sig e can also da e.			
		baked a	anode. W	hen the a	node is d			ent in raw ma lociated alum			
	Noise:	Is not s	ignificant.								
	Energy:	These p	processes	are high	energy ι	isers.					
	Accidents:	Hot pito	h represe	ents a sigr	nificant fir	e hazard.					

INTRODUC	CTION TECHNIQUES EMISSIONS IMPACT										
Manadement	terials Activities & Ground water Waste Energy Accidents Noise Monitoring Closure Installation issues										
Carbon & graphite	Application Form Question 2.3 (cont.)										
	With the application the Operator should:										
	1. Supply the general application requirements for Section 2.3 on page 29 for this aspect of the activities.										
	Indicative BAT requirements										
BAT for carbon bake	1. Minimise sulphur content of raw materials.										
activities associated	2. Effective cleaning of returned anode butts.										
with aluminium	3. Planned maintenance cycles for the ring furnace refractory.										
processes	4. Effective fire prevention and response systems.										
	5. Treated ring furnace exhaust gases to remove fluorides before discharge into the air.										
	2.3.9.2 Manufacture of graphite electrodes										
BREF Section 12.1.1.3	 Process: Manufacture of graphite electrodes is a two-stage process. The first stage is very similar to that used to manufacture pre-baked anodes, except that: recycled raw materials are not normally used, and 										
SI 1973 Section 6.2 Part A (1) (a)	 furnaces other than ring furnaces may be used. After baking at around 1100°C, the electrodes are allowed to cool, then impregnated with 										
	pitch before heating in the graphitising furnace to approximately 2800°C.										
	This temperature is achievable using electric resistance furnaces.										
	The carbon shapes are loaded into the furnace and buried with thermally insulating packing, usually coke or silicon carbide to prevent oxidation during the process. Heat is provided by passing electric current through the shape (Castner furnace) or through both shape and packing (Acheson furnace).										
	The heating cycle takes up to 7 days, with as many as 14 days to cool before the insulation can be removed.										
	After graphitisation, the shapes are finished by turning, drilling and milling according to the customer's requirements.										
	Dusts produced by these activities may be re-used in the blend, or used as insulation material.										
	The environmental comment relevant to anode bake furnaces apply, with the exception of those relating to use of fluoride contaminated recycle material.										
	Additional factor: Dust from machining activities.										
	2.3.9.3 Søderberg electrodes										
BREF Section 12.1.1	There are two forms of Søderberg electrodes: pre-formed electrodes or paste. Both are self-baking electrodes.										
	Pre-formed electrodes are made from Søderberg paste which is pressed into cylindrical moulds . The cylinders are fed into the furnace through steel tubes which act as the electrical contact. The electrodes bake as they near the furnace and seal against the tubes, preventing the emission of baking gases. This type of electrode is used in a variety of electric arc furnaces.										
	Paste electrodes consist of a former suspended immediately above the furnace and are used exclusively for primary aluminium production. Green electrode paste (see Section 2.3.9.1 above) is charged into the former, where it is slowly hardened and baked by the heat of the furnace below. As the electrode is consumed from below, so more paste is added above. Electrical contact is achieved by conductor stubs driven into the sides of the electrode.										

For aluminium electrolysis, the problems of pitch fume generation associated with Søderberg anodes are such that they are not regarded as BAT.

INTRODU		TECH	NIQL	JES	EN	IISSION	IS	IN	ЛРАС	Т
Wanadement			Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Carbon & graphite		pecial cai								
3. april 10	A range of pl graphite elec		ch as sea	als, brusr	ies and c	rucibles are	produce	ed in a simila	r way to	carbon and
	Potential em	Potential emissions are similar, as is BAT. However, certain special processes may be encountered.								
	(a) High-	purity gra	phite n	nanufac	ture					
	Process:	As graphi	ite manu	Ifacture, I	out modif	ied to remov	e incluc	led impurities	s such as	metals.
BREF Section 12.1.2		chlorine a volatile at	and fluor the ope	ine. The erating ter	se gases nperature	react with me. The gase	netallic i s are co	nd decomposed of the second of	form salts remove	s which are
	Water:	Not signif	icant.							
	Waste:	Not signif	icant.							
	Noise:	Not signif	icant.							
	Air:	Loss of fr	eon, dus	sts and a	cid gases	i.				
	Accidents:	Not signif	icant.							
	As this proce provides a u				of freons	of the type r	ecovere	ed from refrig	erators, i	t incidentally
	Application Question 2.3			High-pu	irity grapł	nite				
	With the a	oplication	the O	perator	should:					
	1. Supply activitie		l applica	ation requ	irements	for Section	2.3 on p	bage 29 for th	nis aspec	t of the
BAT for high- purity	Indicative	BAT requ	iremen	its						
graphite	1. No add	itional requ	irement							

		HNIQUES		EMISSIONS			IMPACT				
Management	Materials inputs	Act aba	ivities & atement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Management	inputs abatement water Waste Energy Accidents Noise Monitoring Closure issues
Abatement to air	2.3.10 Abatement of <i>point source</i> emissions to air
Sources	The nature and source of the emissions expected from each activity are given in previous sections and will be confirmed in detail in the Operator's response to Section 3.1.
	In general they comprise the following: Primary exhausts from furnaces Collection hoods around charging doors, tap hole and launders Casting stations and flame cutting locations Raw material conveyor transfer points Storage tank and silo vents Ladle pre-heating stations Exhausts from slag coolers Exhausts from crushing, screening and grinding plant Exhausts from chemical processes. Cross-sectoral guidance on abatement techniques for point source emissions to air can be found in Ref. 12.
	Application Form Question 2.3 (cont.) Control of point source emissions to air
BREF Sections 2.7, 2.8, 2.17	
Subsections 4 of Chapters 3 to 12 describe the nature of the emissions and BAT for the Sectors	 With the application the Operator should: 1. Supply the general application requirements for Section 2.3 on page 29 for control and abatement equipment; and in addition 2. Describe the measures taken to demonstrate the reliable operation of control and abatement systems. These shall include as appropriate: temperature and pressure monitors on filtration plant and associated ductwork power consumption indicators on fans associated with extraction systems temperature monitoring on exhausts from furnaces and after-burners where wet scrubbing systems are used, liquor flowrate pH monitors. 3. Where VOCs are released, the identification of the main chemical constituents of the emissions and assessment of the fate of these chemicals in the environment. These steps will be carried out as in response to Sections 3.1 and 4.1 but need to be understood here in order to demonstrate that the controls are adequate.
	 Indicative BAT requirements The Operator should complete any detailed studies required into abatement or control options (see item 3 in Section 2.3) as an improvement condition to a timescale to be agreed with the Regulator but in any case within the timescale given in Section 1.1. Steam plume elimination – Releases from wet scrubber vents should be hot enough to avoid visible plume formation in the vicinity of the vent. This is to prevent the condensation or

visible plume formation in the vicinity of the vent. This is to prevent the condensation or adsorption of environmentally harmful substances by the condensing water vapour. Exhaust gases from a wet scrubber can be heated by the use of waste heat to raise the temperature of the exhaust gases and prevent immediate condensation on the exit from the vent. This procedure also aids the thermal buoyancy of the plume. Where there is no available waste heat and the vent contains no significant environmentally harmful substances, the applicant may be able to demonstrate that the BAT criteria have nonetheless been met.

Cont.

			HNIQUES		EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Abatement to air

3. Summary of sources of	f potential releases to air a	nd the BAT treatment/abatement options
Process stage	Component in off-gas	Treatment method
Materials handling and storage	Dust and metals	Correct storage, handling and transfer. Dust collection and fabric filter if necessary
Grinding, drying	Dust and metals	Process operation. Gas collection and fabric filter
Sintering/roasting, Smelting,	VOCs, dioxins	After-burner, adsorbent or activated carbon addition
Converting, fire refining	Dust and metal compounds Carbon monoxide Sulphur dioxide	Gas collection, gas cleaning in fabric filter, heat recovery After-burner if necessary Sulphuric acid plant (for sulphidic ores) or scrubber
Slag treatment Leaching and chemical	Dust and metals Sulphur dioxide Carbon monoxide Chlorine	Gas collection, cooling and fabric filter Scrubber After-burner Gas collection and re-use, wet chemical
refining	Onionine	scrubber
Carbonyl refining	Carbon monoxide, hydrogen	Sealed process, recovery and re-use. After-burner and dust removal in fabric filter for tail gas
Solvent extraction	VOC (depends on the solvent used and should be determined locally to assess the possible hazard)	Containment, gas collection, solvent recovery. Carbon adsorption if necessary
Thermal refining	Dust and metals Sulphur dioxide	Gas collection and fabric filter Scrubber if necessary
Molten salt electrolysis	Fluoride, chlorine, PFCs	Process operation. Gas collection, scrubber (alumina) and fabric filter.
Electrode baking, graphitisation	Dust, metals, SO ₂ , fluoride, PAHs, tars	Gas collection, condenser and EP, after- burner or alumina scrubber and fabric filter. Scrubber if necessary for SO ₂
Metal powder production	Dust and metals	Gas collection and fabric filter
	Dust, Ammonia	Gas collection and recovery. Acid medium scrubber
High-temperature reduction	Hydrogen	Sealed process, re-use
Electro-winning	Chlorine, acid mist	Gas collection and re-use, wet scrubber, de-mister
Melting and casting	Dust and metals VOCs, dioxins (organic feed)	Gas collection and fabric filter After-burner (carbon injection)
may contain hot parti		c filter where hot gases, or gases which electrostatic precipitators would be used in ant or for wet gases.

INTRODU	JCTION TECHNIQUES EMISSIONS IMPACT								
Management	MaterialsActivities & abatementGround waterWasteEnergyAccidentsNoiseMonitoringClosureInstallation issues								
	inputs abatement water issues								
Effluent treatment	2.3.11 Abatement of <i>point source</i> emissions to surface water and sewer								
troutmont	The nature and source of the emissions expected from each activity are given in previous sections and will be confirmed in detail in the Operator's response to Section 3.1. In general, waste-water can arise								
	from the process activity, from storm water, from cooling water, from accidental emissions of raw								
	materials, products or waste materials, and from fire fighting. In addition to the BREF and the techniques below, guidance on cost-effective effluent treatment techniques can be found in Envirowise								
	Guides (Ref. 10).								
	The use of wet scrubbers to prevent or minimise emissions to air must be carefully reviewed and considered against the overall environmental impact of the activity. The presumption should be that if								
	there are no pressing reasons to the contrary, then dry systems should be adopted, even where this leads to a marginal increase in emissions into the air.								
	Application Form Question 2.3 (cont.)								
BREF Sections 2.9	With the application the Operator should:								
and 2.17	1. Supply the general application requirements for Section 2.3 on page 29 to prevent or reduce point								
	source emissions to water and land.								
	 Include, where appropriate, off-site treatment in the description of the waste-water treatment system for the activity. 								
	3. Provide, where effluent is discharged, a justification for not cleaning the effluent to a level at which it can be re-used (e.g. by ultrafiltration where appropriate).								
	4. Describe measures taken to increase the reliability with which the required control and abatement performance is delivered (the measures taken to ensure that any decline in quality of the effluent								
	are promptly identified, e.g. continuous measurement, record and alarm systems associated with simple parameters such as temperature, pH or turbidity).								
	5. Identify the main chemical constituents of the treated effluent (including the make-up of the COD)								
	and assessment of the fate of these chemicals in the environment. These steps will be carried out as in response to Sections 3.1 and 4.1 but need to be understood here in order to								
	demonstrate that the controls are adequate. This applies whether treatment is on- or off-site.								
	6. Identify the toxicity of the treated effluent (see Section 2.10). Until the Regulator's toxicity guidance is available, this should, unless already in hand, normally be carried out as part of an								
	improvement programme.								
	 Where there are harmful substances or levels of residual toxicity, identify the causes of the toxicity and the techniques proposed to reduce the potential impacts. 								
	 Consider whether the effluent flow is sufficient to fall within the requirements of the Urban Waste Water Treatment Directive. 								
	Indicative BAT requirements								
	1. The Operator should complete any detailed studies required into abatement or control options								
	(see item 3 in Section 2.3) as an improvement condition to a timescale to be agreed with the Regulator but in any case within the timescale given in Section 1.1.								
	2. The following general principles should be applied in sequence to control emissions to water:								
	 water use should be minimised and waste water re-used or recycled (see Section 2.2.3); contamination risk of process or surface water should be minimised (see Section 2.3.13); 								
	ultimately, surplus water is likely to need treatment to meet the requirements of BAT (and								
	statutory and non-statutory objectives). Generally, effluent streams should be kept separate, as treatment will be more efficient. However, the properties of dissimilar waste streams								
	should be used where possible to avoid adding further chemicals, e.g. neutralising waste acid and alkaline streams. Also, biological treatment can occasionally be inhibited by concentrated								
	streams, while dilution, by mixing streams, can assist treatment;								
	 systems should be engineered to avoid effluent by-passing the treatment plant. 								

INTROD		ECHNIQUES	EMISSIONS	IMPACT								
Management	Materials Activitie	W/aste	Energy Accidents Noise	Monitoring Closure Installation								
Ŭ	inputs abatem	ent water Water		issues								
Effluent treatment	(see Section reasonable this will be	3. All emissions should be controlled, as a minimum, to avoid a breach of water quality standards (see Sections 3.2 and 4.1) but noting that where BAT can deliver prevention or reduction at reasonable cost it should do so (see Section 1.1). Calculations and/or modelling to demonstrate this will be carried out in response to Section 4.1. BAT for effluent treatment is the precipitation of metals using hydroxide, sulphide or a combination of these reagents.										
Chapter 12.17.7	water shou subject to out. Furth emission o	4. BOD is not normally an issue in this sector but if it is present then the nature of the receiving water should be taken into account. However, in IPPC the prevention or reduction of BOD is also subject to BAT and further reductions which can be made at reasonable cost should be carried out. Furthermore, irrespective of the receiving water, the adequacy of the plant to minimise the emission of specific persistent harmful substances must also be considered. Guidance on treatment of persistent substances can be found in references (see Ref. 13).										
	particular o	particular demonstrating that:										
	emissio	 the treatment provided at the sewage treatment works is as good as would be achieved if the emission was treated on-site, based on reduction of load (not concentration) of each substance to the receiving water; 										
	pumpir	 the probability of sewer by-pass, via storm/emergency overflows or at intermediate sewage pumping stations, is acceptably low; 										
	activitie	 action plans in the event of by-pass, e.g. knowing when by-pass is occurring, rescheduling activities such as cleaning or even shutting down when by-pass is occurring; 										
		 a suitable monitoring programme is in place for emissions to sewer, taking into consideration the potential inhibition of any downstream biological processes and action plans for any such event. 										
	6. Summary											
	Source of waste water	Associated process	Minimisation metho	ds Treatment methods								
			Minimisation method Return to process as far a possible									
	waste water	Alumina production, lead–acid battery breaking,	Return to process as far a	s Neutralisation and precipitation, electrolysis g Settlement								
	waste water Process water Indirect cooling water Direct cooling	processAlumina production, lead-acid battery breaking, picklingFurnace cooling for most metals, electrolyte cooling for ZnAl, Cu, Zn castings,	Return to process as far a possible Use of sealed or air coolin system. System monitoring to dete leaks Settlement,	s Neutralisation and precipitation, electrolysis g Settlement ct Settlement,								
	waste water Process water Indirect cooling water	processAlumina production, lead-acid battery breaking, picklingFurnace cooling for most metals, electrolyte cooling for ZnAl, Cu, Zn castings, carbon electrodesCu, Ni, Pb, Zn, precious metals,	Return to process as far a possible Use of sealed or air coolin system. System monitoring to dete leaks	s Neutralisation and precipitation, electrolysis g Settlement ct								
	waste water Process water Indirect cooling water Direct cooling water Slag	processAlumina production, lead-acid battery breaking, picklingFurnace cooling for most metals, electrolyte cooling for ZnAl, Cu, Zn castings, carbon electrodesCu, Ni, Pb, Zn,	Return to process as far a possible Use of sealed or air coolin system. System monitoring to dete leaks Settlement, closed cooling system Sealed system. Electro-winning of electroly	s Neutralisation and precipitation, electrolysis g Settlement ct Settlement, precipitation if needed Settlement, precipitation if needed Neutralisation and								
	waste water Process water Indirect cooling water Direct cooling water Slag granulation	processAlumina production, lead-acid battery breaking, picklingFurnace cooling for most metals, electrolyte cooling for ZnAl, Cu, Zn castings, carbon electrodesCu, Ni, Pb, Zn, precious metals, ferro alloys	Return to process as far a possible Use of sealed or air coolin system. System monitoring to dete leaks Settlement, closed cooling system Sealed system.	s Neutralisation and precipitation, electrolysis g Settlement ct Settlement, precipitation if needed Settlement, precipitation if needed Neutralisation and								
	waste waterProcess waterIndirect cooling waterDirect cooling waterSlag granulationElectrolysisHydro- metallurgy (blow-down)Abatement system (blow- down)	processAlumina production, lead-acid battery breaking, picklingFurnace cooling for most metals, electrolyte cooling for ZnAl, Cu, Zn castings, carbon electrodesCu, Ni, Pb, Zn, precious metals, ferro alloysCu, Ni, ZnZn, CdWet scrubbers, wet EPs and scrubbers for acid plants	Return to process as far a possible Use of sealed or air coolin system. System monitoring to dete leaks Settlement, closed cooling system Sealed system. Electro-winning of electroly bleed Sealed system Re-use of weak acid strea possible	s Neutralisation and precipitation, electrolysis g Settlement ct Settlement, precipitation if needed Settlement, precipitation if needed yte Neutralisation and precipitation settlement, precipitation if needed Settlement, precipitation if needed								
	waste waterProcess waterIndirect cooling waterDirect cooling waterSlag granulationElectrolysisHydro- metallurgy (blow-down)Abatement system (blow-	processAlumina production, lead-acid battery breaking, picklingFurnace cooling for most metals, electrolyte cooling for ZnAl, Cu, Zn castings, carbon electrodesCu, Ni, Pb, Zn, precious metals, ferro alloysCu, Ni, ZnZn, CdWet scrubbers, wet EPs and scrubbers for acid	Return to process as far a possible Use of sealed or air coolin system. System monitoring to dete leaks Settlement, closed cooling system Sealed system. Electro-winning of electroly bleed Sealed system Re-use of weak acid strea	s Neutralisation and precipitation, electrolysis g Settlement ct Settlement, precipitation if needed Settlement, precipitation if needed vte Neutralisation and precipitation and precipitation if needed ms if Settlement, precipitation if needed settlement, precipitation if needed Settlement, precipitation if needed ins to I raw astes nate								

INTRODUC	JCTION TEC		HNIQUES		EMISSIONS			IMPACT		
Manadement	iterials iputs		Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Fugitive emissions	 2.3.12 Control of <u>fugitive</u> emissions to air At most installations where non-ferrous metals processes are carried out, fugitive emissions to air give rise to a significant proportion of the overall environmental impact and should be reduced as far as practicable. Potential sources of fugitive emissions to air which require particular attention are: handling and storage of dusty raw materials such as concentrates, foundry sand, drosses smelting and melting and refining furnaces and their associated extraction systems transfer operations involving molten metal casting and associated activities handling, storage and disposal of wastes such as drosses, slags and skimmings fume treatment plant, particularly handling collected dusts 									
	Арр	adle heating station plication Form estion 2.3 (cont.)	ons.	Fugitive	e emissio	ns to air				
	 With the application the Operator should: Supply the general application requirements for Section 2.3 on page 29 for control of fugitive emissions to air. Identify and, where possible, quantify significant fugitive emissions to air from all relevant sources (including potential sources identified above). Estimate the proportion of total emissions which are attributable to fugitive releases for each substance. These steps will be carried out in response to Section 3.1 but need to be understood here in order to demonstrate that the controls are adequate. 									
	 Indicative BAT requirements The Operator should complete any detailed studies required into abatement or control options (see item 3 in Section 2.3) as an improvement condition to a timescale to be agreed with the Regulator but in any case within the timescale given in Section 1.1. Where there are opportunities for reductions, the Permit may require the updated inventory of fugitive emissions to be submitted on a regular basis. 									
										ntory of
	 3. Dust and fume – The following general techniques should be employed where appropriate: covering of skips and vessels avoidance of outdoor or uncovered stockpiles where outdoor stocking is unavoidable, use of sprays, binders, stockpile management techniques, windbreaks, etc. wheel and road cleaning (avoiding transfer of pollution to water and wind blow) closed conveyors, pneumatic conveying (noting the higher energy needs), minimising drops regular housekeeping sealing of furnaces and reactors minimising open molten metal transfers adequate extraction of process exhausts designed for the maximum rate of emission. 									
	4.	 VOCs – The requirements subsurface f the use of value on being er an enclosed 	or other vo illing via fil apour bala nptied, or	olatile liqu ling pipes nce lines	uids, the f s extende that trans	following tec ed to the bott sfer the vapo	hniques com of tl our from	s should be e ne container, the containe	mployed:	
	5.	Vent systems sh and, where relev	/ant, shoul	ld be fitte	minimise d with kn	breathing er ock-out pots	mission and ap	s (e.g. pressi propriate aba	ure/vacuu atement e	im valves) equipment.
	6.	Odour – See Se	ection 2.3.	14.						

INTRODUCTIO		N TECHNIQU		JES	S EMISSIONS			IMPACT			
	terials puts	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues	
Fugitive emissions2.3.13 Control of <i>fugitive</i> emissions to surface water, sewer and groundwater											
	Application Form Question 2.3 (cont.) Fugitive emissions to water										
BREF Sections 2.4, 2.9 and	With the application the Operator should:										
2.17	 Supply the general application requirements for Section 2.3 on page 29 for control of fugitive emissions to air. 										
	 emissions to air. Identify and where possible quantify any significant fugitive emissions to water from all relevant sources, estimating the proportion of total emissions which are attributable to fugitive releases the each substance; these steps will be carried out as in response to Section 3.1 but need to be understood here in order to demonstrate that the controls are adequate. 									releases for	
	Indi	icative BAT req	uiremen	nts							
	 Where there are opportunities for reductions, the Permit may require the updated inventory of fugitive emissions to be submitted on a regular basis. 										
	2.	Subsurface str		-							
	 establish and record the routing of all installation drains and subsurface pipework; 										
		 identify all subsurface sumps and storage vessels; engineer systems to ensure leakages from pipes, etc., are minimised and, where these occur, 									
		can be readi	y detecte	d, particu	larly whe	re hazardous	s (e.g. li	sted) substa	nces are	involved;	
		 provide, in particular, secondary containment and/or leakage detection for such subsurface pipework, sumps and storage vessels; 									
		 establish an pressure test 							tructures,	e.g.	
	3.	Surfacing – the	Operato	r should.	:						
	 describe the design (relevant information may include as appropriate: capacities; thicknesses; falls; material; permeability; strength/reinforcement; resistance to chemical attack; inspection and maintenance procedures; and quality assurance procedures) and condition of the surfacing of all operational areas; 										
	 have an inspection and maintenance programme of impervious surfaces and containment kerbs; 									ainment	
	 justify where operational areas have not been equipped with: 										
	an impervious surface;										
	 spill containment kerbs; sealed construction joints; connection to a sealed drainage system. 4. Bunds All tanks containing liquids whose spillage could be harmful to the environment should be b For further information on bund sizing and design, see Ref. 13. Bunds should:										
										l be bunded.	
	 be impermeable and resistant to the stored materials; 										
	 have no outlet (i.e. no drains or taps) and drain to a blind collection point; have pipework routed within bunded areas with no penetration of contained surfaces; 									es [.]	
	 be designed to catch leaks from tanks or fittings; 									. .	
	 have a capacity which is the greater of 110% of the largest tank or 25% of the total tankage; be subject to regular visual inspection and any contents pumped out or otherwise removed under manual control after checking for contamination; 										
	 where not frequently inspected, be fitted with a high-level probe and an alarm as appropriate; 										
	 have fill points within the bund where possible or otherwise provide adequate containmer have a routine programmed inspection of bunds (normally visual but extending to water testing where structural integrity is in doubt). 										


			CHNIQUES		EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement		Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.4 Groundwater **Emissions to groundwater**

Groundwater protection . legislation

The Groundwater Regulations (1998) came into force on 1 April 1999 (see Appendix 2 for equivalent legislation in Scotland and Northern Ireland). An IPPC Permit will be subject to the following requirements under these Regulations.

- i. It shall not be granted at all if it would permit the direct discharge of a List I substance (Regulation 4(1)) (except in limited circumstances - see note 1 below).
- If the Permit allows the disposal of a List I substance or any other activity which might lead to an ii. indirect discharge (see note 2 below) of a List I substance then prior investigation (as defined in Regulation 7) is required and the Permit shall not be granted if this reveals that indirect discharges of List I substances would occur and in any event conditions to secure prevention of such discharges must be imposed (Regulation 4(2) and (3)).
- In the case of List II substances. Permits allowing direct discharges or possible indirect iii. discharges cannot be granted unless there has been a prior investigation and conditions must be imposed to prevent groundwater pollution (Regulation 5).
- The Regulations contain further detailed provisions covering surveillance of groundwater iv (Regulation 8); conditions required when direct discharges are permitted (Regulation 9); when indirect discharges are permitted (Regulation 10); and review periods and compliance (Regulation 11).

The principles, powers and responsibilities for groundwater protection in England and Wales, together with the Agency's policies in this regard, are outlined in the Environment Agency's document Policy and Practice for the Protection of Groundwater (PPPG) (see Ref. 26). This outlines the concepts of vulnerability and risk and the likely acceptability from the Agency's viewpoint of certain activities within groundwater protection zones.

- Prior investigation of the potential effect on groundwater of on-site disposal activities or Α discharges to groundwater. Such investigations will vary from case to case, but the Regulator is likely to require a map of the proposed disposal area; a description of the underlying geology. hydrogeology and soil type, including the depth of saturated zone and guality of groundwater; the proximity of the site to any surface waters and abstraction points, and the relationship between ground and surface waters; the composition and volume of waste to be disposed of; and the rate of planned disposal.
- Surveillance this will also vary from case to case, but will include monitoring of groundwater в quality and ensuring the necessary precautions to prevent groundwater pollution are being undertaken.
- The Regulations state that, subject to certain conditions, the discharges of List I substances to Note 1 groundwater may be authorised if the groundwater is "permanently unsuitable for other uses". Advice must be sought from the Regulator where this is being considered as a justification for such discharges.
- Note 2 List I and List II refer to the list in the Groundwater Regulations and should not be confused with the similar lists in the Dangerous Substances Directive.

Identify if there may be a discharge of any List I or List II Application Form substances and if any are identified, explain how the Question 2.4 requirements of the Groundwater Regulations 1998 have been addressed.

With the application the Operator should:

Meeting the requirements of the Groundwater Regulations

- 1. Confirm that there are no direct or indirect emissions to groundwater of List I or List II substances from the installation.
- 2 Where there are such releases, provide the information and surveillance arrangements described in A and B above.

Under these Regulations the permit may not be granted if the situation is not satisfactory. Therefore, with the application, the Operator should supply information on List I and List II substances and, if necessary, prior investigation and surveillance information.

INTRODUC		N TE	CHNIQL	JES	Eľ	VISSIO	NS		MPAC	Т
Management	terials	Activities &	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation
	puts	abatement	water		0,7			5		issues
Groundwater	List I									
		Subject to	subparagraph	ı (2) belov	w. a subs	tance is in I	ist I if it I	pelonas to a	one of the	following
	(1)		groups of sub		n, a case			oolongo to o		lonoming
List I and List II substances			nohalogen cor		and subs	tances whic	h may fo	orm such co	mpounds	in the
II Substances		•	tic environme 10phosphorus		nds:					
			notin compour	-	100,					
		(d) subst	tances which	possess o						
			tic environme wise be in Lis		ing substa	ances which	have the	ose properti	es which	would
			ury and its co	-						
			nium and its co	-						
			ral oils and hy	drocarboi	ns;					
	~	(h) cyani		ат: б :сь.	a hace d	to make a d l				
	2.		ce is not in Lis asis of a low ris						e inapprop	oriate to List
	List II	[
	1(1)		ce is in List II ng families or				t on grou	ndwater and	d it belon	gs to one of
		(a) the fo	ollowing metal	loids and	metals a	nd their com	pounds:			
		zi	nc	tin		C	opper			
		ba	arium	nic	kel	b	eryllium			
		cl	nromium	bor	on		ead			
			ranium		enium		anadium	1		
			rsenic	cob			intimony			
			allium		lybdenun	n te	ellurium			
			anium	silv	-					
			des and their of			-				
		comp	tances which l bounds liable t for human cor	o cause t	he forma					
			or persistent of							
			ation of such o apidly convert					hich are bio	logically f	armless or
		.,	anic compoun	ids of pho	sphorus	and elemen	tal phosp	ohorus;		
		(f) fluori								
		(0)	onia and nitrit							
	(2)		ce is also in Li							
			ongs to one of		-	-		-		
		(b) it has and	been determ	ined by tr	ne Regula	ator to be ina	appropria	ite to list I t	inder para	agraph 1(2);
		(c) it has	been determ ty, persistence				propriate	e to List II ha	aving rega	ard to
	3(1)	The Secre	tary of State n der paragraph	nay reviev	w any de		Regulat	or in relatior	n to the ex	ercise of its
	3(2)	The Secre	tary of State s aph (1) above	hall notify	the Reg					
	4	The Regul under this	ator shall from Schedule in s nary available	n time to t uch manr	ime publi her as it c	sh a summa considers ap	ary of the	effect of its	determin	ations

INTROD	UCTION TEC	HNIQUES	EM	IISSION	IS	I	MPAC	Т
Management	MaterialsActivities &inputsabatement	Ground water Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Waste	2.5 Waste h	andling						
	The normal nature ar confirmed in detail in							
	 furnace slags and 	d droceoe						
	 Iumace slays and 	u ulosses						

- maintenance wastes, especially furnace wastes such as insulation, refractory and linings
- de-watered sludges from water treatment plant
- transport and packaging waste including drums.

Application Form Question 2.5 Characterise and quantify each waste stream and describe the proposed measures for waste management storage and handling.

BAT for waste handling

BREF Sections 2.4, 2.10 and 2.17

- With the application the Operator should:
- 1. Identify and quantify the waste streams including which wastes are covered by the Special Waste Regulations 1996 (SI 1996 No 9720).
- 2. Identify the current or proposed handling arrangements.
- 3. Describe the current or proposed position with regard to the techniques below or any others that are pertinent to the installation.
- 4. Demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the *Guide for Applicants*) or alternative measures.

Indicative BAT requirements

- 1. A system should be maintained to record the quantity, nature, origin and, where relevant, the destination, frequency of collection, mode of transport and treatment method of any waste which is disposed of or recovered.
- 2. Wherever practicable, waste should be segregated and the disposal route identified which should be as close to the point of production as possible.
- 3. Records should be maintained of any waste that is sent off-site (Duty of Care).
- 4. Storage areas should be located away from watercourses and sensitive boundaries, e.g. adjacent to areas of public use, and protected against vandalism.
- 5. Storage areas should be clearly marked and signed and containers should be clearly labelled.
- 6. The maximum storage capacity of storage areas should be stated and not exceeded. The maximum storage period for containers should be specified.
- Appropriate storage facilities should be provided for special requirements such as for substances that are flammable, sensitive to heat or light, etc., and incompatible waste types should be kept separate.
- 8. Containers should be stored with lids, caps and valves secured and in place. This also applies to emptied containers.
- 9. Storage containers, drums, etc., should be regularly inspected.
- 10. Procedures should be in place to deal with damaged or leaking containers.
- 11. Materials such as drosses, which may dissolve or react with water, shall only be stored under cover.
- 12. All appropriate steps to prevent emissions (e.g. liquids, dust, VOCs and odour) from storage or handling should be taken (see Sections 2.3.12, 2.3.13 and 2.3.14).

INTROD	UCTIOI	N TEO	CHNIQ	UES	EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement		Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Waste	2.6	Waste	recove	ery or	dispo	sal				
The Regulations require the Regulator, in setting Permit conditions, to take account of certain general										

principles, including that the installation should be operated in such a way that "waste production is avoided in accordance with Council Directive 75/442/EEC on waste; and where waste is produced it is recovered, or where this is technically or economically impossible it is disposed of, while avoiding or reducing the impact on the environment". The objectives of the National Waste Strategies should also be considered.

In order to meet this requirement the Regulator needs Operators to provide the information below.

Application Form Question 2.6	Describe how each waste stream is proposed to be recovered or disposed of. If you propose any disposal, explain why recovery is technically and economically impossible and describe the
	measures planned to avoid or reduce any impact on the environment.

With the application the Operator should:

- 1. Describe, in respect of each waste stream produced by the installation, whether the waste in question is to be recovered or disposed of, and if a disposal option is planned, to justify why recovery is "technically and economically impossible" together with "the measures planned to avoid or reduce any impact on the environment".
 - 2. Include in the description, the Operator's view as to whether waste disposal is likely to be restricted by the implementation of the Landfill Directive.
 - 3. Describe the current or proposed position with regard to the techniques below or any others that are pertinent to the installation.
 - Demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the *Guide for Applicants*) or alternative measures.

Indicative BAT requirements

 Unless agreed with the Regulator to be inappropriate, the Operator should provide a detailed assessment identifying the best practicable environmental options for waste disposal. For existing activities, this may be carried out as an improvement condition to a timescale to be approved by the Regulator.

BAT for waste

BREF Section

disposal or

recovery

2.10

	INTRODUCTION TEC					EMISSIONS			IMPACT		
Management	Materials inputs	Activit abate	ties & ment	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Source of the residues	Associated metals	Residue	Possible options for dealing with them
Raw materials handling, etc.	All metals	Dust, sweepings	Feed for the main process
Smelting furnace	All metals	Slag	Construction material after slag treatment. Abrasive industry. Parts of slag may be used as
			refractory material, e.g. slag from the production of chromium metal
	Ferro alloys	Rich slag	Raw material for other ferro alloy processes
Converting furnace	Cu	Slag	Recycle to smelter
Refining furnaces	Cu	Slag	Recycle to smelter
	Pb Precious metals (PMs)	Skimmings Skimmings and slag	Recovery of other valuable metals Internal recycle
Slag treatment	Cu and Ni	Cleaned slag	Construction material. Matte produced
Melting furnace	All metals	Skimmings	Return to process after treatment
		Slag and salt slag	Metal recovery, recovery of salt and other material
Electro-refining	Cu	Electrolyte bleed	Recovery of Ni
		Anode remnants Anode slime	Return to converter Recovery of precious metals
Electro-winning	Zn, Ni, Co, PMs	Spent electrolyte	Re-use in leaching process
Fused salt	AI	Spent pot lining	Carburant or disposal
electrolysis		Excess bath Anode stubs	Sale as electrolyte Recovery
	Na and Li	Cell material	Scrap iron after cleaning
Distillation	Hg	Residues (hollines)	Re-use as process feed
	Zn, Cd	Residues	Return to process
Leaching	Zn Cu	Ferrite residues Residues	Safe disposal, re-use of liquor Safe disposal
	Ni/Co	Cu/Fe residues	Recovery, disposal
Sulphuric acid	Primary Cu, Pb,		Regeneration
plant	Zn	Acid sludges Weak acid	Safe disposal Leaching, disposal
Furnace linings	All metals	Refractory	Crush, for possible recovery of
			metallic inclusions. May require treatment to minimise potential
			hazards before final disposal
Milling, grinding	Carbon	Carbon and graphite dusts	Use as raw material in other processes
Pickling	Cu, Ti	Spent acid	Recovery
Dry abatement systems	Most – using fabric filters or EPs	Filter dust	Return to process, recovery of other metals, disposal
Wet abatement systems	Processes using scrubbers or wet EPs	Filter sludge	Return to process, recovery of other metals, disposal
Waste water treatment sludge	Most	Hydroxide or sulphide sludges	Safe disposal, re-use

Energy

2.7 Energy

BAT for energy efficiency under the PPC Regulations will be satisfied provided the Operator meets the following conditions:

either

- the Operator meets the basic energy requirements in Sections 2.7.1 and 2.7.2 below and is a
 participant to a Climate Change Agreement (CCA) or Trading Agreement with the Government
- or
- the Operator meets the basic energy requirements in Sections 2.7.1 and 2.7.2 below and the further sector-specific energy requirements in Section 2.7.3 below.

Note that, even where a Climate Change Agreement or Trading Agreement is in place, it does not preclude the consideration of energy efficiency techniques (including those identified in Section 2.7.3) as part of an integrated assessment of BAT where they impact on other emissions, e.g. where:

- the choice of fuel impacts upon emissions other than carbon, e.g. sulphur in fuel;
- the minimisation of waste by waste-to-energy does not maximise energy efficiency, e.g. by CHP;
- the most energy-intensive abatement leads to the greatest reduction in other emissions.

Further guidance is given in the Energy Efficiency Guidance Note (Ref. 15).

2.7.1 Basic energy requirements (1)

Application Form Question 2.7 (part 1) Provide a breakdown of the energy consumption and generation by source and the associated environmental emissions.

The requirements of this section are basic, low-cost, energy standards which apply whether or not a Climate Change Agreement or Trading Agreement is in place for the installation.

With the application the Operator should:

1. Provide energy consumption information in terms of delivered energy and also, in the case of electricity, converted to primary energy consumption. For the public electricity supply, a conversion factor of 2.6 should be used. Where applicable, the use of factors derived from on-site heat and/or power generation, or from direct (non-grid) suppliers should be used. In the latter cases, the Applicant should provide details of such factors. Where energy is exported from the installation, the Applicant should also provide this information. An example of the format in which this information should be presented is given in Table 2.7.1 below. The Operator should also supplement this information with energy flow diagrams (such as energy balances or "Sankey" diagrams, as appropriate) showing how the energy is used throughout the process.

(Note that the Permit will require this information to be submitted annually.)

Table 2.7.1 - Example breakdown of delivered and primary energy consumption

Energy course	Energy	y consumption	
Energy source	Delivered, MWh	Primary, MWh	% of total
Electricity from public supply			
Electricity from other source*			
Imported steam/hot water*			
Gas		N/A	
Oil		N/A	
Coal		N/A	
Other (Operator to specify)			

- 2. Provide information on Specific Energy consumption. The Operator should define and calculate the specific energy consumption of the activity (or activities) based on primary energy consumption for the products or raw material inputs that most closely match the main purpose or production capacity of the installation. The Operator should provide a comparison of specific energy consumption against any relevant benchmarks available for the sector.
- 3. Provide associated environmental emissions. This is dealt with in the Operator's response to Section 3 of this guidance.

INTRODUC			CHNIQL	JES	EN	<u>/ISSION</u>	IS	IN IN	<u>/PAC</u>	
Management	terials puts	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues
Energy		2 Basic (əquire ı	ments	(2)				
		lication Form stion 2.7 (pa		effici The r	iency . requirem	ents of this s	section	s for improve are basic, lo	ow-cost,	energy
								or not a Clima t is in force f		
	With	the applica	ation the C	Operato	r shoula	:				
								ic, low–cost e ues describec		quirements
	2. F	Provide an e	nergy efficie	ncy plan	that appr	aises differer	nt energy	y options as d	lescribed	l below.
	Basio	c energy re	quiremen	ts						
	1		ecklists of ap	ppropriate	e measur			e in place in th opendix 2 of th		
	· ·	 air conditi 		ess refrig	eration ar	nd cooling sy	rstems (le	eaks, seals, te	emperati	ure control,
	•	•	of motors a							
	•	-		-	-	dures for use	-			
	•		-	-		, insulation);				
		-	ating and ho		-					
			n to avoid hi	-			air [.]			
						sing excess a ties within the		tion		
										These
	9 6	should includ	le insulation unnecessar	, containr ry dischar	ment meth rge of hea	hods (such as	s seals a	gross inefficie and self-closir example, by fi	ng doors)), and
	i s	Services sec ssues may b	tion of the <i>E</i> be of minor in heless find a	Energy Eff mpact and a place in	ficiency G d should in the prog	<i>uidance Note</i> not distract e	e. For er	the requirement nergy-intension the major en where they con	ve indust nergy iss	tries these sues. They
								o the requirer targeting of a		
		An energy ef identifies 2.7.3, tha		es releva	int to the i	nstallation, in	ncluding	those listed a	above and	d in Section
		 estimates 	s the CO2 sa	avings that	at would b	e achieved b	-	measure over		
	ć	 provides costs pe 	information	on the e	equivalent d and the	annual costs	s of imple	CA or Tradii ementation of tation. A proc	f the tech	inique, the
	ļ	An example f	format of the	energy (efficiency	plan is show	n overle	af.		

INTRODUCTION TECHNIQUES					EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Table 2.7.2 Example format for energy efficiency plan

ALL A	PPLICANTS			PLICANTS WITHOU	TCCA
Energy efficiency measure	CO ₂ savings (tonnes)		Equivalent Annual Cost (EAC) ¹ £k	EAC/CO ₂ saved £/tonne	Date for implementation

Notes

 Refer to Energy Efficiency Guidance Note for appraisal methodology. Where other appraisal methodologies have been used, state the method, and provide evidence that appropriate discount rates, asset life and expenditure (£/t) criteria have been employed

The energy efficiency plan is required to ensure that the Operator has considered all relevant techniques. However, where a CCA is in place the Regulator will only enforce implementation of those measures in categories 1-3 above.

2.7.3 Sector-specific energy requirements

Application Form Question 2.7 (part 3) Describe the proposed measures for improvement of energy efficiency (only where the installation is <u>not</u> the subject of a Climate Change Agreement or Trading Agreement)

Where there is no Climate Change Agreement or Trading Agreement in place, the Operator should demonstrate the degree to which the further energy efficiency measures identified in the implementation plan, including those below, have been taken into consideration for this sector and justify where they have not.

With the application the Operator should:

- 1. Identify which of the measures below are applicable to the activities, and include them in the appraisal for the energy efficiency plan in Section 2.7.2.
- 2. Describe the current or proposed position with regard to the techniques below, or any others which are pertinent to the installation.
- 3. Demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the *Guide for Applicants*) or alternative measures.

Indicative BAT requirements

The following techniques should be implemented where they meet the financial criteria in Appendix 4 of the Energy Efficiency Guidance Note (Ref. 15).

Energy efficiency techniques

Within IPPC it is valid to consider both the emission of direct (from on-site fuel use) and indirect (from a remote power station) pollution when considering options for energy and energy efficiency.

Energy recovery before or after abatement is applicable in the majority of cases but local circumstances are important, for example, where there is no outlet for the recovered energy. The BREF Sections 2 in Chapters 3 to 12 give the current range of energy use for the various processes. BAT conclusions for energy use are not given as the raw materials available have a very strong influence on the energy use. BAT conclusions for the principles of energy recovery are given in Chapter 2. They are:

- Production of steam and electricity from the heat raised in waste heat boilers.
- The use of the heat of reaction to smelt or roast concentrates or melt scrap metals in a converter.

Cont.

BAT for energy (cont.)

BREF Section 2.11

1

abatement

inputs

EMISSIONS

```
IMPACT
Installation
```

issues

Energy Accidents Noise Monitoring Closure

- The use of hot process gases to dry feed materials.
- Pre-heating of a furnace charge using the energy content of furnace gases or hot gases from another source.
- The use of recuperative burners or the pre-heating of combustion air.
- The use as a fuel gas of CO produced.

water

- The heating of leach liquors from hot process gases or liquors.
- The use of plastic contents in some raw materials as a fuel, provided that good quality plastic cannot be recovered and VOCs and dioxins are not emitted.
- The use of low-mass refractories where practicable.
- The use of oxygen is recognised to have advantages in many cases and reduces the overall gas volume, allows autogenic operation and can allow smaller abatement plant.
- Process optimisation to minimise hot metal transfers.

Further information will be found in both the existing sector guidance and in the Energy Efficiency Guidance Note (Ref. 15).

2. Energy supply techniques

- Use of CHP.
- Use of less polluting fuels.

Irrespective of whether a Climate Change Agreement or Trading Agreement is in place, where there are other BAT considerations involved, such as:

- the choice of fuel impacts upon emissions other than carbon, e.g. sulphur in fuel;
- where the potential minimisation of waste emissions by recovery of energy from waste conflicts with energy efficiency requirements,

the Operator should provide justification that the proposed or current situation represents BAT.

Where there is an on-site combustion plant other guidance is also relevant. For plants greater than 50 MW, Operators should consult the IPPC Guidance on power generation (reference IPC S2 1.01 and supplement IPC S3 1.01) and the operators of plant of 20–50 MW should consult the Local Authority Air Pollution Control Guidance. On IPPC installations this Guidance will be generally applicable to plant under 20 MW also. For incineration plant IPC S2 5.01 Waste Incineration should be consulted.

INTRODUCTION							IMPACT		
Management Materials Acti inputs aba	vities & G tement	Bround water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

IPPC requires as a general principle that necessary measures should be taken to prevent accidents which may have environmental consequences, and to limit those consequences. This section covers general areas of any installation operations which have the potential for accidental emission.

Some installations will also be subject to the Control of Major Accident Hazards Regulations 1999 (COMAH) (see Appendix 2 for equivalent legislation in Scotland and Northern Ireland). There is an element of overlap between IPPC and COMAH and it is recognised that some systems and information for both regimes may be interchangeable.

The COMAH regime applies to major hazards. For accident aspects covered by COMAH, reference should be made to any reports already held by the Regulator. However, the accident provisions under IPPC may fall beneath the threshold for major accident classification under COMAH and therefore consideration should be given to smaller accidents and incidents as well. Guidance (see Ref. 20) prepared in support of the COMAH Regulations may also be of help to IPPC Operators (whether or not they are covered by the COMAH regime) in considering ways to reduce the risks and consequences of accident.

General management requirements are covered in Section 2.1. For accident management, there are three particular components:

- identification of the hazards posed by the installation/activity;
- assessment of the risks (hazard x probability) of accidents and their possible consequences;
- implementation of **measures to reduce the risks** of accidents, and contingency plans for any accidents that occur.

Application Form Question 2.8 Describe your documented system proposed to be used to identify, assess and minimise the environmental risks and hazards of accidents and their consequences.

With the application the Operator should:

- 1. Provide the accident management plan described in the indicative BAT requirements below describing the current or proposed position with regard to the techniques listed below or any others which are pertinent to the installation.
- 2. Demonstrate that the proposals are BAT, by confirming compliance with the indicative requirements, by justifying departures (as described in Section 1.2 and in the Guide for Applicants) or alternative measures.
- 3. Identify any issues which may be critical.

Indicative BAT requirements

- 1. A structured accident management plan should be submitted to the Regulator which should:
 - (a) Identify the hazards to the environment posed by the installation. Particular areas to consider may include, but should not be limited to, the following:
 - transfer of substances (e.g. loading or unloading from or to vessels)
 - overfilling of vessels
 - failure of plant or equipment (including failure of electrical power supplies and failure of process control equipment)
 - failure of containment (e.g. bund and/or overfilling of drainage sumps)
 - failure to contain fire waters
 - making the wrong connections in drains or other systems
 - preventing incompatible substances coming into contact
 - unwanted reactions and/or runaway reactions
 - emission of an effluent before adequate checking of its composition has taken place
 - steam main issues
 - vandalism
 - charging unsuitable materials into furnaces, e.g. sealed containers, munitions.

Cont.

BAT for control of accidents

BREF Section 2.15

INTRODUC			CHNIC	UES	EMISS	ONS	IMPACT	
Mananement		Activities & abatement		Waste	Energy Accide	nts Noise	Monitoring Closure	nstallatior issues
	_							
Accidents	(4					ards, the proc	cess of assessing the ris	ks can be
				-	basic questions: probability of their	occurrence?	(Source frequency)	
					w much? (Risk eva			
		3. W	here does		-		hat are the pathways ar	nd
BAT for control			ceptors?)	concoqu	oncos? (Consoqui	2222222222	nent – the effects on the	rocontor
of accidents 'cont.)							all risk and its significant	-
com.)			vironment)					
					educe the risk? (Ri ce their environme		ent – measures to preve nces)	nt
		its loc	ation. The	main fac	tors which should b	oe taken into a		
		ac	tivities;			·	d by the installation and t	the
					opulation and the		• •	1
							ise of the activities and t risk control techniques.	ne relativ
	(*	c) Identi	fy the tech	nniques i	necessary to redu	ce the risks i	including:	
		r e	An inventor could have nany appa escape (e.ç	y should environn rently inn g. a tanke	nental consequence ocuous substances r of milk spilled into	ubstances, pr es if they esca s can be envir o a watercours	installations. esent or likely to be pres ape. It should not be for onmentally damaging if se could destroy its ecos f any changes to the inve	gotten tha they system).
		•	procedu	ures shou ibility with	ld be in place for c	hecking raw n	naterials and wastes to e ey may accidentally com	ensure
		•		te storag	e arrangements for	raw materials	s, products and wastes s	should be
		•	to ensub be giver systems reading	re that co n to proce s based c	ess design alarms, on microprocessor o s ultrasonic gauges	trips and othe control and pa	v situations, consideratio er control aspects, e.g. a assing valve control, tank arnings and process inte	utomatic (level
		•			iques, such as suit ent of vehicles, sho		o prevent damage to eq ed as appropriate;	uipment
		•	appropr contain		ainment should be	provided, e.g	bunds and catchpots, b	ouilding
		•	tanks (li	quid or p		neasurement,	ed to prevent overfilling independent high-level	
		•					sed access should be pr ingements where neces	
		•	change	s to proce	edures, abnormal e	vents and fine	all incidents, near-misse dings of maintenance ins	spections
		•	procedu incident		ld be established to	o identify, res	oond to and learn from s	uch
		•	the role be iden		ponsibilities of pers	sonnel involve	ed in accident managem	ent shoul
	•							

INTRODUC			UES	E	MISSION	S	II	MPAC	Т						
Manadement	terials Activities a puts abatemen	Ground	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues						
Accidents		 clear gui 	dance sh	nould be a	available on ho	w each a	accident scer	nario shoi	uld be						
		manageo	d, e.g. co	ntainmer	t or dispersion	, to extin	guish fires o	r let them	burn;						
		 managed, e.g. containment or dispersion, to extinguish fires or let them burn; procedures should be in place to avoid incidents occurring as a result of poor communication among operations staff during shift changes and maintenance or other engineering work; safe shut-down procedures should be in place; communication routes should be established with relevant authorities and emergency services both before and in the event of an accident. Post-accident procedures should include the assessment of harm caused and steps needed to redress this; appropriate control techniques should be in place to limit the consequences of an accident, such as oil spillage equipment, isolation of drains, alerting of relevant authorities and evacuation procedures; personnel training requirements should be identified and provided; the systems for the prevention of fugitive emissions are generally relevant (Sections 2.3.12 and 2.3.13) and, in addition, for drainage systems: procedures should be in place to ensure that the composition of the contents o a bund sump, or sump connected to a drainage system, are checked before treatment or disposal; drainage sumps should be equipped with a high-level alarm or sensor with automatic pump to storage (not to discharge); there should be a system in place 													
BAT for control of accidents (cont.)		emergen procedur	cy servic es shoul	es both l	pefore and in th	ne event	of an accide	nt. Post-	accident						
		accident	such as	oil spilla	ge equipment,										
		-					-								
									ant						
		 procedures should be in place to ensure that the composition of the contents a bund sump, or sump connected to a drainage system, are checked before treatment or disposal; 													
		autor	natic pur	np to sto		charge);	there should	l be a sys							
			level alaı control.	rms, etc.,	should not be	routinely	used as the	primary	method of						
	<i>(ii)</i>	The following the risks as i			pecific techniqı 2 above.	ues ident	tified as nece	essary to	minimise						
							provided wi	th mainte	nance and						
	 adequate redundancy or stand-by plant should be provided with maintenance testing to the same standards as the main plant; process waters, site drainage waters, emergency fire water, chemically contaminated waters and spillages of chemicals should, where appropriate, b contained and, where necessary, routed to the effluent system, with provision contain surges and storm-water flows, and treated before emission to controll waters or sewer. Sufficient storage should be provided to ensure that this con achieved. There should also be spill contingency procedures to minimise the of accidental emission of raw materials, products and waste materials and to prevent their entry into water. Any emergency fire water collection system sh also take account of the additional fire water flows or fire fighting foams. Emergency storage lagoons may be needed to prevent contaminated fire water reaching controlled waters (see Refs. 16 and 17); 														
		accidenta	al emissi nadvisab	ons from	ven to the poss vents and safe ety grounds, at ;	ty relief	valves/bursti	ng discs.	Where this						
		 other tec 	hniques	as identi	ied in existing	sector gu	lidance.								

INTRODUCTION TE		N TEO			EMISSIONS			IMPACT		
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Noise and vibration

2.9 Noise and vibration

Within this section "noise" should be taken to refer to "noise and/or vibration" as appropriate, detectable beyond the site boundary.

The PPC Regulations require installations to be operated in such a way that "all the appropriate preventative measures are taken against pollution, in particular through the application of BAT". The definition of pollution includes "emissions which may be harmful to human health or the quality of the environment, cause offence to human senses or impair or interfere with amenities and other legitimate uses of the environment". BAT is therefore likely to be similar, in practice, to the requirements of the statutory nuisance legislation, which requires the use of "best practicable means" to prevent or minimise noise nuisance.

In the case of noise, "offence to any human senses" can normally be judged by the likelihood of complaints, but in some cases it may be possible to reduce noise emissions still further at reasonable costs, and this may exceptionally therefore be BAT for noise emissions.

For advice on how noise and/or vibration related limits and conditions will be determined see *IPPC Noise – Part 1 Regulation and Permitting* (see Ref. 21).

Application Form Question 2.9

Identify the main sources of noise and vibration (including infrequent sources); the nearest noise-sensitive locations and relevant environmental surveys which have been undertaken; and the proposed techniques and measures for the control of noise.

With the application the Operator should:

- Provide the following information for **each main source of noise and vibration** that falls within the IPPC installation:
 - the source and its location on a scaled plan of the site
 - whether continuous/intermittent, fixed or mobile
 - the hours of operation
 - its description (e.g. clatter, whine, hiss, screech, hum, bangs, clicks, thumps or tonal elements)
 - its contribution to overall site noise emission (categorise each as high, medium or low unless supporting data are available).

A common-sense approach needs to be adopted in determining which sources to include. The ones which need to be considered are those which may have environmental nuisance impact; e.g. a small unit could cause an occupational noise issue in an enclosed space but would be unlikely to cause an environmental issue. Conversely a large unit or a number of smaller units enclosed within a building could, for example, cause a nuisance if doors are left open. It must also be remembered that noise, which is not particularly noticeable during the day, may become more noticeable at night.

2. Provide the information required in (1) for each *infrequent source of noise and vibration* not listed above, that falls within the IPPC installation plus its times of operation (this includes infrequently operated/seasonal operations, cleaning/maintenance activities, on-site deliveries/collections/transport or out-of-hours activities, emergency generators or pumps and alarm testing).

Identify *the nearest noise-sensitive sites* (typically dwellings, parkland and open spaces – schools, hospitals and commercial premises <u>may</u> be, depending upon the activities undertaken there) and any other points/boundary where conditions have been applied by Local Authority officers or as part of a planning consent, relating to:

- (a) the local environment:
 - provide an accurate map or scaled plan showing grid reference, nature of the receiving site, distance and direction from site boundary;
- (b) conditions/limits imposed which relate to other locations (i.e. boundary fence or surrogate for nearest sensitive receptor):
 - any planning conditions imposed by the Local Authority (day/evening/night);
 - other conditions imposed by agreements, e.g. limits on operating times, technologies, etc.;
 - any requirements of any legal notices, etc.

Information needed to determine BAT for noise and vibration

INTRODUC			CHNIQ	UES	F١	/ISSIO	IS		MPAC	т			
Management Mat	terials	Activities &	Ground	Waste	Energy	Accidents	Noise	Monitoring		Installation			
in	puts	abatement	water	Waste	Energy	71001001113	Noise	Monitoring	Cloburc	issues			
Noise and vibration		 bac spe aml vibr 	cific noise pient noise ation data	oise level level (da level (da which ma	y/evening //evening/ ay be expr	(day/night/e /night) L _{Aeq,7} /night) L _{Aeq,7} essed in terr DV) in m s ^{-1.}	; and/or , as appro	opriate;	e velocity	(ppv) in mm			
Information needed to determine BAT for noise and vibration (cont.)		 For noise these are given the meaning as defined in BS4142:1997 "Method for rating industrial noise affecting mixed residential and industrial areas", and to which reference should be made fo a full description. For vibration, the appropriate standard is BS6472:1992 "Evaluation of human exposure to vibration in buildings 1 to 80 Hz". In very general terms "background" is taken to be the equivalent continuous A-weighted noise remaining when the source under investigation is not operational averaged over a representative time period, <i>T</i>. The "ambient" level is the equivalent continuous A-weighted combination of all noise sources far and distant, including the source under investigation, and "specific noise" is the equivalent continuous A-weighted noise level produced by the source under investigation as measured at a selected assessment point. Both are averaged over a time period, <i>T</i>. BS4142 gives advice on the appropriate reference periods. "Worst-case" situations and impulsive or tonal noise should be accounted for separately and not "averaged out" over the measurement period. Provide <i>details of any environmental noise measurement surveys</i>, modelling or any other noise measurements undertaken relevant to the environmental impact of the site, identifying: the purpose/context of the survey the locations where measurements were taken the source(s) investigated or identified the outcomes. 											
	4.												
	5.	Identify any	specific loc	al issues	and propo	osals for imp	rovement	s.					
	6.	Describe the any others w					to the tech	nniques belo	w, any in	Ref. 21 or			
	7.	Demonstrate requirements Applicants) o	s, by justify	ing depa	rtures (as								
	Inc	dicative BA	T require	ments									
	 The Operator should employ basic good practice measures for the control of noise, in adequate maintenance of any parts of plant or equipment whose deterioration may gi increases in noise (e.g. maintenance of bearings, air handling plant, the building fabri specific noise attenuation measures associated with plant, equipment or machinery). 												
	2.	In addition the noise from the the Regulated installation end levels of 50 of exceeded in	ne installat or and, in p xceed the dB <i>L</i> _{Aeq} by	ion does articular, numerica day or 48	not give ris should jus Il value of 5 by night a	se to reason tify where ei the backgrou are exceede	able caus ither rating und sound d. Reasor	e for annoya g levels (<i>L</i> _{Aec} d level (<i>L</i> _{A90,7}	ance, in th $_{q,T}$) from the $_{T}$), or the a	e view of he absolute			
	3.	In some circ been identifi the Operator minimise pro	ed in pre-a should er	pplication	n discussio h noise co	ns or in pre-	vious disc jues as ar	ussions with re considere	the local	authority,			
	4.	Noise survey power levels installations managemen given in Part	for individ depending t plan as p	ual items upon the art of the	of plant) of plant) of plant	or modelling for noise pro	may be noblems. C	ecessary for Operators ma	either ne ay have a	w or existing noise			

IPPC

INTRODUCTION TE							IMPACT		
Management	Materials Activity inputs abar	vities & tement	Ground water	Waste	Energy	Accidents	Noise	Monitoring Cl	losure Installatio

Monitoring

2.10 Monitoring

This section describes monitoring and reporting requirements for emissions to all environmental media. Guidance is provided for the selection of the appropriate monitoring methodologies, frequency of monitoring, compliance assessment criteria and environmental monitoring.

Application Form Question 2.10

employed.

With the application the Operator should:

Describe the proposed measures for monitoring emissions, including any environmental monitoring, and the frequency, measurement methodology and evaluation procedure proposed.

BREF Section 2.2

- Describe the current or proposed position with regard to the monitoring requirements below or any others which are pertinent to the installation for "Emissions monitoring", "Environmental monitoring", "Process monitoring" (where environmentally relevant) and "Monitoring standards"
- 2. Provide, in particular, the information described in requirement 17 below.
- 3. Provide justifications for not using any of the monitoring requirements described.
- 4. Identify shortfalls in the above information which the Operator believes require longer-term studies to establish.

Emissions monitoring

The following monitoring parameters and frequency are normally appropriate in this sector. Generally, monitoring should be undertaken during commissioning, start-up, normal operation and shut-down unless the Regulator agrees that it would be inappropriate to do so.

Where effective surrogates are available they may be used to minimise monitoring costs.

Where monitoring shows that substances are not emitted in significant quantities, consideration can be given to a reduced monitoring frequency.

Monitoring and reporting of emissions to water and sewer

1. Monitoring of process effluents released to controlled waters should include at least:

Parameter	Monitoring frequency						
Flowrate	Continuous and integrated daily flowrate						
рН	Continuous						
Temperature	Continuous						
COD/BOD	Flow weighted sample or composite samples, weekly analysis, reported as flow weighted monthly averages						
Turbidity	Continuous						
Metals which are likely to be released from the activity	Flow weighted sample or composite samples, weekly analysis, reported as flow weighted monthly averages						

2. Monitoring of process effluents released to sewer should include at least:

Parameter	Monitoring frequency
Flowrate	Continuous (using a flow proportional sampler to create a 24 h composite sample which is analysed daily against trade effluent consent) and integrated daily flowrate
рН	Continuous
Temperature	Dependent on process; if process may generate an effluent > 25°C, continuous monitoring would be appropriate
COD/BOD	Flow weighted sample or composite samples, weekly analysis, reported as flow weighted monthly averages
TOC	Dependent on process, see "Monitoring of Process Variables"

INTROD	UCTIO	ON TE	CHNIQ	UES	E	MISSIO	NS	I	MPAC	Т					
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues					
						_									
Monitoring	the		ncies will v	ary depe	nding upo			nonitored. The e receiving wa							
	3.	substances the release l	to establisl imits. This greed with	n that all s should o	relevant : cover the	substances h substances	nave bee listed in 3	out covering a n taken into ac Schedule 5 of t ıble. This shou	count whe	en setting ations					
	4.														
	5.	whose capa "Whole efflu measureme available (R	city for har ent toxicity nts of harr ef. 22) and	rm is unc /" monito n, e.g. dii I the Reg	ertain, pa ring techi rect toxic julator wi	nticularly when niques can th ty assessme	en in cor herefore l ent. Som g further	are more difficu nbination with o be appropriate e guidance on guidance in du nce.	other sub to provide toxicity te	stances. e direct esting is					
	Мо	Monitoring and reporting of emissions to air													
		There is a wide variety of possible releases to air, and specific information may be found in Sections 2.3.1 to 2.3.9.													
	6.	 6. Continuous monitoring and recording are likely to be required under the following circur Where the potential environmental impact is significant or the concentration of the s varies widely. 													
		performa	ince of the	abateme	ent plant.	For example	e, continu	uous monitoring	required to show the ng of dust is needed an maintenance is						
			other contr election, d			equired to ac	chieve sa	tisfactory levels	s of emis	sion (e.g.					
	7.	Gas flow sho	uld be mea	asured, o	r otherwi	se determine	ed, to rela	ate concentratio	ons to ma	iss releases.					
	8.	temperature	and pressเ ceed 3%, เ	ure of the	emissior	n. The water	vapour	ssary to measu content must al for other polluta	so be me	easured if it					
	9.		ll final rele	ases to a	ir should			f releases shou less, free from							
										Cont.					

INTRODUCTION TECHNIQUES				MISSIO		IMPACT				
Management	Materials inputs	Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

Monitoring

Emissions monitoring (cont.) The following table indicates the likely monitoring frequencies for releases to air in this sector:

Cubatanaa	Availability o	of methods for
Substance –	Continuous	Extractive
SO ₂	Y	Y
NO _x	Y	Y
Particulate	Y	Y
VOC	Y	Y
HCI	Y	Y
HF		Y
Metals		Y
Dioxins & furans		Y
CO	Y	Y

Monitoring and reporting of waste emissions

10. For waste emissions the following should be monitored and recorded:

- the physical and chemical composition of the waste;
- its hazard characteristics;
- handling precautions and substances with which it cannot be mixed;
- where waste is disposed of directly to land, for example sludge spreading or an on-site landfill, a programme of monitoring should be established that takes into account the materials, potential contaminants and potential pathways from the land to groundwater, surface water or the food chain.

Environmental monitoring (beyond the installation)

Environmental monitoring

Environmental

monitoring

(cont.)

12. The Operator should consider the need for environmental monitoring to assess the effects of emissions to controlled water, groundwater, air or land or emissions of noise or odour.

Environmental monitoring may be required, e.g. when:

- there are vulnerable receptors;
- the emissions are a significant contributor to an Environmental Quality Standard (EQS) which may be at risk;
- the Operator is looking for departures from standards based on lack of effect on the environment;
- to validate modelling work.
- 13. The need should be considered for:
 - groundwater, where it should be designed to characterise both quality and flow and take into account short- and long-term variations in both. Monitoring will need to take place both up-gradient and down-gradient of the site;
 - surface water, where consideration will be needed for sampling, analysis and reporting for upstream and downstream quality of the controlled water;
 - air, including odour;
 - land contamination, including vegetation, and agricultural products;
 - assessment of health impacts;
 - noise.

INTROD	UCTIO	N TEC	CHNIQ	UES	E	MISSIO	NS	I	MPAC	Т				
Management		Activities & abatement	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues				
lonitoring	14.	Where envi	ronmenta	l monito	ring is ne	eded the fo	llowing	should be co	nsidered	in drawing				
up proposals:														
	 determinands to be monitored, standard reference methods, sampling protocols; 													
		 monitoring strategy, selection of monitoring points, optimisation of monitoring approach 												
		• determin	ation of ba	ckground	levels co	ontributed by	buted by other sources;							
		 uncertainty for the employed methodologies and the resultant overall uncertainty of measurement; 												
		 quality assurance (QA) and quality control (QC) protocols, equipment calibration and maintenance, sample storage and chain of custody/audit trail; 												
			eporting procedures, data storage, interpretation and review of results, reporting format for he provision of information for the Regulator.											
		Guidance or Guidance No												

Monitoring of process variables

Monitoring process variables

BREF Chapter 2.17.2

Equipment

standards

MCERTS

- 15. Some process variables will have potential environmental impact and these should be identified and monitored as appropriate. Examples might be:
 - raw materials monitoring for contaminants where contaminants are likely and there is inadequate supplier information (see Section 2.2.1);
 - oxygen, carbon monoxide, pressure or temperature in the furnace atmosphere or off-gases;
 - plant efficiency where it has an environmental relevance;
 - energy consumption across the plant and at individual points of use in accordance with the energy plan. Frequency – normally continuous and recorded;
 - fresh water use across the activities and at individual points of use should be monitored as part of the water efficiency plan (see Section 2.2.3). Frequency – continuous and recorded;
 - the quantity of each class of waste generated.

Monitoring standards (Standard Reference Methods)

Equipment standards

The Environment Agency has introduced its Monitoring Certification Scheme (MCERTS) to improve the quality of monitoring data and to ensure that the instrumentation and methodologies employed for monitoring are fit for purpose. Performance standards have been published for continuous emissions monitoring systems (CEMs). Other MCERTS standards are under development to cover manual stack emissions monitoring, portable emissions monitoring equipment, ambient air quality monitors, water monitoring instrumentation, data acquisition and Operators' own arrangements, such as for installation, calibration and maintenance of monitoring equipment, position of sampling ports and provision of safe access for manual stack monitoring.

16. As far as possible, Operators should ensure their monitoring arrangements comply with the requirements of MCERTS where available, e.g. using certified instruments and equipment, and using a registered stack testing organisation etc. Where the monitoring arrangements are not in accordance with MCERTS requirements, the Operator should provide justification and describe the monitoring provisions in detail. See Environment Agency Website (Ref. 22) for listing of MCERTS equipment.

INTRODUC		TECHNIC		E	MISSIO	NS	IMPA	-
Mananement	terials Activiti puts abater		Waste	Energy	Accidents	Noise	Monitoring Closu	re Installatior issues
lonitoring tandards cont.)	comply • mo • just • refe • me • crit stra • rep inte • pro • drif • the	with MCERTS nitoring methor tification for con- erence condition asurement unce eria for the ass ategy aimed at porting procedure ervals for the pro- cedures for mo- t correction cal	B requirem ds and pro ntinuous m ns and ave ertainty of essment c demonstra res and da ovision of onitoring du ibration inf neld by sar	ents or fo cedures of nonitoring eraging p of the propo- of non-cor ation of co ata storag information uring star tervals an	r which othe (selection of or spot sam eriods; osed methoc mpliance with ompliance; e of monitori on to the Reg t-up and shu d methods;	r arrange Standard pling; Is and the n Permit li ng results gulator; t-down ar	which monitoring pro- ments have been ma Reference Methods e resultant overall und imits and details of m s, record keeping and abnormal process ils of the people used	ade:); certainty; nonitoring d reporting conditions;
Standards for sampling and analysis BREF: Monitoring REF document in preparation	 18. The anneedin Could Brite Brite Interest of the subject analysis white 	g to be monito mité Européen tish Standards ernational Stan hers - United States - American So - Deutsches Ir - Verein Deuts - Association F ance on stands uidance Note 4 is currently in p ich will also be lance relevant	ds given in red, standa de Norma Institution dardisatior s Environm ciety for To stitut für N cher Inger Française o ards for mo (Monitorio preparation suitable fo	ards shou lisation (((BSI); n Organis nental Pro esting an lormung (nieure (VI de Norma onitoring no, itor g n, This G or calibrat	Id be used ir CEN); ation (ISO); btection Ager d Materials (DIN); DIN; slisation (AFN gaseous rele Ref. 22). A s uidance specton of continu	n the follo ncy (US E ASTM); NOR). eases rele series of u cifies mar uous emis	n the event of other s wing order of priority PA); evant to IPC/IPPC is g updated Guidance No bual methods of sam ssion monitoring instr publications of the Sta	given in the otes covering pling and ruments.

If in doubt the Operator should consult the Regulator.

INTROD	UCTIO	N TE		UES	E	MISSIO	NS		IMPAC	Т
Management	Materials inputs	Main activities	Ground water	Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.11 Decommissioning

The IPPC application requires the preparation of a site report whose purpose, as described in more detail in Refs. 5 and 6 is to provide a point of reference against which later determinations can be made of whether there has been any deterioration of the site and information on the vulnerability of the site.



With the application the Operator should:

1. Supply the site report.

BREF Section 2.16

decommissioning

BAT for

- 2. Describe the current or proposed position with regard to the techniques below or any others that are pertinent to the installation.
- 3. For existing activities, identify shortfalls in the above information that the Operator believes require longer-term studies to establish.

Indicative BAT requirements

1. Operations during the IPPC Permit

Operations during the life of the IPPC Permit should not lead to any deterioration of the site if the requirements of the other sections of this and the specific sector notes are adhered to. Should any instances arise which have, or might have, impacted on the state of the site, the Operator should record them along with any further investigation or ameliorating work carried out. This will ensure that there is a coherent record of the state of the site throughout the period of the IPPC Permit. This is as important for the protection of the Operator as it is for the protection of the environment. Any changes to this record should be submitted to the Regulator.

2. Steps to be taken at the design and build stage of the activities

Care should be taken at the design stage to minimise risks during decommissioning. For existing installations, where potential problems are identified, a programme of improvements should be put in place to a timescale agreed with the Regulator. Designs should ensure that:

- underground tanks and pipework are avoided where possible (unless protected by secondary containment or a suitable monitoring programme);
- there is provision for the draining and clean-out of vessels and pipework prior to dismantling;
- lagoons and landfills are designed with a view to their eventual clean-up or surrender;
- insulation is provided which is readily dismantled without dust or hazard;
- materials used are recyclable (having regard for operational or other environmental objectives).

3. The site closure plan

A site closure plan should be maintained to demonstrate that, in its current state, the installation can be decommissioned to avoid any pollution risk and return the site of operation to a satisfactory state. The plan should be kept updated as material changes occur. Common sense should be used in the level of detail, since the circumstances at closure will affect the final plans. However, even at an early stage, the closure plan should include:

- either the removal or the flushing out of pipelines and vessels where appropriate and their complete emptying of any potentially harmful contents;
- plans of all underground pipes and vessels;
- the method and resource necessary for the clearing of lagoons;
- the method of ensuring that any on-site landfills can meet the equivalent of surrender conditions;
- the removal of asbestos or other potentially harmful materials unless agreed that it is reasonable to leave such liabilities to future owners;

INTRODUC	TION TE	CHNIQUE	S E	MISSIO	NS		IMPAC	T
Management	erials Activities & abatement	Ground water	aste Energy	Accidents	Noise	Monitoring	Closure	Installation issues
BAT for decommissioning (cont.)	the prote testing o	responsibilities ties, the site clo	e and grounds ertain the deg on to return the re not covered under the Ra	water at cons gree of any p e site to a sa d by this legis adioactive Su	struction collution itisfactor slation, t ubstance	and demolit caused by th y state as de out decommi s Act 1993.	tion sites; ne activities a efined by the issioning pla)	and the · initial site ns should be

INTROD	UCTIO	N TE	CHNIQ	UES	E	MISSIO	NS		IMPA	СТ
Management		Activities & abatement		Waste	Energy	Accidents	Noise	Monitoring	Closure	Installation issues

2.12 Installation-wide issues

In some cases it is possible that actions which benefit the environmental performance of the overall installation will increase the emissions from one Permit-holder's activities. For example, taking treated effluent as a raw water supply will probably slightly increase emissions from that activity but could dramatically cut the total emissions from the whole installation.



Where you are not the only operator of the installation, describe the proposed techniques and measures (including those to be taken jointly by yourself and other operators) for ensuring the satisfactory operation of the whole installation.

With the application the Operator should:

1. Where there are a number of separate Permits for the installation (particularly where there are different operators), identify any installation-wide issues and opportunities for further interactions between the Permit-holders whereby the performance of the overall installation may be improved.

In particular:

2. Describe the current or proposed position with regard to the techniques below, or any others which are pertinent to the installation.

Indicative BAT Requirements

The possibilities will be both sector- and site-specific, and include the following:

- 1. Communication procedures between the various Permit-holders; in particular those needed to ensure that the risk of environmental incidents is minimised.
- 2. Benefiting from the economies of scale to justify the installation of a CHP plant.
- 3. The combining of combustible wastes to justify a combined waste-to-energy/CHP plant.
- 4. The waste from one activity being a possible feedstock for another.
- 5. The treated effluent from one activity being of adequate quality to be the raw water feed for another activity.
- 6. The combining of effluent to justify a combined or upgraded effluent treatment plant.
- 7. The avoidance of accidents from one activity which may have a detrimental knock-on effect on the neighbouring activity.
- 8. Land contamination from one activity affecting another or the possibility that one operator owns the land on which the other is situated.

BAT across the whole installation

BREF Section 2.12

EMISSIONS

Benchmark comparison

Benchmark status

3 EMISSION BENCHMARKS

3.1 Emissions inventory and benchmark comparison

Application Form Question 3.1 Describe the nature, quantities and sources of foreseeable emissions into each medium (which will result from the techniques proposed in Section 2).

With the application the Operator should:

- 1. Provide a table of significant emissions of substances (except noise, vibration, odour or heat which are covered in their respective sections) that will result from the proposals in Section 2 and should include, preferably in order of significance:
 - substance (where the substance is a mixture, e.g. VOCs or COD, separate identification of the main constituents or inclusion of an improvement proposal to identify them);
 - source, including height, location and efflux velocity;
 - media to which it is released;
 - any relevant EQS or other obligations;
 - benchmark;
 - proposed emissions normal/max expressed, as appropriate (see Section 3.2), for:
 - mass/unit time
 - concentration
 - annual mass emissions
 - statistical basis (average, percentile, etc.);
 - notes covering the confidence in the ability to meet the benchmark values;
 - if intermittent, the appropriate frequencies;
 - plant loads at which the data are applicable;
 - whether measured or calculated (the method of calculation should be provided).

The response should clearly state whether the emissions are current emission rates or those planned following improvements, and should cover emissions under both normal and abnormal conditions for:

- · point source emissions to surface water, groundwater and sewer;
- waste emissions (refer to Sections 2.5 and 2.6 waste management);
- point source emissions to air;
- significant fugitive emissions to all media, identifying the proportion of each substance released which is due to fugitives rather than point source releases;
- abnormal emissions from emergency relief vents, flares, etc.;
- indirect and direct emission of carbon dioxide associated with energy consumed or generated.

Emissions of carbon dioxide associated with energy use should be broken down by energy type and, in the case of electricity, by source, e.g. public supply, direct supply or on-site generation. Where energy is generated on-site, or from a direct (non-public) supplier, the Operator should specify and use the appropriate factor. Standard factors for carbon dioxide emissions are provided in the Energy Efficiency Guidance Note.

Where VOCs are released, the main chemical constituents of the emissions should be identified. The assessment of the impact of these chemicals on the environment will be carried out as in response to Section 4.1.

For waste, emissions relate to any wastes removed from the installation, or disposed of at the installation under the conditions of the Permit, e.g. landfill. Each waste should have its composition determined and the amounts expressed in terms of cubic metres or tonnes per month.

A suitable table on which to record this information is provided in the electronic version of this Guidance Note.

- 2. Compare the emissions with the benchmark values given in the remainder of this Section.
- 3. Where the benchmarks are not met, revisit the responses made in Section 2 as appropriate (see Section 1.2) and make proposals for improvements or justify not doing so.

IMPACT

Benchmark comparison

Benchmark status

3.2 The emission benchmarks

Guidance is given below on the release concentrations or removal efficiencies achievable when using the best combination of techniques. These benchmark values represent the performance associated with the use of BAT. They are not mandatory release limits and reference should be made to the "Guide for Applicants", to Section 1 above, and to Sections 3.2.1 and 3.2.2 below as to their application.

Limits should be identified in Permits for all pollutants which are likely to be present in emissions in significant quantities whether or not the pollutants are identified in the lists below.

3.2.1 Emissions to air associated with the use of BAT

The emissions quoted below are daily averages based upon continuous monitoring during the period of operation. Standard conditions of 273 K and 101.3 kPa for the dry gas apply. No correction is applied for the oxygen content of the emission. Care should always be taken to convert benchmark and proposed releases to the same reference conditions for comparison. To convert measured values to reference conditions, see Technical Guidance Note M2 (Ref. 22) for more information

Limits in Permits may be set for mean or median values over long or short periods. The periods and limits selected will reflect:

- the manner in which the emission may impact upon the environment
- likely variations which will arise during operation within BAT
- possible failure modes and their consequences
- the capabilities of the monitoring and testing system employed

Where emissions are expressed in terms of concentrations and where continuous monitors are employed, it is recommended that limits are defined in accordance with the table in Section 3.2.1.1 below.

Metals and metal compounds are reported as the element.

Benchmark status

	Benchmark level	Maximum		
Emission	As monthly average or from an extractive sample	allowed daily average when continuous monitoring is used provided monthly value is met	Techniques which may be considered to be BAT	Comment
Total particulate ^a	5mg/Nm ³	10 mg/Nm ³	Fabric filters. Wet scrubber for suitable types of dust	For some high- temperature processes ceramic filters may be BAT
Oxides of	100 mg/Nm ³	200 mg/Nm ³	Low NOx burner	Oxy fuel burner gives
nitrogen [⊳] (as NO₂)	300 mg/Nm ³	500 mg/Nm ³	Oxy fuel burner	reduced gas volume and better energy efficiency
Sulphur dioxide ^c	50 mg/Nm ³	200 mg/Nm ³	If scrubbing system is used	BAT will normally be to control sulphur releases by
	500 mg/Nm ³ , much lower if natural gas used as fuel	500 mg/Nm ³ , much lower if natural gas used as fuel	If control exercised via sulphur content of fuel	employing low sulphur fuels. Using higher sulphur fuels then scrubbing to achieve release limits will give rise to potentially significant cross-media effects
Hydrogen chloride ^d	10 mg/Nm ³	20 mg/Nm ³	Avoid scrap contaminated with chlorinated cutting oils and other chlorine sources	Presence in exhaust is an indicator of possible presence of dioxins
Carbon monoxide ^e	150 mg/Nm ³	300 mg/Nm ³	Effective burner control	
VOCs ^f (as Carbon)	50 mg/Nm ³	100 mg/Nm ³	Pre-treatment of potentially contaminated material. Effective burner control. Control over charging activity	
Dioxins ^g	0.1 ng/Nm ³ ITEQ		Purchase and inspection of scrap to avoid contaminated material. Pre-treatment. Carbon injection to filter	The BAT Reference document refers to higher levels under some circumstances. Where there is uncertainty about the individual isomers that are reported as "less than", a dual approach will be adopted where these values will be assumed to be zero and the result compared to these values as the limit of detection

It will not be necessary to specify limits for all of the above in all installations. For example, limits on products of combustion should not be applied to unheated milling plant handling drosses or slags.

INTRODUCTIONTECHNIQUESEMISSIONSIMPACTBenchmark comparisonBenchmark status

- ^a BREF tables 3.40; 3.41; 4.33; 4.34; 4.35; 5.47; 5.48; 6.16; 8.3; 8.4; 8.5; 8.6; 8.7; 8.8; 8.9; 9.11; 10.9; 11.16; 12.11; 12.12; 12.13; 12.14; 12.15 and 12.16.
- ^b BREF tables 3.40; 4.34; 4.35; 5.47; 5.48; 6.16 and 11.16.
- ^c BREF tables 3.40; 3.41; 5.48; 6.15 and 12.12.
- ^d BREF tables 4.34; 4.35; 6.16 and 11.17.
- e IPC Technical Guidance S2 2.03.
- ^f BREF tables 3.40; 4.35; 5.47; 5.48; 6.15; 6.16; 11.16 and 11.17.
- ^g BREF tables 3.40; 3.41; 4.35; 5.47; 5.48; 6.16 and 11.16.

Emission	Activity	Threshold annual use of solvent	Benchmark value as toluene mg/Nm ³	Basis for the benchmark
Solvents (various), see Solvent Directive 1999/13/EC	Coating and degreasing	2–10 tonnes > 10 tonnes	20 mg/Nm ³ 20 mg/Nm ³	15% fugitive emission, 10% fugitive emission, fugitive emission expressed as % of use
High risk ,Extremely hazardous to health, such as benzene, vinyl chloride and 1,2– dichloroethane	Various		2 –5 mg/Nm3	Parity with previous UK Guidance Notes

Compliance with the Solvent Directive 1999/13/EC is required including the use of a solvent management plan. A reduction scheme may be used instead of emission limits.

3.2.1.2 Additional or varied requirements for specific classes of activity

(a) Copper and copper alloys – additional values for other components based on extractive samples.

BREF Section 3.4 refers

(i) Melting copper, making and melting copper alloys including pre-treatment

-	-		
Emission	Benchmark level	Techniques which may be considered to be BAT	Comment
Phosphorus $(as P_2O_5)^a$	5 mg/Nm ³		When phosphor copper is used for deoxidising copper
	50 mg/Nm ³	Limitation on maximum concentration of phosphorus in product. Wet scrubbing followed by high-energy filters, or high-energy venturi scrubbers	Manufacture of phosphor copper
Copper, lead, zinc, or their compounds ^b	2 mg/Nm ³ in total	Filtration	The suite of metals for which limits are defined may be extended beyond the list
Cadmium, arsenic, nickel or their compounds ^b	0.5 mg/Nm ³ in total	Raw materials purchasing specification. Examination and testing upon receipt	specified to include all those likely to be present, whether present intentionally or as a consequence of error. Also metals which are not present in the raw materials may be excluded
Beryllium ^c	0.005 mg/Nm ³	Avoid, if deliberately introduced, then absolute filtration is required	Used to improving precipitation and hardening in the manufacture of intricate castings

IPC Technical Guidance S2 2.03.

INTRODUCTION **TECHNIQUES**

Benchmark comparison

Benchmark status

- b BREF tables 3.41 and 3.42. с
 - By extrapolation from occupational exposure limits.

(ii) Copper rod, wire and tube production: gas-fired shaft furnaces melting high purity copper

Emission	Benchmark level	Techniques which may be considered to be BAT	Comment
Particulate ab	5 mg/Nm ³	Fabric filter.	
VOCs ^{ab}	50 mg/Nm ³ (15 mg/Nm ³)	Quality control over raw materials and optimised combustion	
CO ^{abc}	1% at the burners (100 mg/Nm ³ if after-burners fitted)	Independent control of fuel/air ratio for each burner, with sequential gas monitoring	
NOx ^{ab}	100 mg/Nm ³	Low NOx burner	After-burners will increase
(as NO ₂)	(300 mg/Nm ³)	Oxy-fuel burner	emissions of NOx substantially

BREF Section 3.1.4.

b BREF table 3.40.

с IPC Technical Guidance S2 2.03.

(b) Aluminium and aluminium alloys additional values for other components based on extractive samples

i) - Primary aluminium production: electrolysis cells ^a

Emission	Benchmark level	Techniques which may be considered to be BAT	Comment
Particulate	5 mg/Nm ³	Fabric filter	
Sulphur dioxide	30 kg/tonne of metal produced	Control via sulphur content of anodes	
Polyfluorinated hydrocarbons	0.1 kg /tonne	Optimise anode effect frequency	
Hydrogen fluoride	0.2mg/Nm ^{3b}	Use fabric filter pre- coated with alumina	The alumina used to be fed to the cell
Total fluoride	0.5mg/Nm ^{3b}		

а BREF table 4.33. b

Total mass release of fluorides from electrolysis, refining and anode manufacture shall not exceed 1.0 Kg F per tonne of metal produced at existing plant and 0.6 kg F per tonne of metal produced at new plant.

ii) Furnaces for melting aluminium and its alloys: holding furnaces ^a

Emission	Benchmark level	Techniques which may be considered to be BAT	Comment
Hydrogen chloride	10 mg/Nm ³	Control of raw materials. Dry scrubbing using lime insulated filter	If present when chlorine is not used, then origin should be identified
Fluorides	1 mg/Nm ³	As above	May be added as a constituent of a refining flux or as a contaminant from a primary smelter

Benchmark status

(c) Lead, zinc and cadmium – additional values for other components based on extractive samples

(i) - All activities ^{abc}

Emission	Benchmark level	Techniques which may be considered to be BAT	Comment		
Copper, lead, zinc, nickel, or their compounds	2 mg/Nm ³ in total as metal	Filtration	The suite of metals for which limits are defined may be extended beyond the list specified to include all those		
Antimony, tin, tellurium	2 mg/Nm ³ in total as metal	Filtration	likely to be present, whether present intentionally or as a consequence of error. Also		
Cadmium, arsenic, mercury, thallium, selenium	0.5 mg/Nm ³ in total as metal	Raw materials purchasing specification. Examination and testing upon receipt	metals which are not present in the raw materials may be excluded		

BREF tables 5.47 and 5.49.

b IPC Guidance IPR 2/4.

с IPC Guidance IPR 2/5.

(ii) Smelting operations where sulphur compounds are evolved ^a

Emission	Benchmark level	Techniques which may be considered to be BAT	Comment		
Sulphur dioxide	99.7% removal	Conversion to sulphuric acid in double contact acid plant	Where the gas stream contains more than 5% sulphur dioxide		
	200 mg/Nm ³	Single contact plant with gas scrubbing	Where the gas contains less than 5% sulphur dioxide or small installations where acid recovery is not practicable		
Acid mist	50 mg/Nm ³				
^a BREE table 5.46					

BREF table 5.46.

Benchmark status

(d) Precious metals ^{ab} – additional values for other components based on extractive samples

Activity	Emission	Benchmark level	Techniques which may be considered to be BAT	
Silver recovery: pre- treatment including incineration of photographic film	Lead, silver and their compounds	No individual element to exceed 2 mg/Nm ³	Filtration	
Smelting silver residues	As above	As above	As above	
Silver remelt operations	Cadmium	0.5 mg/Nm ³	As above	
Miller process	Chlorine	3 mg/Nm ³	Process control and alkaline scrubber	
	Metal chlorides	5 mg/Nm ³	As above	
	Zinc oxide	5 mg/Nm ³	Reheating followed by fabric filter	
All dry PGM activities and processes	Platinum group metals	Total metal content not to exceed 0.05 mg/Nm ³	High-efficiency filtration	
Dissolution of PGMs	Chlorine and nitrosyl chloride	Total not to exceed 2 mg/Nm ³	Counter-current alkaline scrubber	
	Hydrogen chloride	10 mg/Nm ³		
	Oxides of nitrogen	200 mg/Nm ³		
Precipitation	Ammonia	15 mg/Nm ³	Water scrubbing	
Ignition	Ammonium chloride	10 mg/Nm ³	Process control to	
	Hydrogen chloride	10 mg/Nm ³	prevent loss of fine dusts, scrubbing to	
	Oxides of nitrogen	300 mg/Nm ³	remove hydrochloric acid	
	Platinum group metals	Total metal content not to exceed 0.05 mg/Nm ³		
Fire refining	As lead process	Table (i) of Section 3.2.1.2.(c)		

^a BREF Section 6.4, table 6.4.

^b IPC Technical Guidance Note S2 4.04.

Benchmark status

(e) Refractory metals and ferro alloys – additional values for other components based on extractive samples

Activity	Emission	Benchmark level	Techniques which may be considered to be BAT	Comment
Preparation and decomposition of ammonium paratungstate ^a	Ammonia	50 mg/Nm ³	Wet scrubbing	
Recovery of tungsten by the zinc process	Zinc and zinc oxide	5 mg/Nm ³	Filtration	
Molybdenum ^a	Sulphur dioxide, acid mist Ammonia	50 mg/Nm ³		For sulphur releases, see table (ii) of Section 3.2.1.2 (c)
All processes ^b	Chromium, manganese, tungsten, vanadium, molybdenum, titanium, tantalum, niobium and rhenium and their compounds as appropriate	2 mg/Nm ³		

^a BREF table 8.4.

BREF tables 8.3; 8.4; 8.7; 8.9 and 9.11.

(f) Alkali and alkaline earth metals – additional values for other components based on extractive samples

Final discharge to air after chlorine recovery and scrubbing

Emission	Benchmark level	Techniques which may be considered to be BAT	Comment
Chlorine ^a	1 mg/Nm ³	Liquefaction, followed by treatment of inerts in a multi- stage scrubber	Cell room air shall also be treated to this standard by scrubbing

BREF table 10.9.

а

а

(g) Nickel carbonyl process ^a – additional values for other components based on extractive samples

Emission	Benchmark level	Techniques which may be considered to be BAT	Comment
Nickel carbonyl	0.01 mg/Nm ³	Incineration	The release limit should be set at the practicable limit of detection

BREF Section 11.1.3.5, table 11.17.

EMISSIONS

Benchmark comparison

Benchmark status

(h) Carbon manufacture – additional values for other components based on extractive samples

BREF Section 12.4.3.1 refers

Activity	Emission	Benchmark level	Techniques which may be considered to be BAT	Comment
Pitch storage ^a	Volatile hydrocarbons	10 mg/Nm ³	Condenser, adsorber	Gases should be back vented during
	Condensable hydrocarbons	50 mg/Nm ³		delivery
Mixing and grinding ^b	Polycyclic aromatic hydrocarbons ^e VDI i	10 µg/Nm ³	Adsorber/dry scrubber	A regenerative after- burner has been used on some applications
	Polycyclic aromatic hydrocarbons ^e VDI ii	100 µg/Nm ³		
	Total hydrocarbons	25 mg/Nm ³		
Bake furnace	Benzo-[<i>a</i>]- pyrene	0.05 mg/Nm ³ **	Alumina scrubber filter unit.	The alumina used in this filter to be
exhaust gases ^{cd}	Polycyclic aromatic hydrocarbons ^e OSPAR 11	3 mg/Nm ³ **	** Lower values are reported in the BREF but have been challenged. Further	recycled into the reduction cell feed
	Total hydrocarbons	10 mg/Nm ³	reductions should be sought for new plant.	
	Gaseous fluoride	0.2mg/Nm ³		
	Total fluoride	0.5mg/Nm ³		

BREF table 12.11.

b BREF table 12.12.

с BREF table 12.13.

d BREF table 12.14.

е For reporting convention for polycyclic aromatic hydrocarbons see table 12.10 in BREF 12.4.3.1.

3.2.2 VOCs

The term "volatile organic compounds" includes all organic compounds released to air in the gas phase.

Other applicable standards and obligations

(Extracts from standards are quoted for ease of reference. The relevant standards should be consulted for the definitive requirements)

The "Solvents Directive" - The EC Directive on the limitation of emissions of VOCs due to the use of organic solvents in certain activities and installations is likely to be adopted soon.

"Reducing Emissions of VOCs and Levels of Ground Level Ozone: A UK Strategy" was published by the Department of the Environment in October 1993. It sets out how the Government expects to meet its obligations under the UNECE VOCs Protocol to reduce its emissions by 30% (based on 1988 levels) by 1999, including the reductions projected for the major industrial sectors. Waste Treatment included in the "other miscellaneous industries" sector with no specific reduction targets stated.

The UNECE convention on long-range transboundary air pollution - Negotiations are now under way which could lead to a requirement further to reduce emissions of VOCs.

Emission	Activity	Threshold annual use of solvent	Benchmark value as toluene mg/m ³	Basis for the Benchmark
Solvents (various) see Solvent Directive 1999/13/EC	coating and degreasing	2 – 10 tonnes > 10 tonnes	20 mg/m ³ 20 mg/m ³	15% fugitive emission 10% fugitive emission Fugitive emission expressed as % of use.
High Risk Extremely hazardous to health, such as benzene, vinyl chloride and 1,2 - dichloroethane	various		2 –5 mg/m ³	Parity with previous UK Guidance Notes

Benchmark emission values

Compliance with the Solvent Directive 1999/13/EC is required including the use of a solvent management plan. A reduction scheme may be used instead of emission limits.

Benchmark status

3.2.3 Emissions to water associated with the use of BAT

3.2.3.1 Controlled waters

BREF Reference 2.9.2.2

BREF Reference

2.9.2.2 Tables 2.17, 2.19, 2.20, 2.21, 2.22 and 2.23 Waste-water treatment systems can maximise the removal of metals using precipitation, sedimentation and filtration. The reagents used for precipitation will be defined by the mix of metals present, and may include hydroxide, sulphide or a combination of both. Concentrated effluents should be pre-treated before discharge into the final effluent treatment system, and techniques such as electrolysis, reverse osmosis and metal removal using ion exchange systems may need to be employed. Water discharges should be kept to a minimum by using closed cycle cooling systems and by maximising the re-use of treated process water.

Where automatic sampling systems are employed, limits in Permits may be defined such that:

- not more than 5% of samples shall exceed the limit value.
- Where spot samples are taken:
- no spot sample shall exceed the limit value by more than 50%.

Benchmark emissions to water associated with the use of BAT are given in the following table. These are not emission limit values, and site-specific issues such as the raw material, the process and other technical characteristics will be taken into account when setting emission limit values:

Substance	mg/litre
Total hydrocarbon oil	2
Biological oxygen demand (BOD) (5 day ATU at 20°C)	2.5
Chemical oxygen demand (COD) (2 hour)	125
Suspended solids	35
Cadmium and cadmium compounds expressed as Cd	0.01
Mercury and mercury compounds expressed as Hg	0.005
Copper and copper compounds expressed as Cu	0.5
Lead and lead compounds expressed as Pb	0.2
Arsenic and arsenic compounds expressed as As	0.1
Nickel and nickel compounds expressed as Ni	0.5
Zinc and zinc compounds expressed as Zn	0.5
Inorganic fluoride expressed as F	20
Silver expressed as Ag	0.5

The benchmarks are generally applicable for emissions to freshwater rivers. For discharges to estuaries higher values than those given above need to be justified on a site-specific basis at the time of the application. For very sensitive situations lower levels are achievable.

3.2.3.2 Discharges to a sewage treatment system

Discharges to a sewage treatment works will have different values depending on the site-specific conditions. However, the final discharge to controlled waters for the contribution from the IPPC installation, on a mass basis, should be equal to the above values after taking account of the volume of the waste water and effect of the sewage treatment process on the components.

Benchmark status

3.2.4 Standards and obligations

In addition to meeting the requirements of BAT, there are other national and international standards and obligations which must either be safeguarded through the IPPC Permit or, at least, taken into account in setting Permit conditions. This is particularly the case for any EC based EQSs.

(a) EC based EQ standards

IPPC: A Practical Guide (see Ref. 5) explains how these should be taken into account and contains an annex listing the relevant standards. (See Appendix 2 for equivalent legislation in Scotland and Northern Ireland.). They can be summarised as follows.

Air quality

- Statutory Instrument 1989 No 317, Clean Air, The Air Quality Standards Regulations 1989.
- Statutory Instrument 1997 No 3043, Environmental Protection, The Air Quality Regulations 1997.

Water quality

Directive 76/464/EEC on Pollution Caused by Dangerous Substances Discharged to Water contains two lists of substances. List I relates to the most dangerous, and standards are set out in various daughter Directives. List II substances must also be controlled. Annual mean concentration limits for receiving waters for List I substances can be found in SI 1989/2286 and SI 1992/337, the Surface Water (Dangerous Substances Classification) Regulations. Values for List II substances are contained in SI 1997/2560 and SI 1998/389. Daughter Directives cover EQS values for mercury, cadmium, hexachlorocyclohexane, DDT, carbon tetrachloride, pentachlorophenol, aldrin, dieldrin, endrin, isodrin, hexachlorobenzene, hexachlorobutadiene, chloroform, 1,2-dichloroethane, trichloroethane, perchloroethane and trichlorobenzene.

- Other waters with specific uses have water quality concentration limits for certain substances. These are covered by the following Regulations:
 - SI 1991/1597 Bathing Waters (Classification) Regulations;
 - SI 1992/1331 and Direction 1997 Surface Waters (Fishlife) (Classification) Regulations;
 - SI 1997/1332 Surface Waters (Shellfish) (Classification) Regulations;
 - SI 1996/3001 The Surface Waters (Abstraction and Drinking Water) (Classification) Regulations.

Future likely changes include:

- Some air and water quality standards may be replaced by new standards in the near future.
- The (Draft) Solvents Directive on the limitation of emissions of VOCs due to the use of organic solvents in certain activities and installations.

(b) Other standards and obligations

Those which may have relevance to non-ferrous metals processes include:

- Waste Incineration Directive;
- Large Combustion Plant Directive;
- Reducing Emissions of VOCs and Levels of Ground Level Ozone: a UK Strategy;
- Water Quality Objectives assigned water quality objectives to inland rivers and watercourses (ref. Surface (Rivers Ecosystem) Classification);
- The UNECE Convention on Long-Range Transboundary Air Pollution;
- The Montreal Protocol;
- The Habitats Directive (see Section 4.3).

Benchmark status

3.2.5 Units for benchmarks and setting limits in Permits

Releases can be expressed in terms of:

- "concentration" (e.g. mg/l or mg/Nm³) which is a useful day-to-day measure of the effectiveness
 of any abatement plant and is usually measurable and enforceable. The total flow must be
 measured/controlled as well;
- "specific mass release" (e.g. kg/t_{product} or input or other appropriate parameter) which is a
 measure of the overall environmental performance of the plant (including the abatement plant)
 compared with similar plants elsewhere;
- "absolute mass release" (e.g. kg/h, t/yr) which relates directly to environmental impact.

When endeavouring to reduce the environmental impact of an installation, its performance against each of these levels should be considered, as appropriate to the circumstances, in assessing where improvements can best be made.

When setting limits in Permits the most appropriate measure will depend on the purpose of the limit. It may also be appropriate to use surrogate parameters which reflect optimum environmental performance of plant as the routine measurement, supported by less frequent check analyses on the final concentration. Examples of surrogate measures would be the continuous measurement of conductivity (after ion exchange treatment) or total carbon (before a guard column in activated carbon treatment) to indicate when regeneration or replacement is required.

4 IMPACT

4.1 Assessment of the impact of emissions on the environment

The Operator should assess that the emissions resulting from the proposals for the activities/installation will provide a high level of protection for the environment as a whole, in particular having regard to EQSs, etc., revisiting the techniques in Section 2 as necessary (see Section 1.2).



With the application the Operator should:

- 1. Provide a description, including maps as appropriate, of the receiving environment to identify the receptors of pollution. The extent of the area may cover the local, national and international (e.g. transboundary effects) environment as appropriate.
- 2. Identify important receptors. This may include: areas of human population including noise or odour-sensitive areas, flora and fauna (i.e. Habitats Directive sites, special areas of conservation, Sites of Special Scientific Interest (SSSI or in Northern Ireland, ASSI) or other sensitive areas), soil, water, i.e. groundwater (water below the surface of the ground in the saturation zone and in direct contact with the ground and subsoil) and watercourses (e.g. ditches, streams, brooks, rivers), air including the upper atmosphere, landscape, material assets and the cultural heritage.
- 3. Identify the pathways by which the receptors will be exposed (where not self-evident).
- 4. Carry out an assessment of the potential impact of the total emissions from the activities on these receptors. Ref. 7 provides a systematic method for doing this and will also identify where modelling needs to be carried out, to air or water, to improve the understanding of the dispersion of the emissions. The assessment will include comparison (see *IPPC: A Practical Guide* (Ref. 5) and Section 3.2) with:
 - community EQS levels;
 - other statutory obligations;
 - non-statutory obligations;
 - environmental action levels (EALs) and the other environmental and regulatory parameters defined in Ref. 7.

In particular it will be necessary to demonstrate that an appropriate assessment of vent and chimney heights has been made to ensure that there is adequate dispersion of the minimised emission(s) to avoid exceeding local ground-level pollution thresholds and limit national and transboundary pollution impacts, based on the most sensitive receptor, be it human health, soil or terrestrial ecosystems.

Where appropriate the Operator should also recognise the chimney or vent as an emergency emission point and understand the likely behaviour. Process upsets or equipment failure giving rise to abnormally high emission levels over short periods should be assessed. Even if the applicant can demonstrate a very low probability of occurrence, the height of the chimney or vent should nevertheless be set to avoid any significant risk to health. The impact of fugitive emissions can also be assessed in many cases.

Consider whether the responses to Sections 2 and 3 and this assessment adequately demonstrate that the necessary measures have been taken against pollution, in particular by the application of BAT, and that no significant pollution will be caused. Where there is uncertainty about this, the measures in Section 2 should be revisited as appropriate to make further improvements.

Where the same pollutants are being emitted by more than one permitted activity on the installation, the Operator should assess the impact both with and without the neighbouring emissions.

4.2 The Waste Management Licensing Regulations

Application Form Question 4.2 **Explain how the information provided in other parts of the** *application also demonstrates that the requirements of the relevant objectives of the Waste Management Licensing Regulations 1994 have been addressed, or provide additional information in this respect.*

In relation to activities involving the disposal or recovery of waste, the Regulators are required to exercise their functions for the purpose of achieving the relevant objectives as set out in Schedule 4 of the Waste Management Licensing Regulations 1994. (For the equivalent regulations in Scotland, see Appendix 2. In Northern Ireland there are no equivalent regulations at the time of writing. Contact EHS for further information.)

The relevant objectives, contained in paragraph 4, Schedule 4 of the Waste Management Licensing Regulations 1994 (SI 1994/1056 as amended) are extensive, but will only require attention for activities which involve the recovery or disposal of waste. Paragraph 4 (1) is as follows:

- a) "ensuring the waste is recovered or disposed of without endangering human health and without using process or methods which could harm the environment and in particular without:
 - risk to water, air, soil, plants or animals; or
 - causing nuisance through noise or odours; or
 - adversely affecting the countryside or places of special interest;
- b) implementing, as far as material, any plan made under the plan-making provisions".

The application of BAT is likely already to address risks to water, air, soil, plants or animals, odour nuisance and some aspects of effects on the countryside. It will, however, be necessary for you briefly to consider each of these objectives individually and provide a comment on how they are being addressed by your proposals. It is also necessary to ensure that any places of special concern which could be affected, such as SSSIs, are identified and commented upon although, again, these may have been addressed in your assessment for BAT, in which case a cross-reference may suffice.

Operators should identify any development plans made by the local planning authority, including any waste local plan, and comment on the extent to which the proposals accord with the contents of any such plan (see Section 2.6).

4.3 The Habitats Regulations

Application Form Question 4.3		Provide an assessment of whether the installation is likely to have a significant effect on a European site in the UK and if it is, provide an assessment of the implications of the installation for that site, for the purposes of the Conservation (Natural Habitats etc) Regulations 1994 (SI 1994/2716).
	Your response should cover all relevant issues pertinent to your installation, including those below. In doing so you should justify your proposals against any indicative requirements stated.	

An application for an IPPC Permit will be regarded as a new plan or project for the purposes of the Habitats Regulations (for the equivalent regulations in Scotland and Northern Ireland see Appendix 2). Therefore, Operators should provide an initial assessment of whether the installation is likely to have a significant effect on any European site in the UK (either alone or in combination with other relevant plans or projects), and if so, an initial assessment of the implications of the installation for any such site. The application of BAT is likely to have gone some way towards addressing the potential impact of the installation on European sites and putting into place techniques to avoid any significant effects. The Operator should provide a description of how the BAT assessment has specifically taken these matters into account, bearing in mind the conservation objectives of any such site.

European sites are defined in Regulation 10 of the Habitats Regulations to include Special Areas of Conservation (SACs); sites of community importance (sites that have been selected as candidate SAC by member states and adopted by the European Commission but which are not yet formally classified); and Special Protection Areas (SPAs). It is also Government policy (set out in Planning Policy Guidance Note 9 (PPG 9) on nature conservation) that potential SPAs and candidate SACs should also be considered to be European sites for the purposes of Regulation 10.

Information on the location of European sites and their conservation objectives is available from

- English Nature (01733 455000), http://www.english-nature.org.uk
- Countryside Council for Wales (01248 385620), http://www.ccw.gov.uk
- Scottish Natural Heritage (0131 447 4784), http://www.snh.org.uk
- Joint Nature Conservation Committee (01733 866852), http://www.jncc.gov.uk
- Environment and Heritage Service, Northern Ireland, http://www.ehsni.gov.uk

The Regulator will need to consider the Operator's initial assessment, and if it concludes that the installation is likely to have a significant effect on a European site, then the Regulator will need to carry out an "appropriate assessment" of the implications of the installation in view of that site's conservation objectives. Because the Regulations impose a duty on the Regulator to carry out these assessments, it cannot rely on the Operator's initial assessments, and therefore the Regulator must be provided with any relevant information upon which the Operator's assessment is based.

Note that, in many cases, the impact of the Habitats Regulations will have been considered at the planning application stage, in which case the Regulator should be advised of the details.

REFERENCES

REFERENCES

For a full list of available Technical Guidance see Appendix A of the *Guide for Applicants* or visit the Environment Agency web site, http://www.environment-agency.gov.uk. Many of the references below are being made available free of charge for viewing or download on the web site. The same information can also be accessed via the SEPA web site, http://www.sepa.org.uk, or the NIEHS web site, http://www.ehsni.gov.uk. Most titles will also be available in hard copy from The Stationery Office (TSO). Some older titles are not available on the web site but can be obtained from TSO.

- 1. The Pollution Prevention and Control Act (1999) (http://www.click-tso.com).
- 2. The Pollution Prevention and Control Regulations (SI 1973, 2000) (http://www.click-tso.com).
- 3. The EC Reference Document on BAT in the Non-Ferrous Metals Industry (the BREF Note)
- 4. BREF Note on Associated Foundry and Casting Operations
- 5. IPPC: A Practical Guide (for England and Wales) (or equivalents in Scotland and Northern Ireland) (http://www.environment-defra.gov.uk).
- 6. IPPC Part A (1) Installations: Guide for Applicants (includes **preparation of a site report**) (http://www.environment-agency.gov.uk/business/techguide/ippc/107285/)
- 7. Assessment methodologies
 - E1 BPEO Assessment Methodology for IPC
 - IPPC Environmental Assessments for BAT (in preparation as H1)
 - Management system references:
 - Sector specific

8.

- 9. Waste minimisation support references:
 - Environment Agency web site. Waste minimisation information accessible via: http://www.environment-agency.gov.uk/waste
 - Waste Minimisation an environmental good practice guide for industry (helps industry to minimise waste and achieve national environmental goals). Available free to companies who intend to undertake a waste reduction programme (tel 0345 33 77 00)
 - Profiting from Pollution Prevention 3Es methodology (emissions, efficiency, economics). Video and A4 guide aimed at process industries. Available from Environment Agency, North East region (tel 0113 244 0191, ask for regional PIR)
 - Waste Minimisation Interactive Tools (WIMIT). Produced in association with the ENVIROWISE and the BOC Foundation (a software tool designed for small and medium businesses). Available free from The Environmental Helpline (tel 0800 585794)
 - ENVIROWISE is a joint DTI/DEFRA programme, with over 200 separate case studies, good practice guides, leaflets, flyers, software tools and videos covering 12 industry sectors, packaging, solvents and the generic areas of waste minimisation and cleaner technology ENVIROWISE is accessible via a free and confidential helpline (tel 0800 585794) or via the web site, www.envirowise.org.uk/
 - ENVIROWISE, Increased Profit Through Improved Materials Additions: Management/Technical Guide, GG194/195
 - Waste Management Information Bureau. The UK's national referral centre for help on the full range of waste management issues. It produces a database called Waste Info, which is available for online searching and on CD-ROM. Short enquiries are free (tel 01235 463162)
 - Institution of Chemical Engineers Training Package E07 Waste Minimisation. Basic course which contains guide, video, slides, OHPs, etc. (tel 01788 578214)
- 10. Water efficiency references:
 - ENVIROWISE, Simple measures restrict water costs, GC22
 - ENVIROWISE, Effluent costs eliminated by water treatment, GC24
 - ENVIROWISE, Saving money through waste minimisation: Reducing water use, GG26
 - ENVIROWISE Helpline 0800 585794
- 11. Environment Agency (1998) Optimum use of water for industry and agriculture dependent on direct abstraction: Best practice manual. R&D Technical Report W157, WRc Dissemination Centre, Swindon (tel 01793 865012)
- 12. Releases to air references:
 - BREF on Waste Water and Waste Gas Treatment
 - A1 Guidance on effective flaring in the gas, petroleum etc industries, 1993, ISBN 0-11-752916-8
 - A2 Pollution abatement technology for the reduction of solvent vapour emissions, 1994, £5.00, ISBN 0-11-752925-7
 - A3 Pollution abatement technology for particulate and trace gas removal, 1994, £5.00, ISBN 0-11-752983-4
 - Landfill gas flaring
 - Part B PG1/3 Boilers and Furnaces 20–50 MW net thermal input (ISBN 0-11-753146-4-7)
 - Part B PG1/4 Gas Turbines 20–50 MW net thermal input (ISBN 0-11-753147-2)
- 13. Releases to water references
 - BREF on Waste Water and Waste Gas Treatment

- A4 Effluent Treatment Techniques, TGN A4, Environment Agency, ISBN 0-11-310127-9 (EA web site)
- Environment Agency, Pollution Prevention Guidance Note Above-ground oil storage tanks, PPG 2, gives
 information on tanks and bunding which have general relevance beyond just oil (EA web site)
- The Control of Pollution (Oil Storage) Regulations 2001.
- Mason, P. A., Amies, H. J., Sangarapillai, G. Rose, Construction of bunds for oil storage tanks, Construction Industry Research and Information Association (CIRIA), Report 163, 1997, CIRIA, 6 Storey's Gate, Westminster, London SW1P 3AU. Abbreviated versions are also available for masonry and concrete bunds (www.ciria.org.uk online purchase)
- 14. Dispersion Methodology Guide D1 (EA web site summary only)
- 15. IPPC Energy Efficiency Guidance Note (the consultation version, available on the web site, should be used until the final version is published). Other energy efficiency guidance is given in the following EEBPP sources.
 - EEBPP publication Non Ferrous Metals, the Essentials
 - EEBPP publication A manager's guide to optimising furnace performance, GPG253
 - The Environment and Energy Helpline 0800 585794
 - Websites http://www.energy-efficiency.gov.uk and http://www.thecarbontrust.co.uk
- 16. BS 5908: Code of Practice for Fire Precautions in the Chemical and Allied Industries
- 17. Environment Agency, Pollution Prevention Guidance Notes Pollution prevention measures for the control of spillages and fire-fighting run-off, PPG 18, gives information on sizing fire water containment systems (EA web site)
- 18. Investigation of the criteria for, and guidance on, the landspreading of industrial wastes final report to the DEFRA, the Environment Agency and MAFF, May 1998
- 19. Agency guidance on the exemption 7 activity (proposed)
- 20. COMAH guides:
 - A Guide to the Control of Major Accident Hazards Regulations 1999, Health and Safety Executive (HSE) Books L111, 1999, ISBN 0 07176 1604 5
 - Preparing Safety Reports: Control of Major Accident Hazards Regulations 1999, HSE Books HS(G)190, 1999
 - Emergency Planning for Major Accidents: Control of Major Accident Hazards Regulations 1999, HSE Books HS(G)191, 1999
 - Guidance on the Environmental Risk Assessment Aspects of COMAH Safety Reports, Environment Agency, 1999 (EA web site)
 - Guidance on the Interpretation of Major Accidents to the Environment for the Purposes of the COMAH Regulations, DEFRA, 1999, ISBN 753501 X, available from TSO
- 21. Assessment and Control of Environmental Noise and Vibration from Industrial Activities (joint agencies guidance in preparation)
- 22. Monitoring guidance (http://www.environment-agency.gov.uk/business/techguide/monitoring/):
 - M1 Sampling facility requirements for the monitoring of particulates in gaseous releases to atmosphere, March 1993, £5.00, ISBN 0-11-752777-7 (revision due 2002)
 - M2 Monitoring emissions of pollutants at source, January 1994, £10.00, ISBN 0-11-752922-2 (revision due 2002)
 - M3 Standards for IPC Monitoring Part 1: Standards, organisations and the measurement infrastructure, August 1995, £11.00, ISBN 0-11-753133-2 (revision due 2002)
 - M4 Standards for IPC Monitoring Part 2: Standards in support of IPC Monitoring, revised 1998 (revision due 2002)
 - M8 Environmental Monitoring Strategy Ambient Air, 2001, £50. ISBN 0 11 310175 9
 - M9 Monitoring Methods for Ambient Air, 2001, £50. ISBN 0 11 310176 7
 - MCERTS approved equipment link via http://www.environment-agency.gov.uk/business/mcerts/ "Guidance for Business and Industry"
 - Direct Toxicity Assessment for Effluent Control: Technical Guidance (2000), UKWIR 00/TX/02/07
- 23. IPC Guidance Notes: (http://www.environment-agency.gov.uk/business/techguide/IPC/)
 - S2 1.01 Combustion Processes: large boilers and furnaces 50 MW(th) and over
 - S2 1.04 Combustion Processes: Waste and recovered oil burners 3 MW(th) and over
 - S2 4.03 Inorganic Acids and Halogens
- 24. The Categorisation of Volatile Organic Compounds, DOE Research Report No DOE/HMIP/RR/95/009 (http://www.environment-agency.gov.uk/business/techguide/publications/voc/)
- 25. Odour Assessment and Control Guidance for Regulators and Industry (joint agencies guidance in preparation)
- 26. Policy and Practice for the Protection of Groundwater (PPPG) (EA web site)
- 27. Working at Construction and Demolition-sites (PPG 6) (EA web site)

DEFINITIONS AND ABBREVIATIONS

AFNOR APP	Association Française de Normalisation Alkaline peroxide process
AOX	Absorbable organic halogens
ASTM	American Society for Testing and Materials
BAT	Best available techniques – see IPPC A Practical Guide or the Regulations for further definition
BAT criteria	The criteria to be taken into account when assessing BAT, given in Schedule 2 of the PPC Regulations
BOD	Biochemical oxygen demand
BREF	BAT Reference Document
BSI	British Standards Institute
CCLA	Climate change levy agreement
CCTV	Closed circuit Television
CEM	Continuous emissions monitoring
CEN	Comité Européen de Normalisation
CHP	Combined heat and power
COD	Chemical oxygen demand
COS	
	Carbon oxy sulphide
DEFRA	Department of Environment, Food and Rural Affairs
DIN	Deutsches Institut für Normung
EAL	Environmental action level
EMAS	EC Eco Management and Audit Scheme
EMS	Environmental management system
EP	Electrostatic precipitator
EQS	Environmental quality standard
ETP	Effluent treatment plant
Gangue	Valueless material associated with an ore
GBR	General binding rules
Kw	Kilowatt
Kwh	Kilowatt hour
IPPC	Integrated Pollution Prevention and Control
ISO	International Standardisation Organisation
ITEQ	International Toxicity Equivalents
MCERTS	Monitoring Certification Scheme
NIEHS	Northern Ireland Environment and Heritage Service
NOx	Oxides of Nitrogen
OPRA	Operator and Pollution Risk Appraisal
PAH	Polyaromatic hydrocarbons
PFC	Polyfluorinated hydrocarbons
PGM	Platignum group metals
PM ₁₀	Particulate matter <10microns
PMs	Precious Metals
Ppv	Peak particle velocity
QA	Quality assurance
QC	Quality control
SAC	Special Area of Conservation
SCA	Society of Chemical Analysts
SECp	Specific energy consumption
SEPA	Scottish Environment Protection Agency
SPA	Special Protection Area
SSSI	Sites of Special Scientific Interest
TSS	Total suspended solids
TOC	Total organic carbon
US EPA	United States Environmental Protection Agency
VDI	Verein Deutscher Ingenieure
VDV	Vibration dose value
VOC	Volatile organic compound

APPENDIX 1 Some Common Monitoring and Sampling Methods

Determinand	Method	Detection limit and uncertainty	Valid for range (mg/l)	Standard
Suspended solids	Filtration through glass fibre filters	1 mg/l 20%	10–40	ISO 11929: 1997 EN872 Determination of suspended solids
COD	Oxidation with dichromate	12 mg/l 20%	50–400	ISO 6060: 1989 Water Quality- Determination of chemical oxygen demand
BOD₅	Seeding with micro organisms and measurement of oxygen content	2 mg/l 20%	5–30	ISO 5815: 1989 Water Quality – Determination of biological oxygen demand after 5 days, dilution and seeding method EN1899 (BOD 2 Parts)
AOX	Adsorption on activated carbon and combustion	_ 20%	0.4–1.0	ISO 9562: 1998 EN1485 – Determination of adsorbable organically bound halogens.
Total P				BS 6068: Section 2.28 1997 Determination of phosphorus –ammonium molybdate spectrometric method
Total N				BS 6068: Section 2.62 1998 – Determination of nitrogen Part 1 Method using oxidative digestion with peroxydisulphate BS EN ISO 11905
рН				SCA The measurement of electric conductivity and the determination of pH, ISBN 0117514284
Turbidity				SCA Colour and turbidity of waters 1981, ISBN 0117519553 EN27027:1999
Flowrate	Mechanical ultrasonic or electromagnetic gauges			SCA Estimation of flow and load, ISBN 011752364X
Temperature				
TOC				SCA The instrumental determination of total organic carbon and related determinants, 1995, ISNB 0117529796 EN1484:1997
Fatty acids				Determination of volatile fatty acids in sewage sludge, 1979, ISBN 0117514624
Metals				BS 6068: Section 2.60 1998 – Determination of 33 elements by inductively coupled plasma atomic emission spectroscopy
Chlorine				BS6068: Section 2.27 1990 – Method for the determination of total chlorine: iodometric titration method
Chloroform, bromoform				BS 6068: Section 2.58 Determination of highly volatile halogenetaed hydrocarbons – Gas chromatographic methods
Dispersants, Surfactants: Anionic, Cationic, non-ionic				SCA Analysis of surfactants in waters, wastewaters and sludges, ISBN 01176058 EN903:1993 (Used for anionic surfactants)
Pentachloro- phenol				BS5666 Part 6 1983 – Wood preservative and treated timber quantitative analysis of wood preservatives containing pentachlorophenol EN12673:1997 (used for chlorophenol and polychlorinated phenols)
Formaldehyde				SCA The determination of formaldehyde, other volatile aldehydes and alcohols in water
Phosphates and nitrates				BS 6068: Section 2.53 1997 Determination of dissolved ions by liquid chromatography
Sulphites and sulphates				BS 6068: Section 2.53 1997 Determination of dissolved ions by liquid chromatography
Ammonia				BS 6068: Section 2.11 1987 – Method for the determination of ammonium: automated spectrometric method
Grease and oils	IR absorption	0.06 mg/kg		SCA The determination of hydrocarbon oils in waters by solvent extraction IR absorption and gravimetry, ISBN 011751 7283

Table A1.1 Measurement methods for common substances to water

APPENDIX 1 - MONITORING AND SAMPLING METHODS

Table A1.2: Measurement methods for air emissions

Determinand	Method	Av'ging time, detection limit and uncertainty	Compliance criterion	Standard
Formaldehyde	Impingement in 2,4-dinitro-phenyl- hydrazine HPLC	1 h 1 mg/Nm ³ 30%	Two samples taken. Each result below limit after subtraction of measurement uncertainty	NIOSH
Ammonia	lon chromato- graphy	1 h 0.5mg/Nm ³ 25%		US EPA Method 26
VOCs speciated	Adsorption thermal desorption GCMS	1 h 0.1 mg/Nm ³ 30%		BS EN 1076: Workplace atmospheres. Pumped sorbent tubes for the determination of gases and vapours. Requirements and test methods.
Chloroform	Absorption on activated carbon, solvent extraction, GC analysis	1 h 1 mg/Nm ³ 20%		MDHS 28 Chlorinated hydrocarbon solvent vapours in air (modified)
Oxides of sulphur	UV fluorescence automatic analyser	1 h 1 ppm 10%	95% of hourly averages over a year below specified limit	ISO 7935 (BS6069 Section 4.4) Stationary source emissions – Determination of mass concentrations of sulphur dioxide CEN Standard in preparation
	Wet sampling Train, Ion, chromatography	1 h 1 mg/Nm ³ 25%	Two samples taken. Each result below limit after subtraction of measurement uncertainty	ISO 7934 (BS6069 Section 4.1) Method for the determination of the mass concentration of sulphur dioxide – hydrogen peroxide/barium perchlorate method

Measurement uncertainty is defined as total expanded uncertainty at 95% confidence limit calculated in accordance with the Guide to the Expression of Uncertainty in Measurement, ISBN 92-67-10188-9, 1st edn., Geneva, Switzerland, ISO, 1993.

See also Monitoring Guidance (Ref. 22).

APPENDIX 2 Equivalent Legislation In Scotland and Northern Ireland

The legislation referred to in the text is that for England and Wales. The following are the equivalents for Scotland and Northern Ireland.

Table A2.1				
Equivalent				
legislation				

England and Wales	Scotland	Northern Ireland
PPC Regulations (England and Wales) 2000	PPC (Scotland) Regulations 2000; SI 200/323	
Waste Management Licensing Regulations SI 1994 1056	Waste Management Licensing Regulations SI 1994 1056	No equivalent
The Water Resources Act 1991	COPA 1974 (S30A-30E equiv to Part III WRA91) Natural Heritage (Scotland) Act 1991 (Part II equiv to Part I WRA91)	The Water (NI) Order 1999
SI 1989 No 317: Clean Air, The Air Quality Standards Regulations 1989	SI 1989/317: Clean Air, The Air Quality Standards Regulations 1989	The Air Quality Standards Regulations (Northern Ireland) 1990. Statutory Rules of Northern Ireland 1990 No 145
SI 1997 No 3043: Environmental Protection, The Air Quality Regulations 1997	SSI 2000/97 The Air Quality (Scotland) Regs	No equivalent
SI 1989 No 2286 and 1998 No 389: the Surface Water (Dangerous Substances Classification) Regulations. (Values for List II substances are contained in SI 1997/2560 and SI 1998/389)	SI 1990/126 Surface Water (Dangerous Substances) (Classification) (Scotland) Regs	Surface Waters (Dangerous Substances) (Classification) Regulations 1998. Statutory Rules of Northern Ireland 1998 No 397 SI1991/1597:
SI 1991/1597: Bathing Waters (Classification) Regs	SI 1991/1609 Bathing Waters (Classification) (Scotland) Regs	The Quality of Bathing Water Regulations (NI) 1993
SI 1992/1331 and Direction 1997 Surface Waters (Fishlife) (Classification) Regs.	SI 1997/2471 Surface Waters (Fishlife) (Classification) Regs	The Surface Water (Fishlife) (Classification) Regulations (NI) 1997
SI 1997/1332 Surface Waters (Shellfish) (Classification) Regs	SI 1997/2470 Surface Waters (Shellfish) (Classification) Regs	The Surface Water (Shellfish) (Classification) Regulations (NI) 1997
SI 1994/2716 Conservation (Natural Habitats etc) Regulations 1994	SI 1994/2716 Conservation (Natural Habitats etc) Regs	Conservation (Natural Habitats etc) Regulations (Northern Ireland) 1995
Control of Major Accident Hazards Regulations 1999 (COMAH)	SI 1999/743 Control of Major Accident Hazards Regs	Control of Major Accident Hazard Regulations (Northern Ireland) 2000